# Physical Capacity and Complications During and After Inpatient Rehabilitation for Spinal Cord Injury

Janneke Haisma

The Health Research and Development Council of The Netherlands supported the work presented in this thesis; grant number 1435.0003 and 1435.0025

ISBN: 978-90-8559-368-3

Cover illustration: © Peter Struycken, Arcadeverlichting NA<br/>i, 1992-1993, c/o Pictoright Amsterdam 2008

Layout and printing: Optima Grafische Communicatie, Rotterdam, The Netherlands

# Physical Capacity and Complications During and After Inpatient Rehabilitation for Spinal Cord Injury

Fysieke capaciteit en complicaties gedurende en na klinische dwarslaesie revalidatie

# Proefschrift

ter verkrijging van de graad van doctor aan de Erasmus Universiteit Rotterdam op gezag van de rector magnificus

Prof.dr. S.W.J. Lamberts

en volgens besluit van het College voor Promoties.

de openbare verdediging zal plaatsvinden op woensdag 23 april 2008 om 13.45 uur

door

Jantje Aagg Haisma geboren te Groningen

ERASMUS UNIVERSITEIT ROTTERDAM

# Promotiecommissie

Promotor:	Prof.dr. H.J. Stam
Overige leden:	Prof.dr. C.M.F. Dirven Prof.dr. M.T.E. Hopman Prof.dr. E.W. Steyerberg
Copromotoren:	Dr. J.B.J. Bussmann Dr. L.H.V. van der Woude

# Contents

Chapter 1	Introduction	7
Chapter 2	Physical capacity in wheelchair-dependent persons with a spinal cord injury: a critical review of the literature	15
Chapter 3	Changes in physical capacity during and after inpatient rehabilitation in subjects with a spinal cord injury	33
Chapter 4	Prognostic models for physical capacity at discharge and 1 year after discharge from rehabilitation in persons with a spinal cord injury	49
Chapter 5	Complications following spinal cord injury: occurrence and risk factors in a longitudinal study during and after inpatient rehabilitation	69
Chapter 6	Physical fitness in persons with a spinal cord injury: the association with complications and duration of rehabilitation	81
Chapter 7	Physical independence and health-related functional status following a spinal cord injury: a prospective study on the association with physical capacity and complications	93
Chapter 8	Discussion	111
	Summary	123
	Samenvatting	127
	References	131
	Dankwoord	142
	Curriculum vitae	143
	Articles in international journals	144

# Contents

The following manuscripts are part of this thesis and have been published or submitted for publication:

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Bussmann JB. Physical capacity in wheelchair-dependent persons with a spinal cord injury: a critical review of the literature. *Spinal Cord* 2006; 44: 642-652.

Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Dallmeijer AJ, de Groot S, van der Woude LH. Changes in Physical Capacity During and After Inpatient Rehabilitation in Subjects With a Spinal Cord Injury. *Arch Phys Med Rehabil* 2006; 87: 741-748.

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Groot de S, Dallmeijer AJ, Bussmann JB. Prognostic models for physical capacity at discharge and 1 year after discharge from rehabilitation in persons with a spinal cord injury. *Arch Phys Med Rehabil* 2007; 88:1694-1703.

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Post MW, Bussmann JB. Complications following spinal cord injury: occurrence and risk factors in a longitudinal study during and after inpatient rehabilitation. *J Rehabil Med 2007*, 39: 393-398.

Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Post MW, Dallmeijer AJ, van der Woude LH. Physical fitness in persons with a spinal cord injury: the association with complications and duration of rehabilitation. *Clin Rehabil* 2007; 41: 932-940.

Haisma JA, Bussmann JB, Stam HJ, Bergen MP, Sluis TA, van den Berg-Emons HJ, Post MW, van der Woude LH. Physical independence and health-related functional status following a spinal cord injury: a prospective study on the association with physical capacity and complications. Submitted to *J Rehabil Med*.

1

Introduction

# Spinal cord injury

A spinal cord injury (SCI) is an interruption of the neural pathways in the spinal canal and is characterized by muscle weakness, loss of sensation and autonomic dysfunction below the level of the lesion. The extent of these neurological deficits is determined by both the level and completeness of the lesion. A complete lesion results in loss of motor function and sensation in the lowest sacral segment whereas, following an incomplete lesion function in this segment is maintained.<sup>1</sup> The SCI can either have a traumatic or a non-traumatic cause. In traumatic cases, injury is typically the result of a traffic or sporting accident, but an increasing number of injuries result from a low impact fall, for example, in those with osteoporosis.<sup>2,3</sup> A non-traumatic SCI may be caused by metastasis, infection and spinal haemorrhage or infarction.<sup>3,4</sup> An estimated 183 new traumatic SCI occur in The Netherlands per year, of whom 154 survive hospitalization, which corresponds to an incidence of over 10 per 1,000,000 per year.<sup>2</sup> The incidence of non-traumatic SCI is easily underestimated because many are not registered as SCI, but this incidence largely exceeds the incidence of traumatic injuries.<sup>2,4-6</sup> As compared to the able-bodied population, the life expectancy of those with SCI is reduced. For example, a 20-year-old man with a traumatic paraplegia has an estimated life expectancy of 46 years, whereas an able-bodied man the same age, will have another 58 years to live.<sup>3,7</sup> However, because the treatment of complications has improved over the past decades, the life expectancy following SCI has increased.<sup>2</sup> Although the life expectancy may have improved, many patients report a low level of functioning and well-being.<sup>8-10</sup> Functioning following SCI may be threatened because most patients have complications, have a low physical capacity, and depend on others for daily activities.<sup>6,9,11,12</sup> Therefore, it is important to investigate opportunities to optimize functioning following SCI.

If injury or disease to the spinal cord leads to neurological deficit, patients are admitted to a hospital at least until circulatory and respiratory functions have stabilized, after which most of the traumatic cases are discharged to specialized rehabilitation centres.<sup>2,3,5</sup> Clinicians at these centres aim to provide care according to guidelines supported by the Dutch Flemish Spinal cord Society, and they teach and train patients to function as independently as possible, as soon as possible.<sup>3</sup> After inpatient rehabilitation, over 90% of the patients are discharged home and others are discharged to guided living facilities or nursing homes.<sup>5</sup> In The Netherlands the mean duration of hospitalization is 49 days and the mean inpatient rehabilitation is 240 days.<sup>13</sup> Length of stay not only depends on severity of the injury and complications, but also on the availability of services and post-discharge care.<sup>5,13</sup> After discharge, patients will be seen at rehabilitation centres for outpatient training or routine follow-up visits. The occurrence of complications could necessitate

readmission to the hospital or rehabilitation centre.<sup>14</sup> Of secondary importance, but worth considering, are the expenses involved in the life-time care of those with SCI.<sup>7</sup> Based on findings of the Models Systems in the United States, the costs directly attributable to SCI are approximately  $\in$  190,000 in the first year, and  $\in$  19,000 for each subsequent year for a person with a complete paraplegia; the costs for those with tetraplegia may be more than double.

# Spinal cord injury and functioning

A spinal cord injury is a condition that suddenly changes a patient's functioning. The International Classification of Functioning, Disability and Health (ICF) helps visualize how a condition, like SCI, influences functioning, and how the aspects of functioning interact with one another (Figure 1).<sup>15,16</sup>

Both the separate aspects of functioning, and the interactions between them may be influenced by environmental factors and by personal factors. The SCI directly leads to impairments in voluntary control, sensation and autonomic function. One



Figure 1 Interactions as described in the International Classification of Functioning, Disability and Health (ICF) model.<sup>15</sup>



**Figure 2** Interactions between aspects of functioning, complications, environmental and personal factors evaluated in this thesis; adapted from the International Classification of Functioning, Disability and Health (ICF) model.<sup>15</sup>

of the functions that is also impaired is the physical capacity, which is the combined ability of muscles and respiratory and cardiovascular systems to attain a certain level of activity.<sup>12,17,18</sup> Physical capacity is sometimes referred to as physical fitness or fitness. The SCI not only results in neurological impairments, but patients with SCI are also at risk of complications, which cause secondary impairments. Both the SCI and the complications will influence activities and participation.<sup>19,20</sup> It is important to get insight into changes in functioning following SCI, and how these changes interact with complications and environmental or personal factors (Figure 2).

# **Physical capacity**

We can evaluate physical capacity by determining muscle strength and respiratory or cardiovascular function separately, or by determining their combined function (Figure 3). To determine muscle strength, manual muscle testing is used in clinical practice, and isokinetic or handheld dynamometry are used in research.<sup>12,21</sup> Dysfunction of respiratory muscles and respiratory complications affect respiratory function which can be assessed, for example, with spirometry.<sup>22</sup> Cardiovascular changes occur especially in those with a cervical or high thoracic lesion,  $^{23,24}$  but it is difficult to quantify their cardiovascular function because the stroke volume is not easily measured in rehabilitation practice, and heart rate is difficult to interpret.<sup>17,25</sup> However, the combined function of cardiovascular, neuromuscular and respiratory systems can be measured with the aerobic capacity (i.e. the endurance capacity), the anaerobic capacity or the sprint power.<sup>26</sup> The endurance capacity comprises both the peak oxygen uptake and peak aerobic power output during a peak exercise test with hand rim wheelchair propulsion or arm crank exercise. The peak oxygen uptake is reflected by oxygen consumption; the peak power output is the product of the velocity of the wheelchair and the drag force of wheelchair and patient.<sup>27</sup>

Previous studies have used these different methods when they established a component of physical capacity, but comparative values for the level of different components of physical capacity in a general population with SCI is still lacking, and the different methods used need to be critically evaluated (Chapter 2).<sup>17</sup>



Figure 3 Interactions between components of physical capacity.

# Importance of evaluating physical capacity

Following SCI, physical capacity is an important aspect of functioning, which is supported by previous studies that have shown that physical capacity interacts with other aspects of functioning and with well-being.<sup>12,28-30</sup> Its interaction with complications suggests that a low physical capacity predisposes to complications and that, conversely, complications influence physical capacity.<sup>31,32</sup> Furthermore, the interaction between physical capacity and activities suggests that a basic physical capacity is a prerequisite to being active, and that daily activities could influence physical capacity (Figure 2).<sup>33-36</sup> Because most patients are wheelchair bound and have a limited physical capacity, the performance of daily activities is strenuous and, as a consequence, they may be discouraged from being active and involved.<sup>20,37-39</sup> These interactions could indicate an unfortunate sequence of events: a low physical capacity makes activities strenuous, and consequently limits daily activities and reduces social involvement, and the consequent inactive lifestyle further reduces physical capacity and daily activities, and so on.<sup>40</sup>

However, because most of the aforementioned studies focused mainly on male subjects, those with a traumatic paraplegia or relatively active subjects,<sup>12</sup> their results cannot be translated to a general population with SCI. Additionally, the reported interactions were mostly cross-sectional, which cannot help establish the hypothesized sequence of events.<sup>41</sup> Until now prospective data on changes in physical capacity, and the interactions with other aspects of functioning or complications have not been described in a general SCI population.

# Complications

Previous studies demonstrated that complications are a common problem following SCI,<sup>11,42</sup> that they may necessitate readmission to a rehabilitation centre or hospital,<sup>14,43</sup> and that they are associated with functioning (Figure 2). Complications interact with body functions. With a sacral pressure sore, for example, patients cannot sit. If a pressure sore requires a patient to take bed rest, he or she may be at risk of other complications, such as venous thrombosis.<sup>9</sup> Furthermore, complications interfere with activities. If a patient cannot sit in, or ride, his wheelchair he will not be able to perform activites.<sup>12,44</sup> Conversely, on the long term a low level of activity may be associated with complications like glucose intolerance or cardiovascular disease.<sup>45,46</sup> Complications are associated with participation; for example, patients with complications are less likely to be employed.<sup>47</sup> On the other hand, participation in training programs may reduce the risk of complications.<sup>48</sup> Unfortunately, these interactions could aggravate

the previously suggested sequence of events, because complications could further reduce functioning, and the consequent reduced level of functioning could increase the risk of additional complications. Although the studies that focussed on complications may be more representative of a general population with SCI than those determining physical capacity, most are cross-sectional. This means that the hypothesized sequence of events can, again, only be established with new prospective data.

#### Changes in physical capacity and complications over time

Positive changes in physical capacity are expected to be not only the result of natural recovery, but also the effect of training programs during rehabilitation.<sup>3,49</sup> Alternatively, a decline in physical capacity may be caused by the cumulative effect of relative inactivity, as compared to the situation before injury. To establish the net effect of these proposed interactions prospective data were required (Chapter 3).41 Modifiable lifestyle characteristics do influence physical capacity in the able-bodied population, but their effect in those with SCI is largely unkown.<sup>50,51</sup> Furthermore, prognostic models for their level of physical capacity have not been evaluated. Therefore, insight into the prognostic value of person, lifestyle and lesion-related characteristics following SCI are required (Chapter 4). Similarly, in order to evaluate the recovery of complications and to effectively target prevention or screening programs, changes in the occurrence of complications over time and their risk factors need to be investigated (Chapter 5). During rehabilitation, effort is put into both the improvement of physical capacity and into the prevention or treatment of complications;<sup>3</sup> however, the effectiveness of this effort has not been evaluated, and prospective data on the complex interactions between changes in physical capacity, the duration of rehabilitation and the occurrence of complications are lacking (Chapter 6).

# Activities and participation

A low physical capacity and the occurrence of complications become especially noticeable when they interfere with the performance of daily activities (Figure 2). However, in comparison with the able-bodied population, a decline in physical capacity and complications invariably have a greater impact on functioning in those with SCI.<sup>52</sup> Physical independence interacts with social involvement, sense of control and wellbeing, which makes it especially important to investigate.<sup>37,53</sup> Cross-sectional studies showed that physical capacity and complications are associated with physical independence following SCI.<sup>13,20,33,34,54</sup> However, prospective studies on these associations may provide more insight into the directions of the interactions (Chapter 7).

Introduction

# The Umbrella project

The present study is an extension of a Dutch multi-centre study, the Umbrella project. This project investigates the restoration of mobility during SCI rehabilitation. Eight Dutch rehabilitation centres with a specialized spinal cord unit participated. Between 1999 and 2005 they collected findings in 225 subjects with SCI who were wheelchair bound at the start of active rehabilitation. Several outcome measures, which reflect different aspects of functioning, were monitored on 4 separate occasions: at the start of rehabilitation, 3 months into active rehabilitation, at discharge, and 1 year after discharge. The Umbrella project forms the epidemiological backbone of the larger research program 'Physical strain, work capacity and mechanisms of restoration of mobility in the rehabilitation of persons with a spinal cord injury', which is part of the first Rehabilitation Program of the Netherlands Organization of Research and Development (Zon-Mw).<sup>55</sup> The ultimate aim of the Zon-Mw Program is to improve the research infrastructure of rehabilitation practice. Besides the Umbrella project, the Zon-Mw SCI-rehabilitation program includes other research projects which are often more experimental in set-up.<sup>56</sup>

#### Outline of this thesis

Chapter 2 of this thesis presents a review of the reported level of physical capacity, and discusses the consequences of methodological differences between studies. Subsequently, Chapters 3 to 7 discuss findings based on prospective data of the Umbrella project. Chapter 3 describes changes in the level of physical capacity over time, the associations with age, gender, the level and completeness of the lesion, and the interactions between components of physical capacity (Figure 3). Chapter 4 establishes prognostic models for physical capacity at discharge from inpatient rehabilitation, and for physical capacity 1 year after discharge. The prognostic value of person, lifestyle and lesion-related characteristics are discussed. The occurrence and risk factors of complications are established in Chapter 5. Chapter 6 reports on the interaction between changes in physical capacity over time on the one hand, and the occurrence of complications and the duration of phases of rehabilitation on the other. Chapter 7 describes changes in physical independence over time, and determines whether physical independence and functional status are associated with physical capacity and complications. Finally, Chapter 8 discusses the main findings, methodological strengths and weaknesses, implications for clinical practice and suggestions for future research.

2

# Physical capacity in wheelchair-dependent persons with a spinal cord injury: a critical review of the literature

JA Haisma, LH van der Woude, HJ Stam, MP Bergen, TA Sluis, JB Bussmann In: *Spinal Cord* 2006; 44: 642-652

# Abstract

*Objective:* To assess the level of physical capacity (peak oxygen uptake, peak power output, muscle strength of the upper extremity and respiratory function) in wheelchair dependent persons with a spinal cord injury (SCI).

*Methods:* Pubmed (Medline) search of publications from 1980 onwards. Studies were systematically assessed. Weighted means were calculated for baseline values.

*Results:* In tetraplegia the weighted mean for peak oxygen uptake was 0.89 l/min for the wheelchair exercise test (WCE) and 0.87 l/min for armcranking or hand-cycling (ACE). The peak power output was 26 W (WCE) and 40 W (ACE). In paraplegia the peak oxygen uptake was 2.10 l/min (WCE) and 1.51 l/min (ACE), whereas the peak power output was 74 W (ACE) and 85 W (WCE). In paraplegia muscle strength of the upper extremity and respiratory function were comparable to that in the able-bodied population. In tetraplegia muscle strength varied greatly, and respiratory function was reduced to 55-59% of the predicted values for an age, gender and height matched able-bodied population.

*Conclusions:* Physical capacity is reduced and varies in SCI. The variation between results is caused by population and methodological differences. Standardized measurement of physical capacity is needed to further develop comparative values for clinical practice and rehabilitation research.

# Introduction

Physical capacity can be described as the capacity of the cardiovascular system, muscle groups and the respiratory system to provide a level of physical activity.<sup>17</sup> It is reduced in persons with a spinal cord injury (SCI) by the direct loss of motor control and sympathetic influence below the level of lesion. Additionally, the majority of persons with a SCI will be wheelchair users and dependent on arm work for mobility and activities of daily living. Subsequently, an inactive lifestyle may further reduce physical capacity.<sup>12,36</sup> A low level of physical capacity is associated with a decrease in activity,<sup>12,57</sup> functional status<sup>33,58</sup> and participation.<sup>36,59</sup> This may result in the vicious circle of decreased physical capacity leading to decreased activity and participation, which further reduces physical capacity, and so on. Furthermore, a low level of physical capacity is associated with a relapse in different aspects of health, may contribute to a reduction in quality of life.<sup>9,39,58</sup> Hence, the evaluation of physical capacity can give an indication of the potential level of activity, participation and quality of life.<sup>36,58,59</sup>

Clinicians and rehabilitation researchers need comparative values for different components of physical capacity (i.e. peak oxygen uptake, peak power output, muscle strength of the upper extremity and respiratory function).<sup>12</sup> This will help them set targets in SCI rehabilitation.<sup>49</sup> Additionally, monitoring changes in physical capacity may give an indication of the effectiveness of training and rehabilitation programs.<sup>60</sup> Persons with SCI may become motivated to participate in exercise programs, when they learn about their health status and how it evolves.

Although much research has focused on physical capacity in SCI, the reported level of physical capacity varies greatly. This could be attributed to the disparity in population, methodology and presentation of results, which hamper comparisons and generalization.<sup>9,12,17</sup> Furthermore, few studies have addressed more than one component of physical capacity simultaneously.<sup>12,17,49</sup>

Therefore, the purpose of this study was to integrate evidence on different components of physical capacity in SCI, by critically analyzing absolute values and (in) consistencies in the literature. This way a set of comparative values for components of physical capacity in this population may be obtained. We formulated the following research question: What is the reported level of peak oxygen uptake, peak power output, muscle strength of the upper extremity and respiratory function in persons with SCI who are wheelchair-dependent?

# Methods

#### Data search

To gain insight into the level of physical capacity in SCI, we searched Pubmed (Medline), and included publications from 1980 onwards. SCI-related studies were explored with the Medical Subject Headings (MeSH) 'spinal cord disease' and the terms tetraplegia, quadriplegia or paraplegia. This search was combined with the following MeSH headings: 'physical fitness', 'oxygen consumption', 'exercise test' or 'spirometry'. Then the search was combined with one of the following terms: wheelchair ergometry, endurance, physical capacity, manual muscle testing, handheld dynamometry or myometry. Additionally, we scanned the references in studies, and experts checked the resultant list of studies for completeness. Figure 1 shows the flowchart for the search and selection of studies.



Figure 1 Flowchart for search and selection of studies on physical capacity

#### Data selection

#### Topic-related criteria

To be eligible for inclusion, a study had to specify the following items: (1) Results for subjects with SCI who were wheelchair-dependent. (2) Whether results concerned subjects with a tetraplegia or a paraplegia. (3) One of the following outcome parameters: (i) peak oxygen uptake ( $VO_{2peak}$ ; l/min or ml/min) or peak power output ( $PO_{peak}$ ; W or kpm/min) measured by a wheelchair exercise test (WCE), arm-cranking or hand-cycling (ACE); (ii) manual muscle testing grade (MMT) or handheld dynamometry score (HHD; Newton or kg) for the upper extremity; (iii) forced expiratory flow per second (FEV1) or forced vital capacity (FVC) as absolute values in litres, or as a % of the predicted values for an age, gender and height-matched able-bodied population. (4) The method used.

# Methodological criteria

The resultant 67 studies were assessed on their methodological quality. Because we were interested in baseline values of physical capacity, we abbreviated an established checklist for the evaluation of effect studies.<sup>61,62</sup> If studies specified 7 of the following items they were assumed methodologically fit for inclusion. (1) Inclusion and exclusion criteria. (2) Source of selected population. (3) Description of subjects. (4) Inclusion of  $\geq$  10 subjects with a tetraplegia or  $\geq$  10 subjects with paraplegia. (5) Outcome measure. (6) Details on the measuring protocol. (7) Statistical method. (8) Means and standard deviations. (9) Number of dropouts. (10) Reasons for not completing tests.

Two studies were excluded because they fulfilled only 6 of these criteria. Thirteen studies were excluded, because they concerned doubly reported data.

#### Data extraction

A data extraction form was used to collect information on population characteristics, methods and results. We calculated the weighted means of the combined results.

#### Results

Tables 1-3 summarize the data of the selected studies. Figures 2 and 3 show mean results for the studies and their combined weighted means.

# Peak oxygen uptake and peak power output

Most selected studies assessed subjects with paraplegia. Most subjects were men, who participated in sports, were on average 30 years old, and were assessed > 6 years post-injury. There was variability among the profile used, the starting power output or velocity, and the subsequent increments. For paraplegia the mean  $VO_{2peak}$  during the WCE (Table 1A) ranged from 1.10 to 2.51 l/min (weighted mean 2.10 l/min; Figure 2a) and the mean  $PO_{peak}$  ranged from 46 to 102 W (weighted mean 74 W; Figure 2b). During ACE (Table 1B) the mean  $VO_{2peak}$  ranged from 1.03 to 2.34 l/min (weighted mean 1.51 l/min; Figure 2a) and the mean  $PO_{peak}$  ranged from 66 to 117 W (weighted mean 85 W; Figure 2b). In subjects with a tetraplegia the mean  $VO_{2peak}$  during WCE (Table 1C) ranged from 0.76 to 1.03 l/min (weighted mean 0.89 l/min; Figure 2a) and the mean  $PO_{peak}$  ranged from 0.76 to 0.95 l/min (weighted mean 0.87 l/min; Figure 2a) and the PO\_{peak} ranged from 35 to 43 W (weighted mean 40 W; Figure 2b).



**Figure 2** WCE: wheelchair exercise; ACE: arm-cranking or hand-cycling; PP: paraplegia; TP: tetraplegia.

(a) Mean peak oxygen uptake (l/min) for each of the included studies. Their weighted mean is specified.(b) Mean peak power output (W) for each of the included studies. Their weighted mean is specified.

Table 1A Par	aplegia:	peak	endura	nce capa	icity du	ring wl	neelchair exercise							
author	year	и	age	M/F	C/I	ISL	activity	C/D	E/T	start; increments	HR	$VO_2$	DO	Comment
Bernard <sup>71</sup>	2000	12	30	a.	۵.	a.	athletes	C	Н	3 mins 4 km/h; incr 1 km/h/min	176	2.07		
Bougenot <sup>72</sup>	2003	$\sim$	35	0/2	0//	12	no training	С	Е	15 W; incr 10 W/2min		1.99	74	
Cooper <sup>73</sup>	1992	10	32	10/0	۵.	6	athletes; 8h/week	C	ш	2.23 m/s; incr 0.446 m/s or 2.68 m/s; incr 10W/90sec	183	2.46		
Coutts <sup>74</sup>	1995	18	۵.	27/0	a.	o.	athletes	C	E	40 rpm; incr 20 rpm, until 80, then 10 W/90s		2.47		
Dallmeijer <sup>75</sup>	2004	6	36	0/6	۵.	13	4 h/week	C	L	10 W/min	182	1.79	93	PP = able bodied ACE = WCE
Gass <sup>76</sup>	1995	6	31	0/6	0/6	6	no training	C	F	3.5 km/h; incr 0.5 km/h or 0.5% gradient, until 4%, then speed only	177	1.72		ACE = WCE
Janssen <sup>12</sup>	2002	107	35	96/11	۵.	10	6 h/week	C/D	E/T	incr %PO, fixed W, speed or gradient		2.12	75	Related to activity
${\rm Kerk}^{77}$	1995	9	22	2/4	0/9	4	athletes	С	н	3.1 m/s; incr 0.5 m/s/min		2.51		
Knechtle <sup>78</sup>	2001	6	28	۵.	5/4	۵.	athletes	C	Г	8 km/h 1%; incr 0.5%/2min		2.13		
Martel <sup>79</sup>	1991	20	27	20/0	۵.	10	5 h/week	C	ы	5 W, incr 10 W/min	190	1.9	74	PO: ACE>WCE VO <sub>2</sub> :ACE=WCE
Rasche <sup>80</sup>	1993	9	26	6/0	6/0	~	athletes	C/D	T	D: 3 mins, incr 10-15 W; C: 50%max; incr 10-15 W	198 187	D: 2.13 C: 2.18	$100 \\ 104$	VO <sub>2</sub> , PO: D=C.
Schmid <sup>81</sup>	1998	30	35	30/0	30/0	11	۵.	С	Е	20 W; incr 10 W/3min	177	2.08	73	
Schmid <sup>82</sup>	1998	10	30	0/10	10/0	12	sedentary	C	Е	20 W; incr 10 W/3min	180	1.09	46	
Veeger <sup>83</sup>	1991	45	34	40/8	۵.	a.	athletes	C	T	2 km/h or 1%/min	178	2.19	72	VO <sub>2:</sub> F <m< td=""></m<>
Vinet <sup>84</sup>	1996	~	29	0/2	4/3	>2	6 h/week	C	L	4 km/hr; incr 1 km/h/min	176	2.46		
M/F: ratio male (bpm); VO <sub>2</sub> : p	e/female eak oxyε	; C/I: c şen upt	omplete ake (1/n	/incompl- nin); PO:	ete; TSI: peak po	time sir wer out	nce injury (years); C/ put (Watt); ACE: arm	D: cont n-crankii	inuous 1g exer	or discontinuous protocol; E/T cise; WCE: wheelchair exercise	l: ergor e; PP: p	netry or tre araplegia.	admill;	HR: peak heart rate

21

TADIC TD La													
author	year	и	age	M/F	C/I	IST	activity C/1	D start; increments	Ł	IR	$VO_2$	DO	Comment
Dallmeijer <sup>75</sup>	2004	6	36	0/6	a.	13	4 h/week C	incr 10 W/min	1	88	1.88	117	PP = able bodied ACE = WCE
Ellenberg <sup>85</sup>	1989	10	30	7/3	7/3	0,1	no training C	25 W; incr 25 W/3mi	ins 1	70	1.06	66	
$Gass^{76}$	1995	6	31	0/6	۵.	6	ADL only C	20 W; incr 5 W/30s	1	4	1.65		ACE = WCE
Hooker <sup>18</sup>	1993	27	25	27/0	27/0	5	sedentary C	1min; incr 8.2 or 16.5	3 W 1	71	1.26	76	PP < able bodied
Hooker <sup>86</sup>	1993	6	33	0/6	4/5	10	no training C	3 mins; incr 8.2 W/n	1 1ir	20	1.14	70	
Hopman <sup>87</sup>	2004	9	33	0/9	5/1	۵.	4 h/week C	incr 3, 5 or 10 W/mi	n 1	79	1.68		
Hopman <sup>88</sup>	1998	4	28	4/0	3/1	$\sim$	D	2-10 W/min	1	85	1.85	97	
Jacobs <sup>89</sup>	2001	10	39	10/0	10/0	$\sim$	D	65 W; incr 16 W/min	_		1.45	107	
Knechtle <sup>90</sup>	2003	$\sim$	42	۵.	0//	۵.	athletes C	60 W; incr 20 W/2mi	sui		2.34		
Martel <sup>79</sup>	1991	20	27	20/0	۵.	10	5 h/week C	5 W, incr 10 W/min	1	06	1.88	97	PO: ACE>WCE VO <sub>2</sub> :ACE=WCE
Mossberg <sup>91</sup>	1999	11	35	10/1	۵.	6	Ω ٤	20 W; incr 10 W/2mi	I.	78	1.38		Asynchronous = synchronous.
Steinberg <sup>92</sup>	2000	26	30	26/0	26/0	9	athletes and sedentary C	25 W; incr 12.5 W/m	in 1	4	1.44	75	
Yamasaki <sup>93</sup>	1998	22	38	22/0	22/0	13	athletes and sedentary C	0 W; incr 5 W/min	1	09	1.41		Related to activity

uptake (I/min); PO: peak power output (Watt); ACE: arm-cranking exercise; WCE: wheelchair exercise; PP: paraplegia.

Chapter 2

uthor	year	и	age	M/F	C/I	ISI	activity	C/D	E/T	start; increments	HR	$VO_2$	Od	Comment
coutts <sup>74</sup>	1995	6	۵.	0/6	a.	a.	athletes	С	н	20 rpm; incr 10/min		0.95		
)allmeijer <sup>94</sup>	1997	20	33	۵.	14/(	~	athletes and sedentary	U	н	Incr 10% PO <sub>peak</sub> /min		0.85	21	
iass <sup>95</sup>	1980	6	34	0/6	8/1	12	no training	С	H	2% gradient or 0.5 km/h/min	125	0.76		
anssen <sup>12</sup>	2002	59	35	50/9	۵.	$\sim$	4 h/week	C/D	E/T	incr %PO, fixed W, speed or gradient		06.0	25	Related to activity
chmid <sup>81</sup>	1998	20	34	20/0	20/(	) 11	α.	С	Е	10 W; incr 5 W/3mins	110	1.03	33	PO: TP <pp< td=""></pp<>
					20	1004		, 1						
uthor	year	и	age	M/F	C/I	ISI	activity	C/D		increments	HR	$VO_2$	PO	Comment
Hopman <sup>87</sup>	2004	9	26	0/9	5/1	۵.	3-5 h/week	С		3-5 W/min		0.86		
lopman <sup>88</sup>	1998	Ś	34	5/0	5/0	11		С		2-10 W/min	118	0.87	42	
lopman <sup>96</sup>	1996	21	32	18/3	16/5	$\infty$	athletes and sedentary	C		20% PO <sub>peak</sub>		0.78	35	Related to activity
asko <sup>97</sup>	1991	24	29	24/0	۵.	6	ς.	C		4 W/min	123	0.95	43	Related to increments

/F: ratio male/female; C/I: pm); VO <sub>2</sub> : peak oxygen uf
--

Table 2 Mus	cle strei	ngth o	f the u	pper exti	emity	in subj	jects wi	h a tetraplegia	
author	year	и	age	M/F	C/I	ISL	H/M	Strength upper extremity	Comment
Beninato <sup>54</sup>	2004	20	37	16/4	13/7	a.	Μ	Shoulder flex grade 3.7/ext 3.6; Elbow flex 4.4/ext 2.1; wrist flex 1.5/ext 3.3	MMT 20-point scale
Bryden <sup>68</sup>	2004	43	32	36/7	43/0	9	Μ	Elbow ext: 60/74 grade 0; 14/74 grade 1-2 Elbow flex: median 4+; range 3-5	MMT 20-point scale.
Burns <sup>64</sup>	2005	19	54	19/0	9/10	a.	H/M	Elbow flex 2/7 grade 3; 5/7 grade 4 ext 6/12 grade 3; 6/12 grade 4 Make-test flex 7.3 kg: ext 8.0 kg Break-test flex 9.0 kg; ext 11.1 kg	Break test ≻make test
Fujiwara <sup>67</sup>	1999	14	31	12/2	14/0	1.3	Μ	ASIA motor score: 20/50 Shoulder score: 25/30	
Herbison <sup>65</sup>	1996	88	34	78/10	۵.	<u>م.</u>	H/W	Elbow flexors: 26/176 grade 3.5 (↔ 4 kg); 47/176 grade 4.0 (↔ 5.1 kg); 50/176 grade 4.5 (↔ 6.7 kg); 53/176 grade 5 (↔ 9.0 kg).	MMT 10-point scale. HHD more sensitive to change
Hjeltnes <sup>69</sup>	1998	10	25	10/0	10/0	0.3	Μ	ASIA motor score: 19/50	MMT 5-point scale. Related to training.
Marino <sup>70</sup>	1995	50	α.	47/3	50/0	1	М	ASIA: mean 19/50; median 16/50	MMT 5-point scale.
May <sup>63</sup>	1997	12	27	10/2	10/2	Ś	Н	Internal rotation shoulder 14.8 kg External rotation 11.7 kg	
May <sup>63*</sup>	1997	11	28	10/1	8/3	8	Η	Internal rotation shoulder 30.6 kg; external rotation 22.0 kg	Paraplegia
Schwartz <sup>66</sup>	1992	122	۵.	122/0	α.	0.2	H/M	MMT grade elbow flex 5; wrist extensor grade 4. HHD: elbow flex left 8.0 kg, right 9.2 kg; Wrist ext left 6.3, right 6.6 kg	MMT 10-point scale; HHD more sensitive to change
*May <sup>63</sup> specifi muscle testing (max = 50); Sl	es result 3 or hand houlder	s of the 1-held score: 5	ose with dynamc summat	a parapli metry; ↔ ion MMT	egia and : corres of scap	d those sponds ula abd	with a to; ASIA luctors,	etraplegia. M/F: ratio male/female; TSI: time since injury (years); C/I = summation of MMT grade elbow flexion, wrist extension, elbow e. shoulder adductors and extensors (max = 30).	: complete/incomplete; M/H: manual xtension, finger flexion, finger spread

Chapter 2

Table 3 Resp	iratory f	unctic	on follo	owing SC.								
author	year	и	age	M/F	C/I	smoking	IST	FEV1	FEV1%	FVC	FVC%	Comment
Data on subje	cts with	Parapl	legia									
Alemenoff <sup>98</sup>	1995	81	54	۵.	27/54	53N; 28C	23	3.18	90	4.20	86	Related to smoking.
$\operatorname{Bernard}^{71}$	2000	12	30	α.	۵.	۵.	۵.	4.09	98			
$\mathrm{Kerk}^{77}$	1995	9	22	2/4	6/0	۵.	4			3.0		
Linn <sup>99</sup>	2000	41	40	19/22	41/0	Z	14		89		85	Related to smoking
Schilero <sup>100</sup>	2005	15	49	15/0	ο.	6N; 7F; 2C	16	3.17	86	4.18	84	TP <pp; bronchodilatator="" related="" td="" to="" use.<=""></pp;>
Schilero <sup>101</sup>	2004	١ſ	40	α.	2/3	1N; 3F; 1C	19	3.50		4.26		
Silva <sup>102</sup>	1998	12	31	12/0	12/0	Z	۵.	3.62	84	4.05	81	PP <able ace="" bodied;="" related="" td="" to="" training<=""></able>
Data on subje	cts with	Tetrap	olegia									
Almenoff <sup>98</sup>	1995	84	46	۵.	12/72	55N; 29C	16	2.40	64	2.94	59	
Almenoff <sup>103</sup>	1995	25	43	25/0	6/19	10C; 15N	11	2.07	55	2.46	50	
$Gass^{95}$	1980	6	34	α.	8/1	3C	12	2.57	80	2.38		
Grimm <sup>104</sup>	2000	32	42	α.	12/20	14N; 11F; 7C	14	2.47	62	3.07	58	
Liaw <sup>105</sup>	2000	20	34	16/4	20/0	2C	0.2	1.20	40	1.4	37	Related to training
Linn <sup>99</sup>	2000	35	40	16/19	35/0	Z	14		57		52	
Lougheed <sup>106</sup>	2001	9	32	5/1	0/9	2N; 2F; 2C	$\sim$	2.27	57	2.67	53	
Rutchik <sup>28</sup>	1998	10	36	10/0	4/6	4N; 5F; 1C	6	2.25	54	2.81	51	Related to training
Schilero <sup>100</sup>	2005	15	42	15/0	۵.	6N; 7F; 2C	13	2.29	56	2.91	55	TP <pp; <math="" tp="">&lt; able bodied</pp;>
Schilero <sup>101</sup>	2004	Ś	45	α.	2/3	3F; 2C	17	2.28		2.97		Related to bronchodilatator use
Spungen <sup>107</sup>	1999	10	41	10/0	4/6	9N; 1C	16	2.23	56	2.79	53	Related to steroid use
Spungen <sup>108</sup>	1993	34	45	34/0	34/0	12N; 14F; 8C	12	2.97		2.12		
Walker <sup>109</sup>	1989	15	28	11/4	0/15	Z	~2			2.60	61	Related to training
Wang <sup>110</sup>	2002	15	41	۵.	15/0	۵.	4	1.83	61	2.16	59	Related to training
M/F: ratio mal second (liters) PP: paraplegia	e/female; ; FEV1%:	; TSI: t forcec	ime sir 1 expira	nce injury ( atory flow	(years); C in one se	/I: complete/inco cond as % of pre/	mplete; dicted;	; smoking FVC: forc	history N: n ed vital capa	ever; F: fori ıcity (liters);	ner; C: curren FVC% forced	; FEV1: absolute forced expiratory flow in one vital capacity as % of predicted; TP: tetraplegia;

Physical capacity following SCI: a literature review

#### Muscle strength upper extremity

Most selected studies assessed subjects with a tetraplegia. Most subjects were men, who were on average 30 years old. Time since injury ranged from 1 month to 6 years. In paraplegia the mean strength of the shoulder internal rotators was 30.6 kg and of the external rotators 22.0 kg.<sup>63</sup> For subjects with tetraplegia these values were 14.8 and 11.7 kg, respectively.<sup>63</sup> The elbow flexion in subjects with tetraplegia ranged from 4.0 to 9.2 kg.<sup>64-66</sup> The mean strength for wrist extension was 6.5 kg.<sup>64</sup> Elbow extension scored between 8.0 and 11.1 kg.<sup>64</sup>

MMT grades were used varyingly. One study reported a mean shoulder flexion of grade 3.7 and a mean shoulder extension of 3.6.<sup>54</sup> Summation of the score of three muscles in both shoulders resulted in a score of 25 out of 30.<sup>67</sup> Elbow flexion ranged from a grade 2 to 5.<sup>54,64-66,68</sup> Wrist extension ranged from 3.3 to 4.<sup>54,66</sup> Elbow extension ranged from 0 to 2.<sup>64,68</sup> The mean wrist flexion was 1.5.<sup>54</sup> The mean motor score of the 5 key muscles of both upper extremities was 19 or 20 out of 50.<sup>67,69,70</sup>

# Respiratory function

Most selected studies assessed subjects with a tetraplegia. Time since injury usually exceeded 10 years. In paraplegia mean FEV1 ranged from 86-98% of the predicted value for an age, gender and height matched able-bodied population (weighted mean 90%; Figure 3). The mean FVC ranged from 81-86% of the predicted value (weighted mean 85%). In tetraplegia mean FEV1 ranged from 40-80% of the predicted value (weighted mean 59%). The mean FVC ranged from 37-61% of the predicted value (weighted mean 55%).



**Figure 3** Mean respiratory function (as % of predicted values for an age, gender and height matched able-bodied population) for included studies; PP: paraplegia; TP: tetraplegia; FEV1: forced expiratory flow in one second; FVC: forced vital capacity. Weighted mean FEV1 or FVC for included studies is specified.

# Discussion

# Level of peak oxygen uptake and peak power output

Relatively low values for VO<sub>2peak</sub> and PO<sub>peak</sub> were a common finding and can be attributed to the dependency on arm exercise, the extent of paralysis, the reduced sympathetic control and the relative inactivity, which compromise physical capacity in SCL<sup>18,73,90,111</sup> ACE in sedentary able-bodied persons is suggested to induce an oxygen uptake of 70% of the oxygen uptake that can be reached during a treadmill running test.<sup>112</sup> Assuming the 70% ratio applies to persons with a paraplegia, the mean of 1.51 l/min corresponds to 2.16 l/min. This is comparatively low, especially because daily use of a wheelchair may induce a training effect in persons with a paraplegia, and therefore, they may even be compared to an able-bodied population practised in arm-exercise.<sup>73,75</sup> <sup>113</sup> It indicates that factors like paralysis of lower limbs, altered autonomic control and inactivity, do compromise physical capacity in paraplegia.

In paraplegia, the weighted mean VO<sub>2peak</sub> in WCE was higher than during ACE. This is in agreement with another study, which suggested that during WCE a larger muscle mass is activated, because muscles involved in stabilisation of the trunk, and in the (un)coupling of the hand to the rim will be used.<sup>114 79</sup> However, this is inconsistent with studies that assessed the same population alternately with both methods (Figure 4). These studies showed no significant differences in VO<sub>2peak</sub>,<sup>76,79,115,116</sup> or an even higher VO<sub>2peak</sub> during ACE than during WCE.<sup>75,115</sup> The inconsistency could be attributed to the fact that the present review compared results from different populations. The subjects studied during ACE were generally less active, and tested sooner after injury than subjects studied during WCE. Persons with longstanding SCI generally have a higher VO<sub>2peak</sub> than persons with a recent SCI.<sup>49,117</sup> Protocol



**Figure 4** Results of four studies, which compared outcome during WCE (wheelchair exercise) to outcome of ACE (arm-cranking or hand-cycling). Mean peak power output (PO<sub>peak</sub>) and peak oxygen uptake (VO<sub>peak</sub>) for these four studies are indicated. Their weighted mean is specified.

differences could also explain the differences found;<sup>75,78,97,118</sup> overall, the ACE started at a higher power output, and had greater subsequent increments. It is suggested that greater increments may underestimate peak values.<sup>97</sup>

The weighted mean  $PO_{peak}$  was lower in WCE than during ACE. This is consistent with other studies that assessed the same population alternately with both methods (Figure 4),<sup>75,79,115,116</sup> and suggests that the discontinuous and more complex movement pattern of wheelchair propulsion (as opposed to the continuous movement in arm-cranking), may limit its efficiency and lead to a lower  $PO_{peak}$ . The combination of the relatively high  $VO_{2peak}$  with the low  $PO_{peak}$ , supports the finding that wheelchair propulsion is mechanically less efficient than armcranking.<sup>79</sup>

# Measuring peak power output and peak oxygen uptake

VO<sub>2peak</sub> and PO<sub>peak</sub> have proven valid and sensitive outcome measures for the assessment of physical capacity,<sup>17</sup> and should be included in the follow-up of persons with a SCI. The outcome of the maximal exercise tests were expressed as peak levels of oxygen uptake and power output (instead of maximal levels), because some subjects may have been able to activate a larger muscle mass, either through activation of the lower limbs (e.g. in those with incomplete lesion), or without the restraints from overuse and fatigue of the upper extremities.<sup>25,76,88</sup> The large variability in results found in paraplegia may be attributed to differences in measurement protocol<sup>75,78,118</sup> and study population.<sup>12</sup> Because of lack of homogeneity, no consistent conclusions on the influence of a particular protocol can be drawn.

#### *Muscle strength of the upper extremity*

The HHD score for shoulder strength in subjects with a paraplegia compared favourably to an age and gender matched able-bodied population.<sup>119</sup> This seems plausible, as in paraplegia muscle strength may be enhanced by daily use of the upper extremity.<sup>73</sup> However, in tetraplegia, muscle strength additionally depends on the level and completeness of injury, neurological recovery and spasticity. In subjects with high cervical lesions shoulder strength was reduced to 50% of values found in subjects with a paraplegia, or in the able-bodied population.<sup>63,119</sup> The strength of elbow flexion and wrist extension was reduced to 15-30% of that in the able-bodied population.<sup>64,66,119</sup> However, elbow extension showed relatively greater strength (30-50% of that in the able-bodied population).<sup>64,119</sup> The range of reported strength within the same muscle group, may be attributed to subjects with incomplete lesions, who have a haphazard innervation pattern. Additionally, daily wheelchair use may have had a training effect in some subjects, and may cause overuse in others.

# Measuring muscle strength of the upper extremity

Most studies incorporated the widely recognized and practical MMT and this was the motivation for choosing the MMT score as an outcome measure. However, the different scales and summations used hamper comparison between studies. The MMT score has other limitations, when compared to the HHD. Firstly, it is an ordinal scale, and therefore summations are not very meaningful. Secondly, it is a subjective score, probably sensitive to observer-bias.<sup>21</sup> Thirdly, it is limited in the ability to identify change for grades 4-5, and the registration of recovery is restricted by a ceiling-effect.<sup>21,65,66</sup> <sup>120,121</sup> Fourthly, it seems less valid, because it has a limited correlation with isokinetic dynamometry, which is often regarded the gold standard for the assessment of muscle strength, but is not manageable in use.<sup>21,63,122</sup> The HHD score has shown to be valid, and has a good reliability in SCI, both with experienced and inexperienced examiners.<sup>63-65</sup>

Muscle strength is an important component of physical capacity, and is related to functioning.<sup>123</sup> Because outcome measures need to be valid, sensitive to change and reliable, consensus needs to be reached on how to assess strength. HHD seems a valuable tool for the evaluation of muscle strength in some patients with SCI.

# Level of respiratory function

In subjects with a tetraplegia the respiratory function was greatly reduced as compared to an age, gender and height matched able-bodied population, whereas in paraplegia the scores were relatively normal. This is consistent with other reports, which suggest that the level of lesion is inversely correlated with respiratory function.<sup>9,22</sup> When the thoracic and lumbar segments are injured, muscles of expiration are affected, whilst injury to the upper cervical cord additionally affects muscles of inspiration.<sup>9,22</sup> Furthermore, increased inactivity in tetraplegia may add to the reduced respiratory function.<sup>95</sup> Both FEV1 and FVC were reduced, hence the ratio FEV1/FVC remained stable, which suggests the respiratory problems are mostly restrictive in nature.<sup>98</sup> However, it is suggested that with loss of the sympathetic influence from the upper six thoracic segments, the parasympathetic bronchoconstriction remains unopposed.<sup>100,103,104</sup> This, together with airway obstruction following possible mucus collection, may result in additional obstructive problems.<sup>108</sup>

The variability found in outcome in tetraplegia may be attributed to population differences. Studies that included subjects with incomplete lesions,<sup>98,99,124</sup> or those who were able to perform a maximal exercise test,<sup>95</sup> were expected to have higher scores. Smoking was related to a reduced respiratory function in subjects with a paraplegia; surprisingly, however, these studies reported no significant relation between smoking and respiratory function in tetraplegia.<sup>98,99,124</sup> Therefore, on the

basis of these studies, smoking cannot be held responsible for population-related differences in outcome in tetraplegia.

#### Measuring respiratory function

The respiratory function was expressed as a % of predicted values from an age, gender and height-matched able-bodied suitable reference population. However, because level of lesion is an important determinant of respiratory function, one would ideally want to correct for this, but it would require a large comparative database to make a valid correction. All selected studies used the American Thoracic Society (ATS) criteria for accepting individual spirometry results. This may lead to biased results, because the most impaired subjects are not able to meet these criteria. It has been suggested that when the criteria are modified, reproducibility can be guaranteed without loss of results from these subjects.<sup>125</sup> Respiratory complications are responsible for about 25% of deaths following a SCI,<sup>126</sup> and 68% of persons with a SCI experience respiratory complaints.<sup>127</sup> Improved respiratory function may contribute to the prevention of complications; therefore, evaluation of this function is important in SCI rehabilitation.

# Limitations

We aimed at integrating evidence on different aspects of physical capacity. However, the cardiovascular function was not investigated for two reasons. Firstly, stroke volume is not easily measured in rehabilitation practice. Secondly, peak heart-rate is not a valid measure of physical capacity in SCI, and is not sensitive to change during training.<sup>17,25,88</sup> Reasons for this may be that upper body exercise is limited by local fatigue and overuse (rather than cardiovascular strain), and that peak heart-rate varies too much in persons with a tetraplegia.<sup>75,76,79,115</sup>

Overall a positive selection is shown which will bias the results. Firstly, most studies focused on male athletes.<sup>12,69,102</sup> Secondly, only those able to perform the tests were included, possibly excluding subjects with high cervical lesions. Thirdly, only those willing to participate were included. Hence, motivation and state of mood could influence both the selected population and performance during the tests. Due to lack of homogeneity among the included studies, the integration of findings did not provide consistent and comparative values.

#### Conclusions

The level of physical capacity is reduced and varies in persons with a SCI. The variation between results is caused by population and methodological differences. To allow interpretation and comparison of results, researchers should meticulously describe the population and the methods used. Standardized assessment of physical capacity in clinical practice and rehabilitation research is needed for the effective prediction and evaluation of progress in future. The present study provides suggestions for the parameters to be determined and methods to be used. Additionally, it provides a descriptive database for physical capacity in persons with a SCI and, with caution, these data may be used as reference material in rehabilitation and training.



# Changes in physical capacity during and after inpatient rehabilitation in subjects with a spinal cord injury

JA Haisma, JB Bussmann, HJ Stam, TA Sluis, MP Bergen, AJ Dallmeijer, S de Groot, LH van der Woude In: *Archives of Physical Medicine and Rehabilitation* 2006; 87: 741-747

# Abstract

*Objective:* To assess changes in components of physical capacity and its determinants during and after inpatient rehabilitation in persons with a spinal cord injury (SCI). To investigate the association among components of physical capacity.

*Methods:* Prospective multicenter cohort study. At the start of active rehabilitation (n= 186), 3 months later, at discharge, and 1 year after discharge (n= 123), we measured the peak aerobic power output (POpeak), peak oxygen uptake (Vo<sub>2</sub>peak), muscle strength of the upper extremity (manual muscle test, handheld dynamometry) and respiratory function (forced expiratory flow per second, forced vital capacity).

*Results:* Random coefficient analysis demonstrated that the POpeak, Vo<sub>2</sub>peak, strength, and respiratory function improved during inpatient rehabilitation, and that Vo<sub>2</sub>peak, strength, and respiratory function continued to improve after discharge. Age, sex, and level and completeness of lesion were determinants of the (change in) components of physical capacity.

*Conclusions:* Physical capacity improves during inpatient rehabilitation, and some components continue to improve after discharge. Subpopulations have a different level of (change in) physical capacity. The components of physical capacity are related; intervention studies are needed to confirm whether training 1 component could improve another component.

# Introduction

Physical capacity is the combined ability of the cardiovascular, the respiratory and the musculoskeletal systems to attain a certain level of activity.<sup>17</sup> The different components of physical capacity (eg, power output, oxygen uptake, muscle strength, respiratory function) are related.<sup>102,128,129</sup> Additionally, the components are influenced by personal and lesion related factors and by training.<sup>9,12,89,102</sup> <sup>29</sup>

Physical capacity is reduced in persons with a spinal cord injury (SCI) through muscle weakness, loss of autonomic control below the level of injury,<sup>88,111,130</sup> reduced activity, and subsequent changes in metabolic and vascular function.<sup>12,24,31,130</sup> A reduced physical capacity is an important determinant of the health status of persons with an SCI, because it exposes them to an increased risk of complications,<sup>9,31</sup> and is related to a reduced level of functioning and quality of life.<sup>9,31,33,59</sup> Therefore, an important goal in rehabilitation practice is to reverse the debilitative cycle of: a reduced physical capacity leading to reduced activity and functioning, which in turn further reduces physical capacity, and so on.<sup>31,33</sup> This reversal may be achieved by improving the physical capacity through rehabilitation training programs.

Knowledge on the change in physical capacity during inpatient rehabilitation is a prerequisite for the development of optimal rehabilitation programs, and for setting realistic individual rehabilitation goals. After discharge there are less training opportunities and more activities of daily life (ADLs) to perform. Because physical capacity is related to the performance of ADLs, it is especially important to determine whether persons with an SCI can maintain their level of physical capacity after discharge.<sup>35,57,58</sup>

Although much research has focused on physical capacity in SCI this was often limited by the particular study population or the design. Most research has focused on one component of physical capacity (eg, either peak oxygen uptake [Vo<sub>2</sub>peak], or muscle strength, or respiratory function). However, more than one component of physical capacity should be studied simultaneously to analyze the relation between the different components.<sup>17</sup> Follow-up studies have found significant differences between admission and discharge scores,<sup>17,49</sup> but because it was suggested that most changes occur during the first phase of rehabilitation,<sup>131</sup> it is important to investigate the change over shorter periods of time. Because there is some debate as to whether the level of activity during daily life is adequate enough to maintain physical capacity after discharge,<sup>40</sup> a longitudinal study is needed to gain more insight into the changes after inpatient rehabilitation. Age, sex, and level and completeness of the lesion were found to be factors related to the physical capacity in cross-sectional studies,<sup>9,12</sup> and longitudinal data will give insight into the influence of these factors on the change in physical capacity.

The main purpose of this study was to determine changes in the components of physical capacity (ie, the peak aerobic power output [POpeak], Vo<sub>2</sub>peak, muscle strength of the upper extremity and respiratory function) during inpatient rehabilitation and the year after discharge in persons with an SCI, who were wheelchair-dependent. Furthermore, we assessed the association between personal and lesion characteristics, and the (change in) physical capacity over time. Finally, the relations between the different components of physical capacity were studied.

#### Methods

#### Participants

The present study was part of the Dutch research program "Physical strain, work capacity and mechanisms of restoration of mobility in the rehabilitation of persons with a spinal cord injury." Subjects with a SCI who were in their initial inpatient rehabilitation were recruited from 8 participating specialized rehabilitation centres. Subjects were eligible for inclusion if they were between 18 and 65 years of age, were wheelchair-dependent, had sufficient comprehension of the Dutch language to understand the purpose of the study and testing methods, and did not have a progressive disease or a psychiatric condition interfering with constructive participation. Subjects were excluded from the study if they had cardiovascular contraindications for exercise, a resting diastolic blood pressure greater than 90mmHg, or a systolic blood pressure greater than 180mmHg.<sup>132</sup> Additionally, the maximal exercise test was contraindicated in the presence of compromising complaints of the musculoskeletal system. Within two weeks of admission, eligible subjects were informed about the study by a physician and invited to participate. Subjects had one week to consider participating, after which a physician or a nurse not involved in the research program requested consent. The Medical Ethics Committee approved the experimental protocol, and all subjects gave a written informed consent before participating.

#### Design

Subjects admitted to rehabilitation from 1999 to 2004 were included if they met eligibility criteria. They were assessed at 4 points in time (called measurement times): at the start of active inpatient rehabilitation when the subject was able to sit in a wheelchair for at least 3 hours (t1), 3 months later (t2), at discharge (t3), and 1 year after discharge (t4). If the subject was discharged within 1 month after t2, the performed assessment at t2 was considered a "discharge" assessment and, hence, was included in the analysis
of physical capacity at t3. If a subject no longer relied on a wheelchair for mobility the previously collected data were analyzed, but no new data were collected.

Subject characteristics and level and completeness of the lesion and physical capacity scores were determined according to a standardized procedure. Tetraplegia was defined as a lesion at or above the T1 segment, and paraplegia as a lesion below the T1 segment. A complete lesion was defined as motor complete, that is, American Spinal Injury Association (ASIA) grade A or B.<sup>1</sup> An incomplete lesion was defined as ASIA grade C or D.<sup>1</sup> In each of the 8 rehabilitation centers a trained research assistant was responsible for the data collection. At each measurement the same testing equipment (ie, research wheelchair, treadmill,<sup>a</sup> dynamometer,<sup>b</sup> Oxycon Delta<sup>c</sup>) was used after standardized calibration.

#### Procedure

*POpeak and Vo2peak.* To determine POpeak and Vo<sub>2</sub>peak subjects performed a graded maximal wheelchair exercise test on a motor-driven treadmill.<sup>a</sup> Prior to testing subjects were asked to consume a light meal only, refrain from smoking, drinking coffee or alcohol two hours before testing, and to void their bladder The testing protocol and equipment has been described previously by Kilkens et al.<sup>133</sup> For each subject, and at every measurement time, a drag test was used to allow the calculation of the power output for the wheelchair-user system on the treadmill at increasing inclinations.<sup>27</sup>

Subjects performed 2 blocks of submaximal exercise of 3 minutes each, separated by a 2-minutes rest. The inclination of the treadmill was set horizontal during the first block and at .36° during the second block. The velocity of the treadmill was set at 2km/h for subjects with tetraplegia and at 4km/h for subjects with paraplegia. Sometimes (eg, in subjects with a low cervical lesion) a protocol with a velocity of 3km/h was chosen. After 2 minutes of rest, the maximal exercise test followed at the same constant velocity, and the inclination was increased .36° every minute. The test was terminated when the subject was exhausted or could no longer maintain the speed of the treadmill. The Vo<sub>2</sub>peak (in L/min) was defined as the highest value of oxygen consumption recorded during 30 seconds. The POpeak (in watts) was defined as the power output at the highest inclination that could be maintained for at least 30 seconds.

Strength of the upper extremity. To determine the strength of the upper extremity, the shoulder abductors, internal and external rotators, elbow flexors and extensors, and wrist extensors were tested with the manual muscle test (MMT) in both arms. This test was performed in standardized positions, in which subjects performed a movement either with or without gravity, or against resistance.<sup>134</sup> The strength was rated on a scale ranging from 0 to 5.<sup>134</sup> Summing the scores of the 12 muscle groups gave an MMT sum score (maximum, 60).

The muscle groups (with exception of the wrist extensors) that scored 3 or greater on the MMT were tested with handheld dynamometry<sup>b</sup> (HHD) according to a standardized protocol.<sup>135</sup> The break test was used: the subject exerted a maximal force against a dynamometer and the examiner applied sufficient resistance to just overcome the force exerted by the subject.<sup>119</sup> The maximum force (in newtons) of the 10 muscle groups was summated. Only when an MMT or HHD score was available for all muscle groups (ie, 12 or 10, respectively) was a sum score calculated, and included in the estimation of the muscle strength at that measurement time.

*Respiratory function.* Flow-volume curves were made with the Oxycon Delta<sup>c</sup> to assess respiratory function. Three repeated curves were made, and if the resultant curve did not have its characteristic shape an extra measurement was done.<sup>136</sup> The forced vital capacity (FVC) (in liters) and forced expiratory flow per second (FEV<sub>1</sub>) (in liters) for each subject were additionally expressed as a percentage of what this subject was expected to score in comparison with an age, sex, and height-matched able-bodied population.<sup>137</sup>

#### Analyses

We used random coefficient analysis<sup>d</sup> to analyze the change in physical capacity over time and its determinants.<sup>138,139</sup> An important advantage is that the method takes into account the dependency of repeated assessments within 1 subject and within 1 rehabilitation center. Additionally, the analyses can be carried out with missing values.<sup>138,139</sup> This offers the advantage (compared, eg, with repeated-measurements analysis of variance) that more subjects can be included, which gives a more realistic representation of the group performance at each measurement time. Thus, we included subjects in the analyses when 1 or more components of physical capacity at 1 or more measurement times could be determined, and we could adequately assess the change in physical capacity over time with varying group composition.<sup>138,139</sup>

*Change in the physical capacity during and after inpatient rebabilitation.* A basic regression model was made for the change in each component of physical capacity. Time was not included in the model as a single continuous variable, but as a set of dummy variables with the physical capacity at discharge chosen as the reference variable. In this analysis, the physical capacity at discharge was estimated by the intercept. The differences between the physical capacity at discharge and the physical capacity at the other measurement times were estimated by the coefficients for these measurement times. Estimates of the physical capacity at the other measurement times were obtained by adding these coefficients to the intercept. To model had to include all three dummy variables to permit a valid assessment of the physical capacity.

Relation between the (change in) physical capacity and personal and lesion characteristics. To determine whether the variables age, sex, and level and completeness of the lesion were related to the (change in) physical capacity over time, these independent variables and their interaction terms with the dummy variables were alternately added to the basic model. Then, the variables and interaction terms with a *P* value of .10 or less were selected and simultaneously added to the basic model. Through backward elimination the nonsignificant variables (P>.05) were removed. To allow a valid analysis of the relation between an independent variable and the change in physical capacity over time, we had to include all its interaction terms even if the relation proved significant over only 1 time interval.

Relation between the different components of physical capacity. To investigate whether the components of physical capacity were related, the association between the Vo<sub>2</sub>peak, muscle strength and respiratory function, and the (change in) POpeak was first analyzed, and then the association between muscle strength and respiratory function, and the (change in) Vo<sub>2</sub>peak was analyzed separately. In both instances, we controlled for those independent variables that significantly contributed to the POpeak and Vo<sub>2</sub>peak in the respective preceding model, by including these variables in the model. We made models for both the POpeak and Vo<sub>2</sub>peak according to the procedure described in the above paragraph. This resulted in two models with independent variables and the components of physical capacity that were significantly related to the POpeak or Vo<sub>2</sub>peak.

#### Results

#### Participants

At the start of active rehabilitation the group consisted of 186 subjects, and a year after discharge there were 123 subjects. During the course of the study 58 subjects dropped out at some point: 16 subjects became wheelchair-independent, 25 refused further participation, 9 subjects were irretraceable, and 8 subjects died. There were several reasons for not collecting data at 1 particular measurement time: 44 subjects were discharged within 3 months after admission, hence they did not perform a t2 measurement; in 20 instances the neurologic deficit was too large to be able to perform the test; on 14 occasions subjects had a halo traction or other fixation; in 6 instances other pathology (eg, infection, pressure ulcer) prevented assessment, and technical problems prevented assessment on 8 occasions. Table 1 gives the group sizes, means and standard deviations (SDs) for the subject characteristics, lesion characteristics and

Characteristic and		Star	t	Thr	ee Months	Disc	harge	Yea	r After Discharge
score		n	Mean ± SD	n	Mean ± SD	n	Mean ± SD	п	Mean ± SD
Age (y)	TP	75	39±13	66	39±13	66	40±13	42	39±13
	PP	110	41±15	72	41±15	102	41±15	81	42±14
Sex (% men)	TP	76	74	66	73	66	76	42	74
	PP	110	75	72	78	101	74	81	74
Height (m)	TP	75	1.78±0.09	66	1.77±0.10	65	1.77±0.09	42	1.76±0.09
	PP	108	$1.78 \pm 0.10$	70	1.79±0.08	100	$1.79 \pm 0.10$	80	1.78±0.09
Body mass (kg)	TP	75	70±14	65	72±14	64	74±15	42	77±17
	PP	110	73±14	71	74±14	101	75±14	81	77±17
Time since injury (d)	TP	75	108±67	65	212±76	65	388±176	42	787±206
	PP	110	102±62	71	190±75	101	269±117	81	657±138
Complete	TP	76	66	63	60	64	52	41	54
(% complete)	PP	109	69	71	73	102	72	80	73
Peak heart rate (bpm)	ТР	22	112±19	22	111±18	35	117±22	16	122±28
	PP	80	147±21	62	153±26	87	152±25	61	154±27
VO2peak (L/min)	TP	22	0.85±0.25	22	0.85±0.36	35	0.99±0.37	17	1.17±0.43
	PP	80	1.07±0.37	63	1.22±0.37	87	1.32±0.43	62	1.31±0.47
POpeak (W)	TP	23	17±9	23	16±7	35	25±16	16	30 ±21
	PP	76	35±18	63	44±18	90	48±22	61	51±23
MMT sum score	TP	70	39±15	62	41±15	65	45±15	38	47±13
(/60)	PP	105	58±5	71	59±3	99	59±3	79	59±2
HHD sum score	TP	29	1076±495	30	1273±442	35	1473±523	23	1485±647
(N)	PP	81	1668±482	57	1810±488	71	1934±523	57	1992±508
$\text{FEV}_1$ (L)	TP	71	2.42±1.00	64	2.52±0.94	64	2.94±0.98	41	3.09±1.09
	РР	104	2.95±1.00	71	3.11±0.83	100	3.37±0.99	73	3.41±0.98
$FEV_1$	TP	71	64±25	64	66±20	64	77±22	41	81±24
(% predicted)	PP	104	77±22	71	80±18	100	86±20	73	88±21
FVC (L)	TP	71	2.89±1.30	64	2.96±1.05	64	3.55±1.23	41	3.75±1.34
	РР	104	3.59±1.25	71	3.81±1.03	100	4.15±1.21	73	4.21±1.18
FVC	TP	71	63±26	64	65±19	64	77±22	41	82±24
(% predicted)	PP	104	78±25	71	81±19	100	88±21	73	90±20

Table 1 Subject and lesion characteristics and physical capacity scores at the four measurement times

Subjects were included when at least 1 component of physical capacity was determined at that particular measurement time.

Abbreviations: PP, subjects with paraplegia; TP, subjects with tetraplegia.

physical capacity scores at the different measurement times for those subjects in whom at least 1 component of physical capacity could be obtained at that particular measurement time.

## Change in the Physical Capacity During and After Inpatient Rehabilitation

The POpeak,  $VO_2peak$ , muscle strength, and respiratory function improved during inpatient rehabilitation, and the  $VO_2peak$ , muscle strength, and respiratory function continued to improve after discharge. Figures 1 through 3 show the change in physical capacity over time as estimated with the basic regression model.

# *Relation Between the (Change in) Physical Capacity and Personal and Lesion Characteristics*

Table 2 presents data on the association between the (change in) physical capacity and the independent variables. The respective regression coefficients represent the change in outcome score associated with an increase in the independent variable of 1 unit. Because table 2 presents data after backward elimination, the dummy variables, and significantly contributing variables and interaction terms are given.



Figure 1 Change in POpeak (in watts) and Vo, peak (in L/min) as calculated from the basic model



Figure 2 Change in HHD score (newtons) and MMT sum score (maximum, 60) as calculated from the basic model



**Figure 3** Change in  $FEV_1$  and FVC as a percentage of the predicted value (compared with an age, sex, and height-matched able-bodied population) as calculated from the basic model.

*Age.* Age was significantly related to the POpeak and HHD score: an increase in age of 1 year was associated with a decrease of 0.3W or 7N, respectively. During inpatient rehabilitation the improvement in POpeak and HHD was significantly less in older subjects. After discharge the MMT score improved significantly more in older subjects, whereas the percentage of FVC improved significantly less in older subjects.

Sex. Men had a significantly greater POpeak,  $Vo_2peak$ , and HHD score than women (14W, .35L/min, 575N greater, respectively). During inpatient rehabilitation the improvement in POpeak and HHD score was significantly greater in men than in women.

*Level of the lesion.* In subjects with a tetraplegia the POpeak,  $VO_2$ peak, muscle strength, and percentage of FVC were significantly lower than in those with paraplegia. During inpatient rehabilitation, the improvement in muscle strength and percentage of FEV<sub>1</sub> was greater in subjects with a tetraplegia. After discharge the  $VO_2$ peak and MMT improved more in this group.

*Completeness of the lesion*. Subjects with a complete lesion had a significantly greater HHD score, and a significantly lower percentage of FVC than subjects with an incomplete lesion. In subjects with a complete lesion the improvement in percentage of  $\text{FEV}_1$  during inpatient rehabilitation was significantly greater than in subjects with an incomplete lesion.

#### Relation Between the Different Components of Physical Capacity

Table 3 presents data on the association between the Vo<sub>2</sub>peak, muscle strength, and respiratory function, and the POpeak in 1 column, and for the association between muscle strength and respiratory function, and the Vo<sub>2</sub>peak in the other column. The respective regression coefficients represent the change in POpeak or Vo<sub>2</sub>peak associated with an increase in Vo<sub>2</sub>peak, strength or respiratory function of 1 unit. The level of lesion, the Vo<sub>2</sub>peak, the HHD score, and the percentage of FVC were significantly

Table 2 Data on the l	ongitudinal relation	between subject and les	ion characteristics and	the (change in) physica	l capacity	
Independent Variable	POpeak (W)	Vo_peak (L/min)	MMT Sum Score	(N) CHH	FVC (%)	$FEV_1$ (%)
Discharge	49 (38 to 59)	1.13 (1.03 to 1.24)	51 (47 to 54)	1935 (1684 to 2186)	76 (66 to 86)	81 (76 to 86)
Δt1-t3*	-22 (-28 to -15)	-0.18 (-0.29 to -0.07)	-6 (-9 to -4)	-642 (-850 to -434)	-14 (-22 to -7)	-10 (-15 to -5)
$\Delta t2 - t3^{\dagger}$	-10 (-15 to -5)	-0.08 (-0.19 to 0.02)	-3 (-5 to -1)	-379 (- 559 to -199)	-7 (-14 to 0)	-7 (-11 to -2)
$\Delta t 3 - t 4^{\ddagger}$	6 (-2 to 14)	0.22 (0.08 to 0.37)	0.1 (-1.1 to 1.4)	135 (-53 to 323)	12 (5 to 18)	2 (-1 to 6)
Sex <sup>§</sup>	-14 (-20 to -7)	-0.35 (-0.48 to -0.23)	n.s.**	-575 (-742 to -408)	n.s.	n.s.
Age	- 0.3 (-0.5 to -0.1)	n.s.	0.01 (-0.06 to 0.09)	-7 (-13 to -1)	0.16 (-0.05 to 0.37)	n.s.
Level <sup>II</sup>	11 (7 to 15)	0.23 (0.11 to 0.35)	4 (2 to 5)	348 (215 to 481)	6 (1 to 10)	2 (-3 to 8)
Complete <sup>¶</sup>	n.s.	n.s.	n.s.	99 (25 to 173)	-5 (-8 to -2)	-1 (-5 to 3)
Sex * t1-t3 #	6 (1 to 12)	n.s.	n.s.	154 (34 to 274)	n.s.	n.s.
Sex * t2-t3 #	4 (0 to 9)	n.s.	n.s.	132 (20 to 244)	n.s.	n.s.
Sex * t3-t4 #	-1 (-7 to 5)	n.s.	n.s.	-51 (-167 to 65)	n.s.	n.s.
Age * t1-t3	0.2 (0.02 to 0.3)	n.s.	-0.01 (-0.06 to 0.04)	4 (0.1 to 8)	0.09 (-0.09 to 0.27)	n.s.
Age * t2-t3	0.1 (-0.03 to 2.4)	n.s.	0.01 (-0.02 to 0.05)	3 (-1 to 7)	0.09 (-0.08 to 0.26)	n.s.
Age * t3-t4	-0.1 (-0.3 to 0.1)	n.s.	0.04 (0.01 to 0.06)	-1 (-5 to 3)	-0.23 (-0.39 to -0.08)	n.s.
Level * t1-t3	n.s.	-007 (-0.19 to 0.05)	6 (4 to 7)	138 (11 to 265)	n.s.	5 (0.3  to  10)
Level * t2-t3	n.s.	0.01 (-0.11 to 0.13)	2 (1 to 4)	107 (-7 to 221)	n.s.	4 (-0.5 to 8)
Level * t3-t4	n.s.	-0.20 (-0.36 to -0.04)	-1 (-2 to -0.5)	-6 (-129 to 117)	n.s.	-2 (-5 to 2)
Complete * t1-t3	n.s.	n.s.	n.s.	n.s.	n.s.	-6 (-11 to -1)
Complete * t2-t3	n.s.	n.s.	n.s.	n.s.	n.s.	1 (-4 to 6)
Complete * t3-t4	n.s.	n.s.	n.s.	n.s.	n.s.	1 (-3 to 5)
All results are regression outcome associated with difference between t2 ai complete. # Interaction inpatient rehabilitation; **Association not signific	t coefficients ( $\beta$ ) and a n increase in the ir and discharge; <sup>#</sup> The dif terms are represented sex by t2-t3 represent ant (n.s.) and not inc	95% confidence intervals (C dependent variable of 1 un Ference between discharge 1 by the 2 terms joined by a 1s this relation between t2 a sluded in the model.	Is) for the model after ba it. "The difference in phy and 1 year after discharg sterisks. That is, sex by the aterisks. That is, sex by the ater of the sex by the ater of the sex by	ackward elimination. The r sical capacity between the e. <sup>6</sup> 0 = men; 1 = women; <sup>II</sup> 1-t3 represents the relation / t3-t4 represents this relat	egression coefficients rep start of active rehabilitati 0 = tetraplegia; 1 = parapi 1 between sex and the ch ion between discharge ar	resent the change in on and discharge; <sup>†</sup> The legia; <sup>¶</sup> 0 = incomplete; 1= ange in outcome during and 1 year after discharge.

#### Chapter 3

Independent Variable	POpeak (W)	Vo <sub>2</sub> peak (I/min)
Discharge	-3.5 (-30 to -17.1)	0.12 (-0.09 to 0.32)
$\Delta t1 - t3^*$	-3.8 (-5.9 to -1.7)	-0.07 (-0.13 to -0.01)
$\Delta t2 - t3^{\dagger}$	-0.5 (-2.5 to 1.4)	-0.01 (-0.07 to 0.05)
$\Delta t3 - t4^{\ddagger}$	0.7 (-2.3 to 3.6)	0.04 (-0.04 to 0.12)
Sex <sup>§</sup>	n.s.¶	-0.13 (-0.24 to -0.02)
Level <sup>∥</sup>	7.8 (4.9 to 10.6)	n.s.
Vo <sub>2</sub> peak (L/min)	29.3 (25.8 to 32.7)	n.e.¶
HHD (N)	0.0087 (0.0057 to 0.0117)	0.0004 (0.0003 to 0.0005)
FVC (%)	0.09 (0.04 to 0.14)	n.s.
FEV <sub>1</sub> (%)	n.s.	0.0039 (0.0021 to 0.0058)

Table 3 Data on the longitu	dinal relation between the	components of physical cap	bacity
-----------------------------	----------------------------	----------------------------	--------

All results are regression coefficients ( $\beta$ ) and 95% CIs for the model after backward elimination. \*The difference in physical capacity between the start of active rehabilitation and discharge. <sup>†</sup>The difference between t2 and discharge. <sup>‡</sup>The difference between discharge and 1 year after discharge. <sup>§</sup>0 = men; 1 = women; <sup>II</sup>0 = tetraplegia; 1 = paraplegia. <sup>¶</sup>Association not significant (n.s.); independent variable not entered (n.e.).

associated with the POpeak. For example, an increase in strength of 100N was associated with an increase in POpeak of .87W (regression coefficient, .0087, multiplied by 100N) (see table 3). Sex, the HHD score, and percentage of  $FEV_1$  were significantly associated with the Vo<sub>2</sub>peak.

#### Discussion

#### Change in the Physical Capacity During and After Inpatient Rehabilitation

In accordance with Hjeltnes,<sup>131</sup> the increase in physical capacity is suggested to be greater during the early phase of inpatient rehabilitation than at a later stage (see figs 1–3). The fast recovery at the beginning of rehabilitation could be attributed to the start of the learning and training process, but also to natural recovery and the recuperation from trauma and complications.

In contrast to previous suggestions,<sup>29,40</sup> the improvement found in Vo<sub>2</sub>peak, muscle strength, and respiratory function after discharge from inpatient rehabilitation suggests that the physical activity level during everyday life and during possible outpatient rehabilitation or (sporting) activities was sufficient to improve these components of physical capacity in the present study population. Future analysis of the association between outpatient rehabilitation or physical activity, and the level of physical capacity is required.

*Vo2peak and POpeak.* During inpatient rehabilitation the POpeak and Vo2peak improved 41% and 24% (increase in score as a percentage of score at t1, as calculated with basic model), respectively, which is within the range reported by others.<sup>49,69</sup>

The relatively greater improvement in POpeak as compared with the increase in Vo2peak could be attributed to an improved wheelchair propulsion efficiency, which may contribute to an increase in POpeak.<sup>25,133,140</sup> In contrast to other studies,<sup>57,141</sup> the POpeak did not change significantly after inpatient rehabilitation. A reason for this could be that the wheelchair configuration has an influence on the POpeak.<sup>141</sup> One year after discharge, subjects are expected to be trained in the use of their own wheelchair, and the compulsory use of a research wheelchair (instead of a private well-adapted wheelchair) at all measurement times may have underestimated the change in POpeak in our study.

*Strength of the Upper Extremity.* In our study population the MMT improved 7% and the HHD score 20% (increase in score as a percentage of the score at t1) during inpatient rehabilitation. Hjeltnes and Wallberg-Henriksson<sup>69</sup> reported a larger relative change in the MMT score, however, they reported only on subjects with tetraplegia. In our study population the subjects with tetraplegia improved more than those with paraplegia, therefore the results for our subjects with tetraplegia are comparable to previous findings.<sup>69</sup> The MMT score in our subjects changed less than the HHD score, probably because the MMT is limited in its ability to show a continued increase once a muscle group has reached a grade 4 or 5.<sup>21,65,66</sup>

*Respiratory Function.* The improvement we found during inpatient rehabilitation, was similar to findings by Liaw et al,<sup>105</sup> and may be attributed to natural recovery, but also to a rehabilitation program that includes endurance training which may positively influence respiratory function.<sup>102,109,128</sup> To date there are no known longitudinal data, which support the significant improvement in respiratory function found after discharge. Other studies reported a decrease in respiratory functioning 1 year after injury.<sup>142</sup> However, differences in design hamper a valid comparison with these studies.

# *Relation Between the (Change in) Physical Capacity and Personal And Lesion Characteristics*

*Age.* The decrease in POpeak and HHD with an increase in age, was in agreement with cross-sectional studies.<sup>12,99,143</sup> However, to date no prospective study has reported the finding that older subjects showed less improvement in these components than younger subjects.

*Sex.* The relation found between gender and the (change in) HHD score, could be attributed to the relatively small functional muscle mass in women, and may have contributed to the reduced improvement in POpeak in women during inpatient rehabilitation.<sup>144</sup>

Chapter 3

*Level of the lesion.* The influence of the level of the lesion on the components of physical capacity was in agreement with previous studies which have more specifically shown a reduction in physical capacity with an increase in affected segments.<sup>12,92,98,99</sup> The relatively large improvement in VO<sub>2</sub>peak, muscle strength and percentage of FEV<sub>1</sub> in our subjects with a tetraplegia was possibly the result of both training and the recovery of partially denervated segments, whereas the improvement in paraplegia was more the result of training. Dallmeijer et al<sup>57</sup> did not find this difference in recovery, which may partly be explained by the fact that the subjects with a tetraplegia in their study had relatively more musculoskeletal complaints than their subjects with a paraplegia.

*Completeness of the lesion.* The lower HHD score found in our subjects with an incomplete lesion (compared with those with a complete lesion) does not seem logical, and does not coincide with findings by Dallmeijer et al,<sup>38</sup> but could be attributed to our study population. Because we focused on wheelchair-dependent subjects, the relatively well-performing subjects with an incomplete lesion were excluded. Additionally, the inclusion of only those subjects able to perform the tests would have excluded poorly performing subjects with a complete lesion. This selection may have given a distorted picture of the difference in performance between the two groups.

#### Relation Between the Different Components of Physical Capacity

To date no other study has assessed the relations between the components of physical capacity on a longitudinal basis. In agreement with the present findings, correlation studies showed that both the Vo<sub>2</sub>peak and muscle strength were related to the POpeak, and that muscle strength was related to the Vo<sub>2</sub>peak.<sup>12,25,129,144</sup> Although the relation was significant in the present study, a relatively large (compared with the initial score) increase in respiratory function was associated with a relevant change in POpeak or Vo<sub>2</sub>peak. Janssen et al<sup>144</sup> suggested that the different components of physical capacity are limited by the actual active muscle mass and that strength training programs may improve POpeak and Vo<sub>2</sub>peak. Our findings seem to support this suggestion, but have to be interpreted with caution as complete causality cannot be established with our study's design.

#### Limitations

The methodology of the study had some limitations. First, the reported level of physical capacity may have been overestimated because the tested subjects represented a positive selection of all persons with SCI: they had survived the critical period after injury and were not limited by (cardiovascular or musculoskeletal) complaints. Second, there

was an increase in the proportion of subjects able to perform the maximal aerobic exercise test and muscle strength assessments over time, which probably reflected an improvement in functional status of these additional subjects. This was not necessarily revealed by an improvement in the results and, therefore, the improvement in these components may have been underestimated. Third, because the subjects able to perform the tests varied for each component of physical capacity, the results of each component could reflect the performance of different subgroups.

The results on muscle strength of the upper extremity have to be interpreted with caution. By summing the scores, information about the change in strength of specific muscle groups could be lost. The MMT score was chosen because it is widely used in the clinical setting; however, it is an ordinal, subjective scale with each increase in grade representing a different increase in strength.<sup>21</sup> To facilitate the collection of reliable data, a standardized protocol was used to describe the procedure, and experienced research assistants received training both prior to and during the study period. The HHD score is presented by a valid and reliable continuous scale, but it has some restrictions: in the assessment of the wrist extensors its reliability is limited, and it does not cover the lower ranges of strength.<sup>63,66,145,146</sup> The development of new manageable equipment to measure muscle strength through all ranges and for all muscle groups seems important.

#### Conclusions

We found positive changes in the in the different components of physical capacity both during and after inpatient rehabilitation. The continued improvement after discharge was contrary to our expectations and illustrates that it is worthwhile to regularly assess the physical capacity of persons with SCI after discharge. It is also important to create optimal conditions (eg, educational programs and training facilities) to facilitate further improvement. The results demonstrate that subpopulations show different changes in physical capacity. Specific training and follow-up programs should explore these differences in order to help each patient develop his or her maximal potential physical capacity suggest that training one component could contribute to the improvement of other components, but this must be confirmed by intervention studies.

*Suppliers* (a) Angio Lode Ergometer; Lode BV, Zernikepark 16, 9747 AN Groningen, The Netherlands; (b) MicroFET; Biometrics Europe BV, Kabelstraat 11, 1322 AD Almere; (c) Jaeger Toennies, Nikkelstraat 2, 4823 AB Breda, The Netherlands, (d) MlwiN version 1.1; Centre for Multilevel Modelling, Graduate School of Education, University of Bristol, 35 Berkeley Sq, Bristol, BS8 1JA, UK.

# 4

# Prognostic models for physical capacity at discharge and 1 year postdischarge from rehabilitation in persons with spinal cord injury

JA Haisma, LH van der Woude, HJ Stam, MP Bergen, TA Sluis, S de Groot, AJ Dallmeijer, JB Bussmann In: *Archives of Physical Medicine and Rehabilitation* 2007; 88: 1694-1703

## Abstract

*Objective:* To develop prognostic models for the physical capacity at discharge and 1 year after discharge from inpatient rehabilitation in persons with spinal cord injury (SCI).

*Methods:* Inception cohort at 8 Dutch rehabilitation centers. The data collected at start of rehabilitation (n=104), at discharge (n=81), and 1 year later (n=74) were used to make prognostic models. Physical capacity was determined on these occasions by the endurance capacity (peak oxygen uptake –  $Vo_2peak$ ; in L/min – and peak power output – POpeak; in watts) during a maximal exercise test, by arm muscle strength, and by respiratory function. Multiple regression models either with, or without prior outcome, evaluated subject, lifestyle, and lesion-related predictors.

*Results:* Only start VO<sub>2</sub>peak contributed to the prediction of discharge VO<sub>2</sub>peak ( $R^2$ =.51). Discharge VO<sub>2</sub>peak alone contributed to its prediction 1 year later ( $R^2$ =.75). Start POpeak, sex, age, and level of lesion contributed to discharge POpeak ( $R^2$ =.73). Discharge POpeak, hours of employment before injury, and level of lesion contributed to POpeak 1 year later ( $R^2$ =.81). Models without prior outcome explained less variance. Education, employment, body mass index, not smoking, and stabilization of spine positively contributed to endurance capacity. Muscle strength was well predicted ( $R^2$  range, .68–.84). Without prior outcome, respiratory function was poorly predicted.

*Conclusions:* Because prior outcome contributed to an accurate prediction, the early assessment of physical capacity is important when establishing prognoses. Although their accuracy warrants caution, models could complement clinical expertise when informing patients of expected physical outcome, and identifying those at risk of a low physical capacity.

#### Introduction

After spinal cord injury (SCI), there are several reasons why the prediction of physical capacity, defined as the ability of muscles and the cardiorespiratory system to maintain a certain level of activity,<sup>17</sup> is important. First, the prediction of physical capacity allows clinicians to estimate prognoses. With this knowledge they can inform patients, family, and caregivers about the possible consequences of the lesion.<sup>121,147</sup> Second, the prediction of physical capacity at discharge may provide rehabilitation goals.<sup>148</sup> Third, prognoses of outcome 1 year after discharge may indicate the necessity of outpatient training, assistive devices and other postdischarge care.<sup>149,150</sup> Finally, prognoses of physical capacity partly indicate functional consequences, the expected level of well-being, and those at risk of complications.<sup>9,46,59</sup> In summary, during and after spinal cord rehabilitation insight into the expected physical and functional capacity could help set realistic goals, target screening and interventions, plan postdischarge care, and justify follow-up visits.

Previous studies have assessed predictors of physical capacity, but had methodologic limitations. Most were cross-sectional or combinations of cross-sectional and longitudinal associations and were, therefore, not strictly predictive.<sup>36,60,147,151</sup> Furthermore, the prognostic models that were made had different shortcomings.152,153 First, the outcome measure may have been limited by a ceiling effect, which hampered the prediction of outcome at a later phase of recovery.<sup>121,148,153</sup> Second, often one component of physical capacity was investigated, whereas the simultaneous assessment of different aspects (eg, endurance capacity, muscle strength, respiratory function) may more specifically indicate the required treatment strategy,<sup>17,153</sup> Furthermore, because most persons with SCI are wheelchair users, the endurance capacity (ie, the peak oxygen uptake (Vo<sub>2</sub>peak) and peak power output (POpeak) during a wheelchair exercise test) is related to functioning. Therefore, the endurance capacity may give a more valid indication of functional prognoses than would muscle strength alone.<sup>33,133</sup> Third, earlier studies did not always report the prediction accuracy of models.<sup>34</sup> The accuracy of the model, that is, whether the predicted outcome corresponds with the observed outcome, partly determines its clinical relevance.41,154

The rehabilitation of patients with SCI would not only benefit from an accurate prediction of outcome, but also from the identification of modifiable predictors. Unfortunately, most known determinants of physical capacity, like age, sex and level of the lesion, can be taken into consideration, but are not changeable.<sup>60,151</sup> Some modifiable aspects of lifestyle (eg, smoking or alcohol use) and management (eg, stabilization of the spine) were found to be associated with physical capacity.<sup>36,46,124,141,155,156</sup> However, causality cannot be established from the available

cross-sectional data for those with SCI. If we could demonstrate that giving up smoking, increasing participation in employment, increasing sport activity, moderating alcohol or dietary intake, and stabilizing the spine (by surgical fixation, or wearing a halo or corset) predict a favorable outcome, we might have means to enhance physical capacity during an early phase of recovery.

The aim of this study was to develop prognostic models for 4 aspects of physical capacity (ie,  $Vo_2$ peak, POpeak, arm muscle strength, respiratory function) in wheelchair-dependent subjects with SCI. We aimed to assess the contribution of both modifiable (smoking, alcohol consumption, body mass index [BMI], stabilization of spine) and not modifiable characteristics (age, sex, education, employment, level of the lesion) at the start of rehabilitation in predicting physical capacity at discharge, and the contribution of these characteristics at discharge in predicting outcome 1 year after discharge from inpatient rehabilitation.

#### Methods

#### Participants

This study was part of the Dutch research program "Physical Strain, Work Capacity and Mechanism of Restoration of Mobility in the Rehabilitation of Persons with an SCI." Between 2000 and 2005, the 8 participating rehabilitation centers admitted a total of 387 patients with SCI. Patients were eligible for inclusion if they were wheelchair dependent, were between 18 and 65 years of age, had enough comprehension of the Dutch language to understand the purpose of the study and had no progressive disease or psychiatric condition.<sup>151</sup>

All subjects followed a regular inpatient rehabilitation program at specialized spinal cord units. Clinicians at these units largely provide their care according to guidelines supported by the Dutch Flemish Spinal cord Society.<sup>3</sup> Although, this society recommends patients to visit these specialized centers for outpatient follow-up or training,<sup>3</sup> some subjects visited these centers, whereas others visited an outpatient clinic closer to their homes. The medical ethics committee approved the experimental protocol, and prior to participation all subjects gave a written informed consent.

#### Procedure

Data were obtained at the start of active rehabilitation, when the subject began functional, endurance, and strength training in the wheelchair. The start of active rehabilitation was marked by the moment when the subject could to sit in a wheelchair for at least 3 hours at a time. Subsequently, data were obtained at discharge and 1 year after discharge from inpatient rehabilitation.<sup>151</sup> Prior outcome was used to predict physical capacity at follow-up. Therefore, subjects were included if their physical capacity could be assessed on at least 2 occasions; either both at the start of rehabilitation and at discharge (when predicting discharge outcome), or both at discharge and 1 year later (when predicting outcome 1 year after discharge). To optimize the quality of the data, a standardized protocol was followed, whereby 1 research assistant was responsible for the data collection at each rehabilitation center. Furthermore, all 8 research assistants were experienced rehabilitation professionals, who received training and instructions both before and during the study program. Additionally, the equipment (treadmill,<sup>a</sup> Oxycon Delta,<sup>b</sup> handheld dynamometer<sup>c</sup>) was calibrated before use, and the same equipment was used at subsequent assessments.

#### Physical Capacity

The endurance capacity was determined by the Vo<sub>2</sub>peak and the POpeak during a graded maximal wheelchair exercise test on a motor driven treadmill. We asked subjects not to smoke or drink coffee or alcohol 2 hours before the test. Additionally, subjects were asked to void their bladder prior to the exercise test. Subjects started with 2 blocks of submaximal exercise, during which a suitable treadmill velocity was chosen for the graded maximal exercise test. After 2 minutes of rest, the maximal exercise test started. Subjects were encouraged to keep up the speed of the treadmill while its inclination was increased .36° every minute. Finally, the test stopped when the subject was completely exhausted, or could no longer keep up the speed of the treadmill.<sup>133,151</sup> The Vo<sub>2</sub>peak (in L/min) was defined as the highest value of oxygen consumption recorded during 30 seconds. The POpeak (in watts) was determined from the velocity of the treadmill belt and the result of a separate wheelchair drag test.<sup>27</sup> It was defined as the power output at the highest inclination that the subject could maintain for at least 30 seconds.

The strength of 10 muscle groups (shoulder abductors, internal and external rotators, elbow flexors and extensors) was determined with the handheld dynamometer for those muscles with a strength of at least grade 3 during manual muscle testing.<sup>119,151</sup> Subsequently, the maximum force (in newtons) of the muscle groups was summed.<sup>151</sup> Sum scores were only calculated if strength could be determined in these 10 muscle groups. Respiratory function was determined from a standardized lung function test.<sup>137</sup> The forced expiratory volume per second (FEV<sub>1</sub>) was expressed as a percentage of the volume expected for an age, sex, and height-matched ablebodied person.<sup>137</sup>

#### Predictors

Because at discharge or at follow-up 1 year after discharge, prior physical capacity may not be known, we made models with prior physical capacity, and models without prior physical capacity. For the models including prior outcome, physical capacity at the start was included when predicting discharge outcome, and physical capacity at discharge was included when predicting outcome 1 year after discharge. Additionally we determined the contribution of both modifiable and not modifiable characteristics.

Subject and Lifestyle Characteristics. At the start of rehabilitation, self-report information on date and place of birth, education, smoking, alcohol use, and hours of sport participation and employment prior to the injury were collected. The educational level was defined as follows: "low" for primary or prevocational practical education (score, 1); "middle" for lower secondary or vocational practical education (score, 2); and "high" for higher secondary or higher education (score, 3). If subjects (or 1 or both parents) were born outside The Netherlands, they were defined analogous to criteria used by the Central Bureau of Statistics, as not being of Dutch origin (not Dutch, 1; Dutch, 0). Although some subjects refrained from smoking during rehabilitation, most had started smoking again 1 year after discharge. Therefore, we defined a smoker based on cigarette use prior to the lesion (smoker, 1; nonsmoker, 0). Similarly, to determine alcohol use, subjects were asked whether they used beer, wine, or liquor prior to the lesion (alcohol use, 1; no alcohol use, 0). Subjects were asked how many hours a week on average they had participated in sports (team or individual), and how many hours a week they had been employed before the lesion. At each assessment, age and BMI (calculated as kg/m<sup>2</sup>) were calculated.

*Lesion Characteristics.* At each assessment, the physician determined level and completeness of the lesion. Tetraplegia (score, 0) was defined as a lesion at or above the T1 segment, and paraplegia (score, 1) as a lesion below the T1 segment. A complete lesion (score, 1) was defined as motor complete, that is, American Spinal Injury Association (ASIA) grade A or B, and an incomplete lesion (score, 0) as motor incomplete, that is, ASIA grade C or D.<sup>1</sup> A physician consulted medical charts and self-report information to establish whether and how the spine had been stabilized (surgical stabilization, 1; no surgical stabilization, 0; and stabilization by means of halo or corset, 1; no halo or corset, 0).

#### Data Analysis

Subject, lifestyle, and lesion characteristics were summarized with descriptive statistics. Based on these characteristics, we compared the physical capacity between different subject groups with independent t tests. Potential predictors were cross-tabulated and if there was substantial correlation between characteristics (R>.80), only the characteristic with the highest correlation with physical capacity was included in a multivariate model.<sup>157</sup> Multiple linear regression analyses revealed several models predicting four aspects of physical capacity. Characteristics known at the start were included as independent variables when predicting physical capacity at discharge. Characteristics known at discharge were included as independent variables when predicting physical capacity 1 year after discharge.

For the first model, physical capacity at the start was included as an independent variable by forced entry (ie, it was included in the model first, regardless of its significance). Subsequently other characteristics (determined at the start of active rehabilitation) were included following the stepwise forward procedure. Therefore, the characteristic that explained most of the remaining part of the variance (ie, the variance not already explained by prior outcome) was included next. The analysis was repeated, and the characteristic remained included if it made a significant ( $P \le .05$ ) contribution to the predictive power of the model. This procedure, of selecting a characteristic, determining its contribution and analyzing the power of the model, was repeated for all characteristics until a model with the highest predictive power for these characteristics remained. Subsequently, a separate model was made without physical capacity at the start, which, therefore, included the other independent variables following the aforementioned stepwise-forward procedure.

Similarly, we made prognostic models for the physical capacity one year after discharge. The physical capacity at discharge was included by forced entry, and other discharge characteristics were included following the stepwise-forward procedure. A separate model was made without physical capacity at discharge. For each model, the explained variance and the standard deviation (SD) of the residuals was given. Residuals are the differences between the observed and the predicted outcome. The residual SD portrays how accurately the model predicts outcome in this population and, therefore, partly determines how meaningful a model is in the general population.<sup>41</sup> All data were analyzed with SPSS.<sup>d</sup>

#### Results

Of the 387 patients with SCI admitted for rehabilitation during the study period, 225 were considered eligible for the research program. Because we made prognostic models, only those patients whose physical capacity was determined on 2 occasions were included in our study population. This meant that of the 104 subjects for whom we determined the POpeak at the start, 81 were included when predicting the POpeak at discharge. They had started their active rehabilitation 104±66 days postinjury. Seventy-

Subject or Lifestyle Characteristic	<i>Start (n=81)</i>	Discharge (n=74)
Age at start or discharge (y)	40±14	37±13
Men (%)	74	73
Not of Dutch origin (%)*	17	26
Employment before lesion (h/wk)†	34±22	32±21
Education (%): Low‡	38	35
Middle	36	28
High	26	37
Smoker before lesion (%)	46	47
Consumed alcohol before lesion (%)	79	73
BMI start or discharge (kg/m2)	23±4	23±4
Sports before lesion (h/wk)§	3±4	4±5
Lesion characteristic		
Paraplegia at start or discharge (%)	79	80
Complete lesion at start or discharge (%)	64	64
Surgical stabilisation of spine (%)	54	57
Halo or corset (%)	28	37

**Table 1** Characteristics at the start and at discharge from rehabilitation as predictors of endurance capacity at discharge and 1 year after discharge, respectively

Values are mean ± SD or percentage. \*Subject or at least 1 of parents born outside The Netherlands. †Mean hours per week employed before lesion. ‡"Low" is primary or prevocational practical education; "middle" is lower secondary or vocational practical education; "high" is higher secondary or higher education. \$Mean hours per week participating in sports before lesion.

four subjects were included when predicting the POpeak 1 year after discharge. Their discharge took place 258±115 days postinjury. Table 1 presents their descriptive subject, lifestyle, and lesion characteristics.

The population included in the research program showed a similar distribution of these characteristics.<sup>151,158</sup> Of all patients admitted with SCI during the study period, however, relatively few (48%) had a complete lesion.<sup>158</sup> The number of subjects included when predicting different aspects of physical capacity at the 2 occasions did not coincide for several reasons. First, hindered by cardiovascular contraindications, limited wheelchair skills or complications, not all subjects completed a maximal exercise test on all 3 testing occasions.<sup>151</sup> Second, fewer subjects were included when predicting strength, because only those with reasonable strength in 10 arm muscle groups were included. Table 2 presents the mean physical capacity at each assessment both as a predictor and as an outcome. Table 3 gives the mean physical capacity found in several subject groups. The significance levels indicated in table 3 were based solely on one-to-one associations determined with independent t tests.

Tables 4 and 5 present results of the multiple linear regression analyses for physical capacity at discharge and 1 year later, respectively. To facilitate interpretation of these results, examples are given in the footnotes. For each aspect of physical capacity results of the model with, and the model without prior outcome are given.

	Predi at dis	icting physical c scharge	apacity	Prec 1 ye	licting physical cap ar after discharge	pacity
Variable Predictor	п	Predictor (start)	Outcome (discharge)	п	Predictor (discharge)	Outcome (1 yr later)
Vo2peak (L/min)	75	1.05±0.29	1.31±0.41	73	1.27±0.46	1.33±0.54
POpeak (W)	81	31±16	45±19	74	46±24	48±26
Arm muscle strength (N)	63	1534±503	1807±501	55	1756±481	1860±551
FEV <sub>1</sub> (%)*	113	77±23	87±20	93	87±20	90±20

Table 2 Physical capacity as Predictor or as Outcome

Values are n or mean  $\pm$  SD. \*Expressed as percentage of the flow expected for an age, sex, and height-matched able-bodied person.

Table 4 tells us, for example, that the regression equation predicting discharge Vo<sub>2</sub>peak reads: 0.25±1.01\* start Vo<sub>2</sub>peak. This means that only Vo<sub>2</sub>peak at the start contributed to the Vo, peak at discharge. Table 5 tells us that, again, only the discharge Vo<sub>2</sub>peak contributed to its prediction 1 year later. The POpeak at the start, sex, age, and level of the lesion predicted POpeak at discharge (Table 4). The discharge POpeak, hours of employment before injury, and level of the lesion contributed to the prediction of POpeak 1 year later (Table 5). Several not modifiable characteristics were identified as predictors of discharge outcome: age, sex, employment, education, and level of the lesion. These characteristics, together with ethnicity, also contributed to the prediction of outcome 1 year after discharge. Some modifiable predictors of discharge outcome were identified: BMI, smoking, and stabilization of the spine. Only BMI made a modifiable contribution to the prediction of outcome 1 year after discharge. Tables 4 and 5 also present the variance explained by each predictor in these multivariate models. They indicate that the variance in discharge outcome is explained mostly by prior physical capacity (variance explained, 51%-70%) (see table 4). The variance in physical capacity 1 year after discharge is even more so explained by outcome at discharge (variance explained, 74%-82%) (see table 5). Furthermore, level of the lesion, sex, and employment make relatively large contributions to the explained variance. To indicate the prognostic value of the models, the total explained variance (adjusted  $R^2$ ) and the residual standard deviations are given.

Table 3 Physical capacity :	at discharge :	and 1 year after disch	arge as found in	subject groups define	ed at the start	and at discharge, re-	spectively	
	Vo_peak (L/n	iin)	POpeak (W)		Arm Muscle S	strength (N)	$FEV_1$ (%)	
	discharge	year after discharge	discharge	year after discharge	discharge	year after discharge	discharge	year after discharge
Age at prior assessment								
<40 (y)	$1.36\pm0.40$	$1.35\pm0.56$	47±19	49±26	1981±531†	1961±556	85±18	88±18
≥40 (y)	$1.21 \pm 0.42$	$1.25 \pm 0.47$	40±20	46±26	1685±505†	1712±520	88±21	90±22
Sex								
Men	$1.39\pm0.37$	$1.39\pm0.52^{*}$	49±19†	52±26†	2049±474†	2096±497†	80±21	87±21
Women	$1.04\pm0.41$	$1.09\pm0.48^{*}$	29±12†	34±19†	1303±340†	1398±338†	83±23	83±22
Ethnicity								
Not Dutch	$1.20\pm0.36$	$1.09\pm0.47$	43±19	44±20	1821±374	1758±385	79±18	81±23
Dutch	$1.31 \pm 0.42$	$1.37\pm0.54$	44±20	49±27	1806±594	1842±601	83±23	87±21
Education‡								
Low	1.14±0.36†	$1.19\pm0.39$	38±20*	39土22*	1639±459	1746±545	80±21	82±22
Middle/high	1.39±0.42†	$1.40\pm0.59$	48±18*	53±26*	1845±557	1928±566	84±23	88±21
Employment before lesion (h/wk)								
≤16	$1.10\pm0.32$	$1.13\pm0.28^{*}$	35±15†	35±16†	1505±488†	1591±536*	80±25	82±24
>16	$1.37\pm0.42$	$1.38 \pm 0.57^{*}$	47±20†	52±26†	1935±544†	1941±544*	83±20	88±21
Level of lesion								
Paraplegia	$1.34\pm0.428$	$1.37\pm0.54^{*}$	48±19†	52±25†	1985±517†	1910±566	89±19*	89±20
Tetraplegia	$1.11\pm 0.34^{*}$	$1.10\pm0.42^{*}$	27±14†	27±17†	1468±392†	1608±470	81±19*	89±21
Surgically stabilized spine								
Yes	$1.31 \pm 0.40$	$1.36\pm0.53$	46±21	51±27	1885±533	1815±543	82±22	85±21
No	$1.27\pm0.43$	$1.24\pm0.51$	41±18	43±22	1705±580	1814±582	82±21	87±20

Chapter 4

HAID OF COISE								
Yes	$1.37 \pm 0.38$	$1.42 \pm 0.56$	50±19	56±26*	1885±519	1886±553	85±22	86±20
No	$1.26 \pm 0.43$	$1.24\pm0.50$	42±19	42±24*	1770±579	1787±591	81±22	85±23
Smoking								
Smoker	$1.19\pm0.39$	$1.22 \pm 0.52$	38±19†	42±24	1803±560	1802±518	79±18	83±17
Nonsmoker	$1.36 \pm 0.42$	$1.39 \pm 0.39$	49±20†	52±26	1809±574	1843±603	84±24	88±25
Alcohol								
Some	$1.31 \pm 0.42$	$1.33 \pm 0.55$	44±20	49±26	1902±537†	1923±547*	82±22	85±22
None	$1.22 \pm 0.41$	$1.23 \pm 0.46$	$41\pm 20$	42±21	1459±511†	1541±501*	82±21	89±20
BMI prior assessment								
Overweight§	$1.29 \pm 0.46$	$1.31 \pm 0.46$	45±21	43±26	$1889\pm605$	1791±525	86±25	87±25
Not overweight	$1.29\pm0.40$	$1.32 \pm 0.56$	44±19	49±25	1761±544	1821±592	80±20	85±20
Sports before lesion								
<3 (h/wk)	$1.27 \pm 0.41$	$1.31 \pm 0.50$	$41\pm 20$	45±25	1756±522	1792±518	80±23	84±24
≥3 (h/wk)	$1.33 \pm 0.41$	$1.32 \pm 0.56$	47±19	51±26	1879±596	$1889\pm 651$	84±20	88±18
Values are mean + SD_Co	mparisons hetwe	satuora andread	mada with indeper	Mant ttasts *D< 05.4	iine si "uno I"+ 10 >0	citocotico activo	unda Leader	"middle"

IIIII illi) 2 Values are mean  $\pm$  SD. Comparisons between subject groups made with independent *t* tests. \**P* $\leq$ .05;  $\frac{1}{7}$ *P* $\leq$ .01.  $\frac{1}{7}$ "Low" is primary or prevocational practical to lower secondary or vocational practical education; "high" is higher secondary or higher education. (BMI  $\geq$ 25kg/m<sup>2</sup>.

Table 4 Progn	ostic models	for	physical cap	acity a	at discharge: ste	spwis	e forward mu.	ltiple l	inear regressio	on and	ulyses with p	redicto	ors defined at	start	
	$Vo_2 peak (I/n=75)$	(mim)			$POpeak (W) \\ (n=81)$				Arm Muscle . (n=63)	Streng	th (N)		$FEV_1$ (%) (n=113)		
	With Start		Without Start		With Start		Without Start		With Strength		Without Str	ength	With Start	Witho	out Start
	VO2PCAIN		V02PCar		1 Open		1 Opcar		מו טומוו		al 01a11		1,77,1	1 7 4	
Independent Variable	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta\pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta\pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE ~\%$
Constant	0.25±0.13		0.56±0.16		15±5		-2±11		882±124		667±254		24±5		91±3
Start PC	$1.01\pm0.12^{\ddagger}$	51	n.e.		0.80±0.08 #	65	n.e.		0.57±0.07#	70	n.e.		0.72±0.05#	59	n.e.
Sex*			$0.40\pm0.10^{\dagger\dagger}$	13	$10\pm3^{\dagger\dagger}$	١ſ	20±3††	23	$401\pm63^{\dagger\dagger}$	6	674±73#	39			
Age (y)					$-0.23\pm0.08$ **	0	$-0.29\pm0.12$	3	$-8.08\pm1.90^{\dagger\dagger}$	4	$-12\pm3^{\ddagger}$	6			
Education†			$0.13\pm0.05$	Ś							$100 \pm 44$	3			
Employment‡													$0.12 \pm 0.06$	2	
Smoking§							-8±3	3							-10±4 5
BMI (kg/m <sup>2</sup> )							$1.20\pm0.43^{**}$	7			30土9**	4			
Lesion level <sup>¶</sup>			$0.26 \pm 0.11$	9	8±3	7	$21 \pm 4^{\dagger\dagger}$	20	177±69	7	456±83#	15			
Surgical stabilization													6±2	7	
Halo or corset#							8±3	4							
Accuracy	R <sup>2</sup> =.51 RSD=.29		R <sup>2</sup> =.24 RSD=.36		R <sup>2</sup> =.73 RSD=10		R <sup>2</sup> =.54 RSD=13		R <sup>2</sup> =.84 RSD=203		<i>R</i> <sup>2</sup> =.70 RSD=277		R <sup>2</sup> =.63 RSD=12	$R^{2}=.0$ RSD=	5 20
$P\leq .05$ unless as	indicated belo	OW. A	who be reviations:	n.e.: n	not entered; PC, 1	physic	al capacity; RSI	D, resid	lual standard de	eviatio	n. *men=1; w	omen=(	. †1=low, 2=m	iddle;	3=high. ‡Mear

hours per week before lesion. [Smoker=1; nonsmoker=0. <sup>II</sup>Paraplegia=1; tetraplegia=0. <sup>II</sup>Stabilization of spine=1; no surgical stabilization=0. #Halo or corset stabilizing spine=1; no halo or corset=0. \*\*P≤.001.

## Chapter 4

	Vo2peak* (L/ (n= 73)	(nim)			POpeak† (W) (n= 74)	2			Arm Muscle (n=55)	Strengt	$(N) q_1$		$FEV_1 (\%)$ $(n=93)$	
	With Dischar Vo <sub>2</sub> peak	ge	Without Dise Vo <sub>2</sub> peak	charge	With Dischar POpeak	ge	Without Discl POpeak	harge	With Dischar Strength	ge	Without Disc Strength	charge	With Discharge	e FEV <sub>1</sub>
Independent Variable	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE$	%	$\beta \pm SE$	%	β±SE	%
Constant	0.04±0.09		$0.78\pm0.16$		-4±4		12±10		39±117		447±300		16±5	
PC at discharge	1.02±0.07¶	75	n.e.		0.86±0.06 ¶	79	n.e.		1.04±0.06 ¶	82	n.e.		0.86±0.05 ¶	74
Sex*							13±5	4			732±91 ¶	33		
Age (y)							$-0.35\pm0.17$	3			−17±4 ¶	~		
Ethnicity									-190±75	0	$-383\pm112^{\parallel}$	Ś		
Education <sup>†</sup>											127±50	4		
Employment <sup>‡</sup>			$0.01\pm0.00^{\parallel}$	6	0.20±0.07∥	1	$0.48\pm0.11$	15						
Lesion level§			$0.32 \pm 0.14$	Ś	8±4	1	30±6¶	20			568±102¶	12		
BMI (kg/m <sup>2</sup> )											46±12 <sup>II</sup>	_		
Accuracy	R <sup>2</sup> =.75 RSD=.27		$R^2$ =.14 RSD=.50		<i>R</i> <sup>2</sup> =.81 RSD=11		$R^2$ =.42 RSD=20		R <sup>2</sup> =.84 RSD=223		R <sup>2</sup> =.68 RSD=313		$R^{2=.74}$ RSD=10	

61



Figure 1 Bland-Altman plot illustrates the distribution of the differences between the observed and the predicted discharge POpeak.

Overall, POpeak and muscle strength were relatively well predicted, with explained variances ranging between 42% and 84%. Without prior outcome,  $VO_2$ peak and respiratory function were poorly predicted. The residual SD, which is a measure of the prediction accuracy of a model, described the association between the predicted and the observed outcome.<sup>157</sup> As an example, Figure 1 illustrates the association between the predicted and the observed discharge POpeak. This Bland-Altman plot depicts the distribution of the differences between the predicted and the observed discharge POpeak and shows that in this population the predicted POpeak may be 20W higher or lower than the observed outcome.

### Discussion

Prognostic models revealed that prior physical capacity made an important contribution to the prediction of outcome, and identified both modifiable and not modifiable predictors. Considering the proportion of variance explained by our models, they compared favorably to those predicting physical capacity in different populations,<sup>119,159</sup> and to those predicting other health measures after SCI.<sup>20,30,160,161</sup> Our models are unique, because they were based on longitudinal data, reflected different aspects of physical capacity and included subject, lifestyle, and lesion related predictors. However, this hampers a constructive comparison with models predicting other outcome measures, in different populations with other predictors for several reasons. First, inherent differences in the predictability between outcome measures complicate a comparison of the accuracy of the models. Physical capacity is an objective measure of health following SCI, therefore, it may be more predictable than subjective measures such as life satisfaction.<sup>17,30</sup> At the same time, physical capacity may be influenced by natural recovery, training, motivation and concurrent disease,<sup>17,149</sup> which could make it less predictable than the ability to walk or quality of life, which are both stable during different phases of recovery.<sup>150,10</sup> Second, unique predictors (like level of lesion or stabilization of the spine) apply to those with SCI but are not generalizable.<sup>9,119</sup> Third, because a prognosis indicates that a predictor precedes the predicted in time, models based on cross-sectional data were not prognostic, further complicating comparisons.<sup>36,141,154</sup> Knowledge on the predicted physical capacity may provide objective information of a patient's expected functional status, which helps to evaluate symptoms, the effect of interventions and the necessity to refer the patient to postdischarge facilities.<sup>121,154</sup>

One must remember, however, that because the accuracy of prognostic models is limited,<sup>41,154</sup> they may only complement clinical expertise, and caution is needed when using them as decision tools in rehabilitation practice. Below we discuss the possible value of these prognostic models as informative guidelines, and in identifying those at risk of a low physical capacity.

#### Prior Physical Capacity as a Predictor of Outcome at Follow-Up

In agreement with others, prior outcome was the strongest predictor in our study,<sup>30,34</sup> which alone explained 51% to 82% of the variance. The finding that characteristics like level of lesion and sex were of little prognostic value (in addition to prior physical capacity), suggests that prior outcome may encapsulate the variance caused by these characteristics. Because knowledge of prior physical capacity seems essential in making an accurate prediction, and because it may summarize the influence of several subject, lifestyle, and lesion characteristics, we recommend that physical capacity be monitored regularly starting at an early phase of rehabilitation. Despite the strong association with prior physical capacity, a previous study fortunately showed that physical capacity did change during active rehabilitation and that some aspects continued to change after discharge.<sup>151</sup> In agreement with this study,<sup>151</sup> the current models suggested that changes in physical capacity partly depended on characteristics that were of prognostic value besides prior physical capacity, such as age, level of the lesion, and also employment, for example. Additionally, factors like rehabilitation or follow-up programs could have influenced changes in physical capacity. Our subjects followed a regular rehabilitation program, but unfortunately we did not have valid information on its contents. However, it would not have been rational to include

rehabilitation characteristics, unknown at the start of active rehabilitation, in these prognostic models.

Without knowledge of prior physical capacity, its prediction proved challenging and models were less accurate. This difference was especially noticeable for the models predicting outcome 1 year after discharge. The predictive value of the investigated characteristics was less strong 1 year after discharge. It could be that after discharge other factors, not currently investigated (eg, housing, community care, social support) were associated with physical capacity. Therefore, future investigation of the influence of these factors seems necessary. In the meantime, we recommend that their possible influence receives attention at follow-up visits.<sup>162</sup>

#### Prognostic Models for Different Aspects of Physical Capacity

This study revealed relatively good prognostic models for the POpeak and arm muscle strength, which explained more variance as compared with those predicting Vo2peak and respiratory function. The POpeak may be influenced by a person's skills and efficiency. Therefore, it is more closely related to functioning than VO<sub>2</sub>peak.<sup>25</sup> VO<sub>2</sub>peak is related to organ function and, therefore, more influenced by complications.<sup>163</sup> Respiratory function was less predictable for several possible reasons. First, respiratory function was expressed as a percentage of the expected flow in an age, sex, and height-matched able-bodied person. Therefore, these characteristics were corrected for and could not contribute to the explained variance. Second, other not currently investigated predictors, such as comorbidity, may have contributed specifically to the variance in respiratory function. Third, relatively large differences in respiratory function between subjects may overshadow the associations with predictors within subjects.<sup>41</sup> Because a small proportion in variance of respiratory function was explained, the clinical relevance of these models is limited. However, the prediction of POpeak and muscle strength could complement clinical professional knowledge when informing patients on the outcome and when planning training and follow-up programs.

#### Prognostic Factors

Table 3 illustrated different levels of physical capacity in several subject groups, but these differences need to be interpreted with care, and multivariate models were needed to establish associations. The one-to-one associations found may have been caused by chance alone, because many differences were tested. Furthermore, these associations did not correct for possible confounders. As expected, the multivariate models showed that level of lesion and sex were important predictors of physical capacity.<sup>12,119,124,153</sup> However, these models also identified other predictors, which

made small, but significant, contributions to the explained variance, some of which are modifiable.

Not surprisingly, smoking predicted a poor respiratory function, but the negative association found with the peak power output suggests that giving up smoking may lead to a more favorable physical capacity.<sup>45,124</sup> In the able-bodied population, moderate alcohol consumption is suggested to be cardioprotective, whereas the consumption of large quantities will have adverse health effects. A predictive effect of alcohol was not established in our multivariate models, and the influence of alcohol intake remains undecided.<sup>45</sup> Another study showed that alcohol use following SCI was similar to that of the able-bodied population, but revealed an increased incidence in alcohol abuse, which may negatively affect health.<sup>164</sup> Therefore, future research needs to focus on a more detailed investigation of the effect of alcohol use and abuse after SCI.

Unexpectedly, BMI was a positive predictor of physical capacity. However, as with other indicators of body composition, like skinfolds and abdominal circumference, it is difficult to interpret the BMI after  $SCI.^{165}$  In an early phase of recovery, a low body mass may be attributed to a catabolic state following injury, concurrent pathology, or complications. Therefore, a low BMI may be negatively associated with physical capacity. In contrast, during a later phase of recovery, relative inactivity and altered metabolism and fat distribution may cause an increase in BMI to be negatively associated with physical capacity.<sup>46,159</sup> These mechanisms did not influence the association between BMI and physical capacity 1 year after discharge, but may become apparent at a longer follow-up period. Surprisingly, hours of sports did not contribute to the prediction of physical capacity.<sup>9,12</sup> It may be that sport participation changed after the lesion. Prospective data on sport participation, its barriers and its association with physical capacity are needed.<sup>166</sup> Contrary to others, our models revealed a positive predictive effect of both conservative and surgical stabilization of the spine.<sup>34,155</sup> Perhaps the immobilization period is shorter after surgical stabilization, resulting in an early start of rehabilitation and insignificant deconditioning.<sup>3</sup> Because stabilization seemed to have no prognostic value after discharge, its influence may be restricted to an early phase of recovery.

Not modifiable predictors are valuable because they may identify those at risk of a low physical capacity. In the present study, more hours of paid employment before injury positively predicted physical capacity especially after discharge, which may correspond with findings that those previously employed had a shorter inpatient stay and that physical capacity was associated with employment after injury.<sup>59,147,160</sup> Perhaps those previously employed have greater physical or mental learning skills, which may enable them to adapt to wheelchair skills more easily, or may provide them with opportunities for re-education. Therefore, those previously employed may

return to a more active lifestyle through being employed after discharge. Further prospective data are needed to establish these proposed explanations.<sup>167</sup> Although no significant colinearity existed between sex and hours of employment, and although both predictors contributed to the explained variance simultaneously, it is difficult to rule out that the predictive effect of employment was mediated through a covariate like sex.<sup>147,157</sup> However, the promising outcome for those in the habit of working indicates the importance of vocational rehabilitation, and clinicians need to create and evaluate opportunities to return to employment after discharge from rehabilitation.<sup>167</sup>

Foreign subjects tended to have a lower level of different aspects of physical capacity. Not being Dutch negatively predicted strength and its recovery during the year after discharge. Studies of the Model Spinal Cord Injury Systems showed similar associations with ethnicity, and their authors attributed the differences to ethnic minorities being at an economic disadvantageous position and having less access to resources.<sup>160,168</sup> In this study, foreign subjects originated from different countries, had diverse cultural backgrounds and their comprehension of the Dutch language differed. Therefore, our results cannot be generalized, and are difficult to compare with the abovementioned American studies.

The positive predictive value of education has been reported previously.<sup>34,47</sup> However, this seems contradictory to the finding that the less educated had a shorter inpatient rehabilitation, as reported by Eastwood.<sup>160</sup> Hypothesizing, it might be that those with lower level of education did physically more demanding work before injury, which may have made it easier to learn (wheelchair) skills. As a result, they may have been discharged sooner than those previously holding, for example, desk jobs. In contrast, those with a higher level of education may have had more knowledge of how (and means) to lead a healthy lifestyle, which may have positively contributed to their physical capacity. Education is important in the prevention of complications and in the maintenance or improvement of physical and functional capacity. Therefore, we recommend that clinicians give complete information to all patients, not just to those who show an interest in a healthy lifestyle.

#### Limitations

Some methodological limitations need to be considered. First, the selected population limits the generalizability of our data. We included patients who required a wheelchair for daily functioning, which explains why a relatively large number of subjects with a complete lesion were included.<sup>158</sup> However, in comparison with other studies, which focused mainly on young male athletes with a paraplegia,<sup>12,163</sup> our data seem to better reflect outcome in a general population with SCI.<sup>158</sup> Ideally the difference between the

predicted and the observed outcome is as small as possible. This brings us to a second limitation, because, as is illustrated in figure 1, we found discrepancies between these values.<sup>157</sup> These discrepancies could be even larger if the models were crossvalidated, that is, if we compared the predicted outcome based on these models with the observed outcome in another SCI population. Therefore, future cross-validation with other populations seems essential before the clinical validity of the prognostic models can be established. The third consideration is the variance that could not be explained by the models. Factors not currently investigated (such as motivation, complications, rehabilitation or training, living conditions and social support) may all have contributed to the level of physical capacity. In summary, some caution is needed these models are used as clinical guidelines. We do not recommend these models as solitary clinical decision tools; they should only be used in combination with clinical expertise and knowledge of the individual patient.

#### Conclusions

This study revealed prognostic models for several aspects of physical capacity, which showed that POpeak and arm muscle strength were relatively well predicted in comparison with Vo<sub>2</sub>peak and respiratory function. Because the prediction of physical capacity at an early phase of recovery is recommended. Furthermore, the systematic monitoring of physical capacity may help set realistic targets during and after inpatient rehabilitation. Besides estimating prognoses, the presented models provide insight into the possible positive predictive effect of not smoking, employment and stabilization of the spine. Caution is warranted concerning the accuracy of the models, but in combination with clinical expertise and knowledge of the individual patient, they may provide meaningful information of a patient's expected physical outcome, which could help to postdischarge facilities.

*Suppliers* (a) Bonte Technology BV, Ampèrestraat 25b, 8013 PT Zwolle; (b) Jaeger Toennies, Nikkelstraat 2, 4823 AB Breda; (c) MicroFET; Biometrics Europe BV, Kabelstraat 11, 1322 AD Almere, The Netherlands, (d) Version 12.0.01; SPSS Inc, 233 S Wacker Dr, 11th Fl, Chicago, IL 60606.

5

# Complications following spinal cord injury: occurrence and risk-factors in a longitudinal study during and after inpatient rehabilitation

JA Haisma, LH van der Woude, HJ Stam, MP Bergen, TA Sluis, MW Post, JB Bussmann In: *Journal of Rehabilitation Medicine* 2007; 39: 393-398

#### Abstract

*Objective:* To assess the occurrence and risk factors of complications following a spinal cord injury during and after inpatient rehabilitation.

*Methods:* Multicentre prospective study, which included persons with a spinal cord injury admitted to specialized rehabilitation centers. Complications were registered at the start of active rehabilitation (n=212), three months later (n=143), at discharge (n=191) and one year after discharge (n=143).

*Results:* Multilevel random coefficient analyses revealed that complications were common following a spinal cord injury. Most subjects reported neurogenic and musculoskeletal pain, or had spasticity at each assessment. The occurrence of complications did not change significantly. Therefore, during the year after discharge complications remained common. For example, urinary tract infections and pressure sores affected 49% and 36% of the population, respectively. The degree of pain decreased, whereas the degree of spasticity increased significantly during inpatient rehabilitation. Overall, an increase in age, increase in body mass index, a traumatic lesion, a tetraplegia and a complete lesion exposed to a risk of complications.

*Conclusion:* Complications are common following a spinal cord injury. They need specific attention after discharge from inpatient rehabilitation and within subpopulations.

#### Introduction

A spinal cord injury (SCI) is often followed by complications, which add to the detrimental effect that loss of motor, sensory and autonomic function have on a person's health, social participation and quality of life.<sup>11,47,162,169</sup> There is some debate on the operational definition of complications, on whether complications are strictly causally related to the SCI and on how to distinguish complications from co-morbidity.<sup>19</sup> However, a condition may only be defined as being a complication if it has a chronological relation with the SCI, i.e. the SCI needs to precede the condition. Additionally, although this condition may also occur in the general population, one assumes those with SCI to be at increased risk.<sup>19</sup>

The range of conditions following an SCI can be categorized into neurological consequences and secondary complications. Neurological consequences result from the injury itself, following interruption and decentralization of the nervous system, and may be regarded as sequelae to the injury.<sup>170</sup> Examples are neurogenic pain or spasticity; the latter being part of the upper motor neuron syndrome.<sup>44,171</sup> Secondary complications follow the ensuing loss of pulmonary function, loss of bladder control or reduced mobility. Examples are pulmonary infections and pressure sores.<sup>11,172,173</sup> Neurological consequences and secondary complications behave similarly; they often require, and react to, treatment, and they affect the rehabilitation process. Therefore, we refer to them all as complications of SCI.

Complications have a considerable impact on those with SCI. A high incidence in complications is associated with a lower level of health-related aspects such as physical capacity, activities and functional outcome.<sup>11,13,47,162</sup> Complications may interfere with the start of active rehabilitation, can form a disappointing set-back during rehabilitation, and frequently lead to rehospitalization.<sup>14,44,174</sup> Additionally, complications are an important cause of mortality following an SCI.<sup>19,126</sup> In order to optimize the individual rehabilitation process and outcome, it is important to timely predict and prevent complications or to recognize and treat complications them.<sup>19</sup>

Previous studies have investigated complications following SCI and their riskfactors. They illustrated the association between subject and lesion characteristics and the occurrence of complications.<sup>11,42,171,175</sup> They also showed that the diversity and the occurrence of complications changed over time.<sup>11,172,173,176</sup> However, these studies have limitations. Firstly, most studies were cross-sectional, whereas a longitudinal study in which data are collected prospectively could better establish which factors actually lead to complications. Secondly, in most studies only 1 or a few complications were assessed, whereas the simultaneous investigation of a range of complications will give insight into their diversity, and into how their occurrences compare. Therefore, we have formulated the following three research questions: (1) What is the occurrence of complications (pain, spasticity, hypotension, autonomic dysreflexia, pressure sores, urinary tract infections, pulmonary infections, venous thromboembolism, edema, heterotopic ossification, other cardiovascular disease and other musculoskeletal complaints) in subjects with SCI during and after inpatient rehabilitation? (2) What is the degree of pain and spasticity and does this change during and after inpatient rehabilitation? (3) What are risk-factors of these complications?

#### Method

The present study was part of the Dutch research program on the restoration of mobility in persons with an SCI. Subjects admitted to 1 of the 8 participating rehabilitation centers between May 2000 and September 2003 were included if they met eligibility criteria. During their first period of inpatient rehabilitation, they were eligible for inclusion if they were between 18 and 65 years of age, were wheelchair-dependent, had sufficient comprehension of the Dutch language to understand the purpose of the study and did not have a progressive disease or a psychiatric condition interfering with constructive participation.

#### Design

Subjects were assessed 4 times according to a standardized protocol: at the start of active inpatient rehabilitation, defined as the moment when the subject was able to sit in a wheelchair for  $\geq$  3 h (t1), 3 months later (t2), at discharge (t3), and 1 year after discharge from inpatient rehabilitation (t4). If the subject was discharged within 1 month after t2, the assessment at t2 was considered a 'discharge' assessment, and was included in the analyses of t3. The Medical Ethics Committee approved the protocol and prior to participation all subjects gave a written informed consent.

#### Complications

Based on the subject's history and medical chart, the physicians registered each complication on a standardized list. At t1, complications then present or that had occurred since admission to rehabilitation were registered. At t2 and t3, complications then present or that had occurred since t1 and t2, respectively, were registered. At t4, complications then present or that had occurred since discharge were registered. The occurrence of a complication was registered as follows: 0 = no complication; 1 = presence or history of this complication. Other cardiovascular disease and musculoskeletal
complaints were grouped, and included conditions such as myocardial disease and bursitis occurring after the lesion, respectively.

If the subject reported having pain the research assistants completed a standardized list addressing the nature of the pain. We defined musculoskeletal pain as nociceptive pain originating from bone, joint or muscle structures following trauma or overuse.<sup>44</sup> Thirteen locations (on the upper and lower limbs, the neck and the back) were assigned severity scores with a 5-point Likert-scale (1-5; ranging from 'not severe' to 'very severe'), and frequency scores (1-3; ranging from 'once a week or less' to 'more than 3 times a week'). We defined neurogenic pain as at level or below level pain originating from spinal cord ischemia or trauma.<sup>44</sup> Hence, nine neurogenic pain characteristics (other pain, numbness, itching, tingling, cold, warm, perspiration, girdle zone pain and phantom feeling) were assigned frequency and severity scores. A sumscore was made for the product (severity × frequency) of these locations or pain characteristics. Therefore, sumscores ranging from 1 to 195, or from 1 to 135 could be attained for musculoskeletal or neurogenic pain, respectively.

Additionally, the research assistant determined the presence of spasticity, defined as the velocity dependent increase in muscle tone combined with exaggerated reflexes, through a direct standardized examination.<sup>44</sup> The left and right hip adductors, knee flexors and extensors, ankle extensors, and elbow flexors and extensors were examined. In the presence of spasticity, each muscle group was assigned severity scores ranging from 1 to 3 ('1': catch; '2': clonus < 5 beats; '3': clonus  $\geq$  5 beats). These scores were summated, which gave a sumscore ranging from 1 to 36.

# Risk-factors

The associations with the following potential risk-factors were assessed: age, gender, smoking status (smoker vs. non-smoker), body mass index (BMI: body mass in kilograms divided by height in meters squared; kg/m<sup>2</sup>), the cause (traumatic vs. non-traumatic), the level and the completeness of the lesion. Although some smokers refrained from smoking during inpatient rehabilitation, nearly all resumed smoking after discharge. Therefore, subjects were defined as being a smoker if they smoked prior to the injury. Tetraplegia was defined as a lesion at or above the first thoracic segment, and paraplegia as a lesion below the first thoracic segment. A complete lesion was diagnosed in the absence of sensory or motor function in the sacral segments, i.e. American Spinal Injury Association (ASIA) category A. An incomplete lesion was defined as ASIA categories B, C or D.<sup>1</sup>

#### Statistics

Random coefficient analyses (MlwiN version 1.1; Centre for Multilevel Modelling, Institute of Education, London, United Kingdom) were used to estimate the occurrence of complications, the change in the degree of pain and spasticity, and the association with risk-factors.<sup>138,139</sup> The longitudinal aspect of the study is enhanced by the fact that the analyses can be done with missing values and varying group composition. Therefore, all assessments can be included in the analyses, which gives a more accurate description, compared with repeated measurements analysis of variance, for example, of complications and their risk-factors at each assessment time.

*Complications*. A logistic random coefficient model was made for the occurrence of each complication. Time was included in the model as a set of 3 dummy variables (each with their own regression coefficient). The discharge assessment was chosen as their reference and was estimated by the intercept. The occurrence of a complication during the other intervals was estimated as follows: 1/ {1 + exp[ - (intercept + regression coefficient)]}.<sup>138</sup> The sumscores for pain and spasticity were estimated with similar models for continuous outcome variables. Again, time was modelled as 3 dummy variables, and the score at discharge was estimated by the intercept. The sumscores at the other assessments were calculated by adding the intercept to the regression coefficient of the dummy variable.

*Risk-factors*. All risk-factors were simultaneously added to the previously described models. With these multivariate models, we corrected for the contribution of each risk-factor. The regression coefficients for the risk-factors were converted to odds ratio's (ORs); OR = exp[regression coefficient]. An OR of 1 indicated there was no association with this particular variable, whereas an OR <1 indicated a decreased risk, and an OR >1 indicated an increased risk of this complication in the presence of the risk-factor. For the continuous outcome variables, the regression coefficient indicated the difference in the sumscore associated with the difference in the risk-factor of 1 unit.

# Results

Table 1 gives the descriptive characteristics of the subjects. Subjects were lost to followup for several reasons: 9 subjects died, 5 moved abroad, 26 refused further participation, 7 developed psychiatric disease or a progressive condition, 5 were irretraceable and 3 subjects dropped-out for unknown reasons. Forty-four subjects were discharged within 1 month after t2, and therefore, no data were included for them at t2. The mean (SD) duration between injury and admission to rehabilitation was 44 (43) days.

Characteristic	start	3 months	discharge	1 yr after discharge
Subjects (n)	212	143	191	143
Age (years)	40 (14)	41 (14)	41 (14)	41 (14)
Gender (% male)	74	75	73	73
Body mass index (kg/m <sup>2</sup> )	22.7 (3.8)	23.1 (4.0)	23.4 (4.0)	24.5 (4.5)
Smoker (% smoker)	44	44	45	48
Days since previous assessment*	49 (44)	103 (33)	162 (125)	397 (57)
Cause (% traumatic)	74	78	75	76
Level (% tetraplegia)	41	48	40	35
Completeness (% complete)	45	44	49	50

Table 1 Descriptive subject characteristics

\* At the start days since admission to rehabilitation are given.

#### Complications

Figure 1 shows the estimated occurrence of a complication during each interval. All data are derived from random coefficient analyses. Most subjects reported neurogenic and musculoskeletal pain, or had spasticity at each assessment. Common secondary complications were urinary tract infections and pressure sores, reported by 47% and 36% of the population at t1, respectively. Like most complications, they remained common during the year after discharge, occurring in 49% and 36% of the population at t4, respectively. Additionally, Figure 1 shows the estimated degree of pain and spasticity at each assessment time. The degree of pain decreased, whereas the degree of spasticity increased significantly during inpatient rehabilitation.

### Risk-factors

Table 2 gives the ORs for the association between the risk-factors and the occurrence of a complication. An increase in age, tetraplegia and completeness of the lesion were the most frequently identified risk-factors. Those with a high BMI and those with a traumatic lesion were at increased risk of several complications; the largest effect was seen for cardiovascular disease. The risk associated with gender or smoking varied for different complications.

Table 3 gives the regression coefficients for the association between the degree of pain or spasticity and the risk-factors. Overall, an increase in age was associated with an increase in pain. Men and those with a traumatic lesion had a higher degree of spasticity. Those with a tetraplegia and those with an incomplete lesion reported more musculoskeletal pain and showed a higher degree of spasticity.



**Figure 1** The estimated occurrence of complications & reported degree of pain or spasticity: random coefficient modelling. \* significant difference ( $p \le 0.05$ ) between the sumscore at start and that at discharge. <sup>†</sup>sumscores determined in those who reported pain or had spasticity. Abbreviations: AD = autonomic dysreflexia; UTI = urinarytract infection, DVT/PE = thromboembolism; HO = heterotopic ossification; CVD = other cardiovascular disease

#### Discussion

### Complications

The present study showed that the occurrence of a range of complications was high both during and after inpatient rehabilitation. Most of our findings coincided with

				0,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
	Potential Risk Facto	DIS					
Complication	Age (yrs)	Gender*	Smoking <sup>*</sup>	BMI (kg/m <sup>2</sup> )	Cause*	Level*	Complete <sup>*</sup>
Neur. pain	1.02 (0.99–1.05)	0.52 (0.23-1.21)	1.20 (0.63–2.27)	0.99 (0.91–1.08)	0.99 (0.41-2.41)	0.86 (0.44–1.70)	0.57 (0.29–1.11)
Spasticity	<b>0.97</b> (0.96–0.99)	<b>2.53</b> (1.60–3.98)	1.42 (0.93–2.16)	<b>1.11</b> (1.04–1.18)	1.07 (0.64–1.79)	<b>0.13</b> (0.08–0.23)	0.95 (0.61–1.48)
Hypotension	0.99 (0.97–1.02)	0.82 (0.43–1.54)	0.73 (0.41–1.29)	0.97 (0.90–1.04)	1.38 (0.59–3.25)	<b>0.09</b> (0.05–0.18)	<b>2.44</b> (1.32–4.52)
Autonomic dysreflexia	0.98 (0.95–1.00)	1.69 (0.81-3.52)	1.21 (0.68–2.15)	1.07 (0.99–1.15)	0.94 (0.40-2.21)	<b>0.14</b> (0.07–0.27)	<b>2.36</b> (1.26–4.42)
Pressure sore	1.01 (1.00–1.03)	1.08 (0.72–1.63)	1.34 (0.93–1.93)	0.98 (0.93–1.03)	1.08 (0.67–1.74)	<b>0.53</b> (0.36–0.78)	<b>1.73</b> (1.17–2.56)
Urinary tract infection	1.00 (0.99–1.02)	0.87 (0.60-1.28)	0.78 (0.56–1.11)	0.97 (0.93–1.02)	<b>1.59</b> (1.02–2.47)	0.52 (0.36-0.75)	<b>1.81</b> (1.26–2.60)
Pulmonary infection	<b>1.05</b> (1.02–1.08)	1.03 (0.51-2.07)	1.49 (0.79–2.80)	0.93 (0.85–1.01)	1.04 (0.45-2.42)	0.26 (0.13-0.53)	$3.53^{\dagger}$ (1.74–7.17)
Thrombo embolism	1.01 (0.97–1.06)	0.69 (0.22–2.12)	2.10 (0.72–6.11)	0.92 (0.79–1.08)	**	1.35 (0.42-4.32)	1.84 (0.59–5.74)
Edema	<b>1.04</b> (1.03–1.06)	0.81 (0.53-1.24)	<b>1.62</b> (1.09–2.41)	<b>1.10</b> (1.04–1.16)	<b>2.05</b> (1.23–3.43)	1.09 (0.72–1.65)	1.46 (0.97–2.21)
Heterotopic ossification	0.98 (0.96–1.01)	<b>11.38</b> (2.74–47.32)	<b>0.50</b> (0.27–0.91)	1.01 (0.93–1.09)	0.79 (0.35–1.78)	0.80 (0.42–1.49)	<b>2.45</b> (1.29–4.67)
Cardiovasc. disease	<b>1.05</b> (1.01–1.10)	0.71 (0.24-2.13)	1.28 (0.41-3.96)	<b>1.28</b> (1.11–1.46)	<b>5.06</b> (1.25–20.46)	0.90 (0.31–2.63)	0.60 (0.19–1.94)
Musc. pain	1.00 (0.98–1.02)	<b>0.63</b> (0.40–0.99)	0.78 (0.53–1.14)	<b>1.07</b> (1.01–1.13)	<b>1.92</b> (1.17–3.15)	0.66 (0.43–1.00)	0.73 (0.48–1.09)
Other musc. complaints	<b>1.02</b> (1.00–1.05)	1.11 (0.56–2.19)	1.06 (0.58–1.93)	0.96 (0.89–1.04)	1.33 (0.60–2.96)	0.86 (0.45–1.64)	1.53 (0.81–2.89)
Odds ratio's (CI) are give	en; significant associa	tions ( $p \le 0.05$ ) printed	d bold. <sup>*</sup> Gender: me	n =1, women =0; smo	king: smoker =1, non-	-smoker =0; cause: tra	aumatic =1,

Table 2 Association with risk-factors: multivariate logistic random coefficient modeling

non-traumatic =0; level: paraplegia =1, tetraplegia =0; completeness: complete =1, incomplete =0. <sup>‡</sup>Cause' could not be modeled; in this population all subjects with thromboembolism had a traumatic lesion.

Degree	Age (yrs)	Gender*	Smoking*	BMI** (kg/m <sup>2</sup> )	Cause*	Level®	Complete*
neur. pain	<b>0.12</b> (0.01–0.23)	-2.20 (-5.30–0.89)	-2.29 (-5.10–0.52)	0.27 (-0.11–0.64)	-1.28 (-4.86–2.30)	<b>-9.84</b> (-12.816.88)	2.86 (-0.10–5.82)
spasticity	0.00 (-0.03–0.03)	<b>1.02</b> (0.13–1.90)	0.64 (-0.11–1.38)	0.02 (-0.09–0.12)	<b>2.33</b> (1.35–3.32)	-1.07 (-1.86-0.28)	-2.21 (-3.02—1.39)
musc.pain†	<b>0.14</b> (0.01–0.28)	-2.33 (-6.08–1.43)	-2.13 (-5.58–1.32)	0.06 (-0.39–0.50)	-1.24 (-5.76–3.27)	- <b>10.03†</b> (-13.61–6.44)	- <b>4.60</b> (-8.26-0.94)

Table 3 Determinants of degree of pain or spasticity: multivariate random coefficient modeling

Regression coefficients (CI) are given; significant associations ( $p \le 0.05$ ) are printed bold. \*gender: men =1, women =0; smoking: smoker =1, non-smoker =0; cause: traumatic =1, non-traumatic =0; level: paraplegia = 1, tetraplegia = 0; completeness: complete =1, incomplete =0. \*\* BMI = body mass index.

previous studies, but inconsistencies may be attributed to the variation in both the selected population and in the design. Excluding those with progressive disease or those over the age of 65, for example, will have influenced our findings. We registered complications based on clinical symptoms often confirmed by additional examination, whereas some others searched for pathology in the absence of clinical symptoms. Furthermore, our longitudinal design will have revealed different information on the association with risk-factors than previous cross-sectional data did.

Although pain and spasticity were common, understanding their impact remains challenging. The reported degree of pain is influenced by both psychosocial and physical factors, which interfere with the interpretation of changes over time and the associations with risk-factors.<sup>177,178</sup> It remains to say, that when pain is experienced, it does often necessitate intervention. Besides increased stretch reflexes, spasticity encompasses increased muscle tone, involuntary movements and primitive reflexes. Lack of effective measurement techniques makes the quantification of all components of spasticity difficult.<sup>44</sup> Furthermore, in contrast to pain, the degree of spasticity is not necessarily related to individual functional complaints or to the required treatment. Because pain and spasticity are common sequelae to SCI, which may require intervention, it is important to reach consensus on how to monitor their clinical impact.

Several mechanisms may explain why complications are common after discharge from inpatient rehabilitation.<sup>14,172,173</sup> After discharge, both the demanding activities of daily living (ADLs) and the reduction in structured training moments could make subjects susceptible to complications associated with overuse.<sup>11</sup> Additionally, many subjects have less guidance or peer-control in self-care.<sup>179</sup> The higher demands of ADLs, together with the reduced feed-back, could make subjects less conscientious towards preventive measures, like skin-checks and bladder management. The active involvement of the spouse may contribute to the prevention of complications, and further analyses of our data showed a tendency for those living alone to be at increased risk. Although the interval evaluated after discharge is long as compared to the intervals during rehabilitation, complications remain a common problem. This

should warrant decision makers to invest in effective follow-up programs.<sup>14,162,179</sup> We think it an important task for rehabilitation medicine to strife for the long-term prevention of complications.

#### Risk-factors

This study indicated subpopulations inherently at risk of complications. Level and completeness of the lesion determine the extent of pareses and loss of respiratory and autonomic function,<sup>11,22,151</sup> which are inevitably associated with the risk of complications.<sup>11,173,178</sup> The physiological age-related decline in cardiorespiratory function and mobility, leave the elderly susceptible to complications.<sup>172,180</sup> Concurrent extraspinal injury may be an additional risk-factor.<sup>2</sup> Although concurrent injury was poorly registered in the present study, further investigation revealed it seemed more common in traumatic lesions. This may partly explain why those with a traumatic lesion were at risk of some complications. However, a standardized registration of concurrent injury is needed to establish these relations. A predisposition of men may partly explain their risk of heterotopic ossification.<sup>174</sup> Additionally, it could be that in men symptoms were more apparent, or more often attributed to heterotopic ossification. Although type of lesion, age and gender cannot be influenced, their identification as important risk-factors allows us to target screening and prevention programs.

Besides these unchangeable risk-factors, the effect of some life-style risks was investigated. An increase in BMI exposed to a risk of several complications. However, the interpretation of the BMI is difficult in those with SCI, especially during the early phase of rehabilitation.<sup>165</sup> Changes in body weight may be attributed to the increase in upper body muscle mass, to the decrease in muscle mass of the lower limbs, to the collection of extra-vascular fluids and to an increase in fat-mass.<sup>165</sup> The associations with other modifiable life-style risks, like alcohol consumption and inactivity, remain ambiguous.<sup>164,181</sup> We anticipate that the cumulative damaging effect of smoking, alcohol consumption, dietary intake and inactivity becomes more evident at a later phase post-injury. Therefore, the relation between lifestyle exposure and complications needs to be investigated in a longitudinal study with a longer follow-up period.

#### Limitations

The present study design needs some consideration. One should bear in mind that we investigated medical complications, whereas SCI also has psychosocial complications beyond the scope of this investigation. In general, data on the occurrence of complications during an interval, do not inform us on their severity or duration. Because the

occurrence of a complication may have delayed discharge, the population assessed three months into active rehabilitation was probably more prone to complications. The assessed population and the varying intervals should be considered when interpreting results. However, random coefficient analyses have allowed us to include all present subjects at each assessment time and this has provided more realistic data on the occurrence of complications during each interval. Our longitudinal design gave insight into the timing of complications, and contributed to the understanding of the nature of risk-factors during different phases of rehabilitation.

In conclusion, complications are common following an SCI and subpopulations are at increased risk. Educational programs for patients and their relatives need to focus on the prevention and early recognition of complications. Structural follow-up visits after inpatient rehabilitation need to be implemented at specialized rehabilitation centers. Besides addressing functional outcome and social participation, these visits need to also focus on complications. Only through the timely prevention, surveillance and treatment of complications can their impact be reduced, and can the individual rehabilitation outcome be optimized.

*Acknowledgements* This article was presented in part at the 8th International Neurotrauma Symposium, 21-25 May 2006, Rotterdam, The Netherlands, and at the 45th ISCoS Annual Scientific Meeting, 25-28 June 2006, Boston USA. The Health Research and Development Council of The Netherlands (grant no. 1435.0003; 1435.0025) supported this study.

# 6

# Physical fitness in people with spinal cord injury: the association with complications and duration of rehabilitation

JA Haisma, JB Bussmann, HJ Stam, TA Sluis, MP Bergen, MW Post, AJ Dallmeijer, LH van der Woude In: *Clinical Rebabilitation* 2007; 21: 932-940

# Abstract

*Objective:* To assess the association between physical fitness and its recovery over time on the one hand, and complications and duration of phases of rehabilitation on the other.

*Methods:* Persons with a spinal cord injury admitted for initial rehabilitation were assessed at the start of active rehabilitation (n= 110), three months later (n= 92), at discharge (n= 137) and one year after discharge from inpatient rehabilitation (n= 91). Physical fitness was defined as aerobic capacity, determined at each occasion by the peak oxygen uptake (peak  $Vo_2$ ; l/min) and the peak power output (peakPO, Watt). Spasticity, pain, the occurrence of complications, bedrest and duration of rehabilitation were potential determinants of physical fitness. Spasticity, musculoskeletal and neurogenic pain were determined on each occasion ('1'= present; '0'= absent). During inpatient rehabilitation, complications (urinary tract infection, pulmonary infection or pressure sore) and bed rest were registered ('1'= complication; '0'= no complications, and '1'= bed rest; '0'= no bed rest). Complications and bed rest occurring during the year after discharge were registered similarly.

*Results:* Multilevel random coefficient analyses revealed associations in multivariate models ( $p \le 0.05$ ). The peak oxygen uptake was negatively associated with complications after discharge. The recovery of peak power output over time was negatively associated with bed rest and spasticity. Both physical fitness and its recovery were negatively associated with the duration of active rehabilitation.

*Conclusion:* Results suggest that reducing complications, spasticity or bed rest may improve fitness. A longer duration of active rehabilitation is not associated with an increase in physical fitness.

### Introduction

Physical fitness, defined as the ability of the cardiorespiratory and musculoskeletal systems to attain a certain level of activity, is reduced following a spinal cord injury (SCI) for several reasons.<sup>17</sup> The level of fitness is established by the aerobic capacity, for which the peak oxygen uptake (peak  $Vo_2$ ; l/min) is the golden standard,<sup>9,132</sup> and by the maximal maintainable workload, which is determined by the peak power output (peak PO; Watt).<sup>25</sup> Interruption of the spinal cord causes muscle pareses and autonomic disturbance, which both limit the level of fitness.<sup>9,153,182</sup> Additionally, the occurrence of complications is associated with a lower level of fitness, worsened by complications, which increases the vulnerability to more complications, further reducing fitness, and so on.<sup>46</sup>

Physical fitness is a meaningful health indicator, not only because of its proposed relation to complications, but also because it is associated with functioning and quality of life.<sup>9,183</sup> Additionally, the level of fitness is modifiable; it changes over time and can give an indication of rehabilitation outcome.<sup>49,151</sup> Longitudinal data show that the recovery of fitness over time is associated with age, gender and level of the lesion.<sup>49,141,151</sup> Unfortunately, these data do not indicate modifiable determinants and, therefore, do not provide tools to improve rehabilitation practice.

During spinal cord rehabilitation effort is put into the treatment of complications and into the improvement of the physical fitness. Therefore, the duration of this rehabilitation not only depends on the severity of the injury, but also on the occurrence of complications and on rehabilitation goals.<sup>13,160</sup> Complications are a common problem interfering with constructive rehabilitation,<sup>14,176</sup> and although fitness improves during rehabilitation, it remains low.<sup>12,151</sup> Prospective data, collected during and after inpatient rehabilitation, are needed to get insight into whether the effort put into a rehabilitation program is accompanied by an improved outcome.<sup>13</sup> The complex relations between fitness, the occurrence of complications and the duration of rehabilitation need to be unravelled. This will provide tools to optimize individual outcome and rehabilitation strategies. Therefore, we formulated the following research question: what is the association between physical fitness and its recovery over time during and after inpatient rehabilitation on the one hand, and the occurrence of complications and the duration of different functional phases of rehabilitation on the other, in wheelchair dependent subjects with SCI?

## Methods

#### Subjects

The present study was part of the Dutch research program on the recovery of mobility following SCI. Between 2000 and 2005, all consecutive patients with a traumatic or non-traumatic SCI admitted to one of the eight participating rehabilitation centres were included if they were wheelchair dependent, between 18 and 65 years of age, if they had no contra-indications for exercise, and could complete an exercise test on at least one occasion.<sup>132,151</sup> For a detailed description of the in- and exclusion criteria, design and procedure, we refer the reader to a previous publication.<sup>151</sup> The Medical Ethics Committee approved the experimental protocol, and prior to participation all subjects gave a written informed consent.

# Design

In this prospective cohort study, subjects were assessed four times: at the start of active rehabilitation, when the subject was able to sit in a wheelchair more than three hours at a time (t1), three months later (t2), at discharge (t3) and one year after discharge from inpatient rehabilitation (t4).<sup>151</sup> At each rehabilitation centre, one trained research assistant was responsible for the data collection. Date of birth and gender were noted at the start of the study, and on each occasion the age was calculated. At each occasion, a physician determined level and completeness of the lesion. Paraplegia (score = 1) was defined as a lesion below the first thoracic segment; tetraplegia (score = 0) as a lesion at or above the first thoracic segment. A complete lesion (score = 1) was defined as motor complete, i.e. ASIA category A or B; an incomplete lesion (score = 0) as ASIA category C or D.<sup>1</sup>

# Procedure

*The level of physical fitness.* To establish the peak oxygen uptake and peak power output, subjects performed a graded maximal wheelchair exercise test on a motor-driven treadmill, which is a valid method to determine physical fitness following SCL.<sup>17</sup> A detailed description of the procedure for this population was given previously.<sup>151</sup> The equipment was calibrated prior to each assessment. The peak oxygen uptake (l/min) was defined as the highest value of oxygen consumption recorded during 30 seconds. The peak power output (Watts) was defined as the workload at the highest inclination that the subject could maintain for at least 30 seconds.<sup>27</sup> The research manual provided normative values as a guideline.<sup>12</sup> For the peak oxygen uptake, the expected mean

score ranged from 0.61 to 0.83 l/min for those with a tetraplegia, and from 1.34 to 1.44 l/min for those with a paraplegia. For the peak power output, the expected mean score ranged from 14 to 24 Watts (W) for those with a tetraplegia, and from 54 to 75 W for those with a paraplegia.

Complications and duration of rehabilitation. On each occasion, the research assistant determined spasticity (physical examination) and pain (reported at interview). Spasticity, defined as the velocity dependent increase in muscle tone combined with exaggerated reflexes,<sup>44,184</sup> was assessed at the upper and lower limbs. Starting position of subject and research assistant, and the direction and speed of movements were standardized. Movements were tested during a slow passive stretch, and then spasticity was determined by a fast passive stretch (< 1 second). Spasticity was scored as follows: '0' = no spasticity, '1' = spasticity and catch during stretch (i.e. sudden increase in muscle tone blocking further movement), 2' = clonus (involuntary rhythmic muscle contraction) less than five beats, '3' = clonus five beats or more. To facilitate analyses, the score was recoded into one binary score: '0' was recoded as 'no spasticity' (score = 0); '1', '2' and '3' were recoded as 'spasticity present' (score = 1). Self-report information on musculoskeletal pain (due to trauma, inflammation or overuse) at the upper and lower extremities, and the neck and back was registered. For thirteen locations, the presence and severity of pain was scored on a 5-point Likert scale. These scores were recoded into one binary score for the presence of pain (score = 1, regardless of location, severity or frequency), or the overall absence of musculoskeletal pain (score = 0). Similarly, neurogenic pain (due to spinal cord or nerve root trauma) was registered if a numb, tingling, burning, phantom, hot or cold feeling was reported.<sup>44</sup> A binary score was created for the presence (score = 1), or the overall absence (score = 0) of neurogenic pain.

At each occasion, the physician used medical charts and self-report information to determine whether a complication (urinary tract infection, pulmonary infection or pressure sore) had occurred during the previous period, and whether this had resulted in bed rest. Binary scores were created for both complications and bed rest. Either the subject experienced at least one complication during rehabilitation (score = 1), or the subject had no complications (score = 0). Subsequently, this resulted in bed rest during rehabilitation (score = 1), or not (score = 0). Similarly, both complications and bed rest after inpatient rehabilitation were assigned binary scores.

The inpatient rehabilitation period was divided into two functional phases for each subject. The first, the acute rehabilitation, was defined as the number of days between admission to rehabilitation, and the moment when the subject could sit in a wheelchair more than three hours at a time (i.e. the start of active rehabilitation). The second, the active rehabilitation, was defined as the number of days between the start of active rehabilitation, and discharge from inpatient rehabilitation.

#### Statistics

Random coefficient analysis (MlwiN version 1.1; Centre for Multilevel Modelling, Institute of Education, London, UK) was used to assess the association between physical fitness and its recovery over time on the one hand, and complications and duration of rehabilitation on the other.<sup>138 139</sup> Separate models were made for the peak oxygen uptake and the peak power output. In a previous publication we have described a similar modelling procedure.<sup>151</sup> Because we investigated fitness and its recovery, time was consistently modelled as three variables; the regression coefficients of these variables each represented the difference between the level of fitness at one occasion, and the fitness at discharge. Then, we expanded these models. Firstly, to assess which variables were associated with the level of fitness, complication and rehabilitationrelated variables were added to the models one-by-one. Secondly, to assess whether these variables were associated with the recovery of fitness over time, interaction terms (between time and the variable) were added to the models.

Subsequently, all variables and interaction terms found to be associated with fitness in one-to-one relations ( $p \le 0.10$ ) were simultaneously entered into a multivariate model. By alternately removing non-significant variables and repeating the analyses, a final model with significantly associated variables remained ( $p \le 0.05$ ). To permit valid assessments, both components of a significant interaction term had to be included in the model. Furthermore, three interactions with time had to be modelled, even if an association proved significant over one time interval. We tested and corrected for the confounding effect of gender and level or completeness of the lesion by adding these variables to the models.

# Results

The study-population consisted of 160 subjects. Of these subjects, we assessed 110 at the start, 92 three months later, 137 at discharge, and 91 one year after discharge from inpatient rehabilitation. Table 1 gives their characteristics and physical fitness at each occasion. The mean (SD) duration of acute rehabilitation was 47 (42) days, and the mean active rehabilitation was 93 (113) days. In 29 subjects, no assessment was done at the start of active rehabilitation, in which case subjects were included three months later. Fifty-four subjects were discharged within three months after admission; hence they did not perform an assessment three months into active rehabilitation. There were other reasons for not collecting data at a specific occasion or for dropping out: four subjects died, seven moved abroad or could not be contacted, two developed psychological problems interfering with participation, seven became independent of

Characteristic and physical	start	three months	discharge	year after discharge
fitness	n = 110	n = 92	n = 137	n = 91
age (years)	40 (14)	38 (14)	39 (14)	37 (13)
men	81 (74%)	72 (78%)	101 (74%)	69 (76%)
tetraplegia	24 (22%)	23 (25%)	37 (27%)	16 (18%)
complete lesion	69 (63%)	64 (70%)	85 (65%)	63 (72%)
traumatic lesion	80 (73%)	69 (76%)	103 (76%)	70 (78%)
spasticity	62 (56%)	70 (76%)	92 (67%)	61 (67%)
neurogenic pain	97 (88%)	83 (90%)	122 (89%)	83 (91%)
musculoskeletal pain	83 (76%)	69 (75%)	92 (67%)	56 (62%)
complications total	not applicable	not applicable	130 (81%) <sup>a</sup>	57 (63%)
pressure sore			73 (46%)	32 (35%)
urinary tract infection			110 (69%)	40 (44%)
pulmonary infection			19 (12%)	5 (6%)
bed rest total	not applicable	not applicable	67 (42%) <sup>a</sup>	21 (23%)
due to pressure sore			45 (28%)	13 (14%)
due to urinary tract infection			29 (18%)	6 (7%)
due to pulmonary infection			12 (8%)	3 (3%)
peakVo2 <sup>b</sup> (l/min)	1.01 (0.29)	1.13 (0.41)	1.21 (0.44)	1.31 (0.51)
peak PO <sup>c</sup> (Watts)	29.7 (15.5)	36.9 (20.8)	40.8 (22.8)	47.8 (25.0)

Table 1 Descriptive statistics of subject characteristics and physical fitness

Values are mean (SD), or number of subjects with this characteristic (%). <sup>a</sup> complications or bed rest during inpatient rehabilitation; subjects assessed at start, three months and/or discharge are summarized and n = 160; <sup>b</sup> peak  $V_{0_2} =$  peak oxygen uptake; <sup>c</sup> peak PO = peak power output.

a wheelchair, twelve refused further participation, and in two instances there were technical problems. Reasons for not collecting data in this population have previously been described in more detail.<sup>151</sup> Overall, 22 (14%) subjects were assessed once, 46 (29%) twice, 53 (33%) on three occasions, and 39 (24%) were assessed on all four occasions.

#### The association with complications and duration of rehabilitation

Table 2 gives the results of the final multivariate models. There were one-to-one associations between the level of peak oxygen uptake and complications during rehabilitation, between the recovery of peak oxygen uptake over time and spasticity, and between the recovery of peak power output and pain ( $p \le 0.10$ ). However, these associations were not significant (p > 0.05) in multivariate models. Below Table 2 examples are given for the interpretation of the regression coefficients.

The level of the peak oxygen uptake was negatively associated with complications after discharge. Both the level of peak oxygen uptake and its recovery over time were negatively associated with the duration of active rehabilitation. The level of peak power output was also negatively associated with active rehabilitation. Figure 1 illustrates the association between the peak power output and bed rest.

#### Chapter 6

Independent variable	peak VO2 (l/min)	peak PO (Watt)
discharge	1.16 (0.97, 1.35)	50 (41, 59)
$t1 - t3 (0/1)^a$	-0.24 (-0.37, -0.12)	-14 (-18, -10) <sup>a</sup>
$t2 - t3 (0/1)^a$	-0.03 (-0.17, 0.11)	-4 (-8, 0)
$t4 - t3 (0/1)^a$	0.20 (0.07, 0.33)	<b>7 (3, 12)</b> <sup>a</sup>
spasticity (0/1)	n.e. <sup>b</sup>	0 (-4, 5)
bed rest during rehabilitation (0/1)	n.e.	-5 (-11, 3)
complication after rehabilitation (0/1) <sup>c</sup>	-0.12 (-0.21, -0.03) <sup>c</sup>	n.e.
active rehabilitation (days) <sup>d</sup>	-10-10 <sup>-4</sup> (-16-10 <sup>-4</sup> , -4-10 <sup>-4</sup> )	-0.06 (-0.08, -0.03)
bed rest* t1 - t3 (0/1) <sup>e</sup>	n.e.	5 (0, 10)
bed rest* t2 - t3 (0/1)	n.e.	-1 (-4, 3)
bed rest* t4 - t3 (0/1)	n.e.	-3 (-8, 3)
spasticity* t1 - t3 (0/1)	n.s <sup>f</sup>	-3 (-8, 2)
spasticity* t2 - t3 (0/1)	n.s.	-1 (-5, 4)
spasticity* t4 - t3 (0/1)	n.s.	-7 (-12, -1)
active rehabilitation* t1 – t3 (days)	2.10-4 (- 5.10-4, 8.10-4)	n.s.
active rehabilitation * t2 – t3 (days)	-2.10-4 (-8.10-4, 5.10-4)	n.s.
active rehabilitation * t4 – t3 (days)	-8·10 <sup>-4</sup> (-14·10 <sup>-4</sup> , -2·10 <sup>-4</sup> )	n.s.

**Table 2** The multivariate association between the (recovery of) physical fitness, and complications & duration of rehabilitation: multilevel random coefficient analyses

All results are regression coefficients (95% CI) in the final multivariate model. Significant associations ( $p \le 0.05$ ) are printed bold. The regression coefficient indicates the difference in fitness associated with an increase in the independent variable of one unit. Interaction terms are joined by asterisks (\*). A correction was made for gender, level and completeness of the lesion.

<sup>a</sup> time modelled as three variables; t1–t3 indicates the difference between fitness at the start and that at discharge, t2–t3 indicates the difference between fitness at three months and that at discharge; t4–t3 indicates the difference between fitness one year after discharge and that at discharge. For the peak power output, for example, the regression coefficient for t1–t3<sup>a</sup> indicates the difference between start and discharge was –14 Watts (W), i.e. during active rehabilitation it improved 14 W; the regression coefficient for t4– t3<sup>a</sup> indicates that after inpatient rehabilitation it improved 7 W.

<sup>b</sup> n.e. = variable not entered, because no significant one-to-one association with fitness (p > 0.10).

<sup>c</sup> if a complication occurred (variable takes on a '1') the peak oxygen uptake was 0.12 l/min lower than in the absence of a complication (variable takes on a '0').

 $^{\rm d}$  increase in duration of 1 day was associated with a difference in peak oxygen uptake of  $-10\cdot10^{-4}$  l/min, and peak power output of -0.06 W.

<sup>e</sup> the interaction term, bed rest\*t1– t3, for example, indicates that the difference between start and discharge peak power output was 5 W smaller in the presence of bed rest, i.e. there was less recovery during rehabilitation.

<sup>f</sup> n.s. = variable not significant and removed from multivariate model (p > 0.05).

During active rehabilitation, the recovery of the peak power output was negatively associated with bed rest. After discharge, this recovery was negatively associated with spasticity.



**Figure 1** The association between the peak power output (peak PO; Watt) and bed rest (0/1), when all other variables in the final model remain constant. During active rehabilitation, i.e. between start and discharge, the recovery of peak PO over time was negatively associated with bed rest.

## Discussion

This study revealed associations between physical fitness, and both complications and the duration of rehabilitation. Although causality cannot be established, the associations found describe a combination of longitudinal and cross-sectional relations and, therefore, provide more valid information than cross-sectional data alone.<sup>138</sup> The overall level of fitness was low, which may partly be attributed to the fact that this study did not limit itself to men or wheelchair athletes or persons with a paraplegia.<sup>12,153</sup> However, it does illustrate the necessity to utilize every opportunity to improve fitness following SCI. Our results may help clinicians more adequately target prevention and treatment strategies and, therefore, interpretations and suggestions for clinical practice are presented.

### The association with complications

Unexpectedly, physical fitness and its recovery over time were not associated with complications during inpatient rehabilitation in our multivariate models. However, the recovery of fitness was negatively associated with bed rest, which suggests that not the actual occurrence of complications, but its consequences, i.e. bed rest, are of influence. Our findings seem to correspond with studies in healthy able bodied subjects, which show the deconditioning effect of bed rest.<sup>185,186</sup> Clinicians need to be aware of the possible negative associations with bed rest. We would recommend them to carefully minimize bed rest, prevent deconditioning if bed rest is required, and to anticipate a lower level of fitness after bed rest by gradually resuming activities. Studies have shown the positive effect of exercise programs in the able bodied population,<sup>187,188</sup> but rehabilitation research needs to investigate how deconditioning during bed rest can be prevented in those with SCI.

A possible increase in the severity of spasticity over time could explain the negative association found with fitness after discharge.<sup>189</sup> Both contractures and sudden contractions of muscles interfere with functioning in most patients.<sup>190</sup> Besides hampering the performance during the actual exercise test, spasticity could lead to a reduced activity, which may eventually affect fitness.<sup>190</sup> Considering this long-term effect, clinicians should assess spasticity at follow-up visits. Although the quantification of spasticity remains challenging,<sup>191,192</sup> clinicians need to assess spasticity, investigate its interference with activities of daily life, and initiate treatment if necessary.<sup>44,190</sup>

Results seem to correspond with the postulated downward spiral caused by the association between physical fitness and complications. Its association with bed rest and spasticity may negatively affect the inherently low fitness. The association found with complications after discharge suggest that those with a lower physical fitness are at risk of complications, further reducing fitness, and so on. However, whether complications affect fitness, or whether a low physical fitness predisposes to complications after discharge remains undecided. Although cause and effect cannot be established with the present design, we recommend structural follow-up visits, not only to address functioning and community reintegration, but also to determine the level of fitness, the presence of spasticity and the occurrence of complications.

#### The association with duration of rehabilitation

Goals change during rehabilitation, and this may explain the negative association found between fitness and the duration of active rehabilitation. Initially treatment will focus on medical stability, psychological adaptation, strengthening and aerobic training, which will contribute to an increase in fitness.<sup>193</sup> When a certain level of fitness has been reached, rehabilitation may become dominated by functional goals, such as community ambulation or vocational rehabilitation; aerobic training will become a secondary goal.<sup>193</sup> As a consequence, one would expect an increase in duration of rehabilitation to be positively associated with functioning, but others failed to establish this relation.<sup>13,155</sup> These seemingly remarkable findings call for a more detailed evaluation of the effect of rehabilitation. Therefore, we assessed the association between fitness and the hours of exercise-related therapy by means of Spearman correlations. This exploratory investigation revealed no association for the absolute level of fitness, or the relative recovery of fitness over time. Because of the negative association with the duration of active rehabilitation, we recommend aerobic training and activity programs to be continued to prevent deconditioning during later phases of rehabilitation.

Subsequently, results could imply that the duration of rehabilitation was not determined by a training goal, but by arranging post-discharge facilities, such as community care, home adaptations or assistive equipment.<sup>13,179</sup> If this is the case, we believe that the realization of these facilities needs to be improved. First, because

patients need to adjust to life outside the rehabilitation centre, and secondly, because budgetary problems in health services ask for an efficient organization.<sup>160</sup> With early insight into the expected individual outcome, procedures could be started sooner, sometimes even prior to inpatient rehabilitation. Therefore, the early evaluation of patients' predicted outcome and needs, and the structural monitoring of their progress is important.<sup>121</sup> Additionally, we see it as a task for decision-makers, and financers and suppliers of equipment to reduce procedural difficulties.

The current design and analytic procedure need some consideration. One year after discharge the population comprised fewer subjects with a tetraplegia. This may have influenced results, but the analyses allowed us to include longitudinal data with a different group composition.<sup>138</sup> Since the peak power output is more influenced by the technique of wheelchair-propulsion, results for the peak oxygen uptake and peak power output differed to some extent.<sup>25</sup> Those not restricted by bed rest or spasticity, may adapt to wheelchair skills more easily and, as a consequence, these variables may be associated with the peak power output. Besides the described difficulties in defining and objectively quantifying spasticity or pain,<sup>44,191,192</sup> their binary scores need to be considered when interpreting data. This recoding enabled us to investigate their association with fitness, whilst regarding the duration of rehabilitation, in one multivariate model. This statistical choice may have resulted in the loss of information on the association with the degree of spasticity, pain or complications. However, secondary analyses revealed that, although one-to-one associations with more detailed variables (i.e. variables not recoded into binary scores) existed, these were not significant in multivariate models. Therefore, recoding had no effect on the conclusive findings of this study. In the present study, the association between fitness and given (medical) therapy, guidance or education could unfortunately not be evaluated. Although difficult to quantify, the thorough registration of rehabilitation characteristics and their conditions and intensities, will facilitate an investigation of the effect of rehabilitation in future.

# Clinical messages

- The level of physical fitness is negatively associated with the occurrence of complications after discharge
- The recovery of physical fitness over time is negatively associated with bed rest and spasticity
- Fitness and its recovery over time are negatively associated with the duration of active rehabilitation

7

Physical independence and health-related functional status following a spinal cord injury: a prospective study on the association with physical capacity and complications

> JA Haisma, JB Bussmann, HJ Stam, MP Bergen, TA Sluis, HJ van den Berg-Emons, MW Post, LH van der Woude Submitted to: *Journal of Rehabilitation Medicine*

# Abstract

*Objective:* To determine changes in physical independence during and after inpatient rehabilitation for spinal cord injury (SCI). To investigate the associations between physical capacity and complications on the one hand, and (changes in) physical independence and health related functional status on the other.

*Methods:* Physical independence, physical capacity and complications were determined at the start of rehabilitation (n = 176), 3 months later (n= 124), at discharge (n= 160), and 1 year after discharge from inpatient rehabilitation (n= 133). The Functional Independence Measure was computed as FIMmotor, and as a dichotomous rating for physical independence, whereby a subject was categorized as independent if FIMmotor was  $\geq$  78 (score = 1), and as dependent if FIMmotor was < 78 (score = 0). One year after discharge, physical and social dimensions of health-related functional status (Sickness Impact Profile 68) were determined.

*Results:* Multilevel random coefficient analyses revealed that physical independence improved during inpatient rehabilitation. Hereafter FIMmotor stabilized, whereas the probability of independence significantly improved from 0.33 to 0.55. Physical independence was positively associated with peak power output, and negatively associated with complications and a complete lesion. Multiple regression analyses showed these factors also predicted 16 to 29% of health-related functional status.

*Conclusion:* Physical independence improves during rehabilitation, and physical independence and health related functional status are associated with peak power output and complications.

### Introduction

A spinal cord injury (SCI) affects previously undemanding daily activities. Patients may suddenly rely on the assistance of others for seemingly straightforward tasks, and may need to adapt to the prospect of physical dependence.<sup>6</sup> Although regaining physical independence is perhaps not the sole purpose of rehabilitation, independent functioning does form a key to being active and socially involved, and may contribute to a sense of control over one's life.<sup>3,53</sup> Clinicians require insight into changes in physical independence to adequately inform patients and relatives about the expected level of activity and participation, to evaluate progress, and to determine the necessity of post-discharge care and assistance.<sup>194</sup>

Aspects of activity and participation following SCI can be determined validly in different ways.<sup>195,196</sup> The Functional Independence Measure (FIM) establishes independence in the performance of several activities; its items on mobility and selfcare (FIMmotor) focus on physical independence.<sup>195,197</sup> The abbreviated Sickness Impact Profile (SIP68) determines health-related functional status; its physical and social dimensions establish whether a health condition affects activity and participation.<sup>196,198</sup> The FIMmotor and the SIP68 complement one another as outcome measures of spinal cord rehabilitation: the FIMmotor provides more detailed information on the actual level of independence in mobility and self-care, and the SIP68 covers higher levels of activities, like difficulties in doing handwork, it determines participation, and it is less limited by a ceiling effect than the FIMmotor.<sup>196,198</sup>

As is illustrated in the International Classification of Functioning, Disability and Health (ICF), body function, health condition, and personal or environmental factors each influence activity and participation.<sup>15,52</sup> In previous retrospective studies, patients indicated that their level of activity and their participation were threatened by physical limitations, such as inadequate strength or endurance capacity, and by poor health condition and complications related to pain, incontinence and ageing.<sup>37,199</sup> In agreement with these subjective findings, cross-sectional studies showed that physical capacity, complications, spasticity, type of lesion and age were associated with activity limitations.<sup>13,20,33,133</sup> Factors like type of injury and age cannot be changed. However, the level of physical capacity is low in those with SCI and training can improve this.<sup>200</sup> Furthermore, complications are common and rehabilitation programs aim to minimize their consequences.<sup>3,201</sup> Cross-sectional associations have been established, but if a prospective study can confirm the relation between changes in physical independence on the one hand, and changes in physical capacity and complications on the other, this will help clinicians to adequately target training, treatment and screening programs. Furthermore, if physical capacity and complications at discharge from inpatient rehabilitation predict outcome at follow-up, this suggests the beneficial effect of training, treatment and screening programs on the level of physical independence and health-related functional status.

This prospective study of a cohort of wheelchair-bound subjects with SCI had 3 objectives. The first objective was to determine changes in physical independence (FIMmotor and the probability of independence) over time. Therefore, we established physical independence on 3 occasions during inpatient rehabilitation, and once again, 1 year after discharge. The second objective was to investigate the association between changes in physical independence on the one hand, and changes in physical capacity and complications on the other. Therefore, we also determined physical capacity and complications on each occasion. Because the likelihood of independence after discharge provides a clear message about prognoses, and because the prediction of health-related functional status becomes especially meaningful after discharge when higher levels of activities and participation are required, the third objective was to determine whether discharge physical capacity and complications predicted the probability of independence, and health-related functional status. Therefore, besides independence, physical capacity and complications, we determined health-related functional status 1 year after discharge from inpatient rehabilitation.

### Method

This study was part of the Dutch research program on 'Physical strain, work capacity and mechanism of restoration of mobility in the rehabilitation of persons with SCI'. Between July 2000 and November 2005 the program included patients with SCI admitted to 1 of the 8 participating Dutch rehabilitation centres. At their initial inpatient rehabilitation, patients were eligible to participate in the research program if they were between 18 and 65 years of age, were wheelchair-bound, had sufficient comprehension of the Dutch language to understand the purpose of the study, and did not have a progressive disease or psychiatric condition interfering with constructive participation. For patients to be included in our study sample, both their physical independence and physical capacity had to be assessed at least once. Reasons for excluding patients from the maximal exercise test, e.g. cardiovascular disease or musculoskeletal complaints, have been described previously.<sup>151</sup> The Medical Ethics Committee approved the protocol, and prior to participation all patients gave a written informed consent.

#### Procedure

Subjects were assessed according to a standardized protocol on 4 occasions: at the start of active inpatient rehabilitation, defined as the moment when the subject could sit in a wheelchair for at least 3 hours at a time (t1), 3 months into active rehabilitation (t2), at discharge (t3), and 1 year after discharge from inpatient rehabilitation (t4). On each assessment occasion, we determined physical independence, physical capacity and complications. Additionally, health-related functional status was assessed at t4. At each rehabilitation centre, an experienced rehabilitation professional was appointed as research assistant and was, therefore, responsible for the data collection. To optimize the quality of the collected data, research assistants were trained prior to and during the program, and data were collected and registered according to a strict protocol.

#### Physical independence and health-related functional status

At each assessment (t1-t4), the Dutch FIM (version 5.0) was completed.<sup>195</sup> The FIMmotor score was computed as the sum score of 13 FIM items on mobility and self-care.<sup>197</sup> In addition, a dichotomous rating for physical independence was derived from the FIM. This meant that if the FIMmotor was  $\geq$  78, a subject was categorized as independent (binary score = 1), and if FIMmotor was < 78, a subject was categorized as dependent on the assistance of others (binary score = 0). The percentage of the study sample categorized as independent reflects the probability of independence. Health-related functional status was determined 1 year after discharge (t4) with the SIP68. For each SIP68 item subjects indicated whether their health condition currently limited this activity (score = 1) or not (score = 0). The physical dimension of the SIP68 was the sum score of items on somatic autonomy and mobility control (range 0 to 29). The social dimension was the sum score, the more health-related functional status was affected. Because subjects were wheelchair-bound, items on mobility were recoded as recommended by Post et al.<sup>198</sup>

#### Independent variables

*Physical capacity*. At each assessment, physical capacity was established by the endurance capacity (peak oxygen uptake and peak power output) and muscle strength. The endurance capacity was determined during a graded peak wheelchair exercise test on a motor-driven treadmill. The peak oxygen uptake (L/min) was defined as the highest average oxygen consumption recorded during 30 seconds, and the peak power output (Watts) as the workload at the highest inclination that the subject could maintain for at least 30 seconds.<sup>27</sup> Muscle strength was tested in 10 muscle groups (shoulder abductors, internal and external rotators and elbow flexors and extensors).

For those muscle groups with a strength of at least grade 3 during manual muscle testing, strength was determined with handheld dynamometry.<sup>119,202</sup> Subsequently, arm muscle strength was derived from a summation of the strength of these 10 muscle groups (Newton).<sup>151</sup>

Complications. At each assessment, the occurrence of complications and the presence of pain and spasticity were registered.<sup>201</sup> The physician used medical charts and self-reported information to determine whether a complication (urinary tract infection, pulmonary infection or pressure sore) had occurred since admission (reported at the start, t1) or since the previous assessment (reported at t2, t3 or t4, respectively), and whether this had resulted in bed rest during this period. Either the subject had at least 1 complication since the previous assessment (score = 1), or the subject had no complications (score = 0). Subsequently, these complications resulted in bed rest (score = 1) or not (score = 0). Self-reported information on musculoskeletal pain at the arms, legs, neck and back was registered with a 5-point Likert scale. These scores were recoded into the presence of pain (score = 1), or the overall absence of musculoskeletal pain (score = 0). Similarly, neurogenic pain (numb, tingling, burning, phantom, hot or cold feeling) was recorded as being present (score = 1) or absent (score =0).44 Spasticity was assessed at the arms and legs by a fast passive stretch.<sup>184</sup> The presence of spasticity (score = 1), or the overall absence of spasticity (score = 0) was registered.

*Subject and lesion characteristics.* Age, level and completeness of the lesion were registered at each occasion. Tetraplegia (score = 0) was defined as a lesion at or above the first thoracic segment, and paraplegia (score = 1) as a lesion below the first thoracic segment. A complete lesion was defined as motor complete, i.e. American Spinal Injury Association (ASIA) category A or B, and an incomplete lesion as ASIA category C or D.<sup>1</sup>

#### Statistics

Descriptive statistics (SPSS version 12.0.01) summarized subject and lesion characteristics and outcome at each occasion (Table 1). The total study sample included 182 subjects, of whom 176 were included at the start (t1), 124 were assessed at 3 months (t2), 160 were assessed at discharge (t3), and 133 were assessed 1 year after discharge (t4). At the start of active rehabilitation, the mean (SD) time since injury was 88 (61) days, and at discharge from inpatient rehabilitation, 290 (140) days had passed since injury.

# Changes in physical independence over time

Random coefficient analyses (MlwiN version 1.1; Centre for Multilevel Modelling, Institute of Education, London, UK) determined changes in physical independence over time, whereby we took into account the repeated assessments within 1 subject

	Start	Three months	Discharge	One yr after discharge
Characteristic *	n = 176	n = 124	n = 160	n = 133
Age (years)	40 (14)	41 (14)	40 (14)	41 (14)
Men (%)*	76 (133)	77 (95)	74 (118)	72 (96)
Paraplegia (%)	69 (121)	67 (83)	70 (111)	74 (97)
Complete lesion (%)†	67 (117)	48 (58)	48 (75)	53 (68)
Peak oxygen uptake (L/min)	1.03 (0.36)	1.15 (0.42)	1.22 (0.44)	1.32 (0.51)
Peak power output (watts)	31 (18)	37 (21)	41 (23)	48 (25)
Arm muscle strength (Newton)	1547 (533)	1678 (554)	1805 (538)	1864 (608)
Complications (%) <sup>‡</sup>	62 (109)	71 (87)	52 (82)	66 (87)
Bed rest (%)¶	24 (42)	35 (43)	17 (27)	30 (39)
Musculoskeletal pain (%)	79 (139)	77 (95)	68 (109)	64 (85)
Spasticity (%)	66 (102)	74 (89)	68 (104)	70 (86)
Neurogenic pain (%)	92 (161)	93 (114)	90 (144)	94 (124)
Independence and functional sta	itus			
FIMmotor**	44 (18)	58 (20)	69 (17)	69 (19)
Independent (%)	5 (8)	17 (21)	34 (55)	45 (60)
Physical SIP68**	n.a.	n.a.	n.a.	12 (7)
Social SIP68**	n.a.	n.a.	n.a.	6 (4)

Table 1 Descriptive statistics of subject characteristics and physical independence

\* Mean (SD) or percentage of study sample (N) with this characteristic is given. <sup>†</sup>Complete lesion: motor complete, i.e. American Spinal Injury Association (ASIA) category A or B; <sup>†</sup>Complications: those who had at least 1 complication since admission (reported at the start of active rehabilitation) or since previous occasion; <sup>¶</sup>Bed rest: those who had bed rest for these complications. <sup>\*</sup>FIMmotor = level of functional independence (range 13 to 91); physical SIP68 = physical dimension of Sickness Impact Profile 68 (range 0 to 29); social SIP68 = social dimension of SIP68 (range 0 to 22); n.a. = not applicable, because not determined at these occasions.

and within 1 rehabilitation centre.<sup>138</sup> One model was made for changes in FIMmotor, and a separate model was made for changes in the probability of independence. In the FIMmotor model, the intercept represented discharge FIMmotor (t3). Three dummy variables were modelled (for t1, t2 and t4) and their regression coefficients each represented the difference between the FIMmotor at t1, t2 or t4, and that at discharge (t3), respectively. Therefore, these regression coefficients and their confidence intervals reflected changes in FIMmotor over time and the significance of these changes.

The model for the probability of independence was a logistic random coefficient model. Therefore, the regression coefficients for the dummy variables for t1, t2 and t4 indicated the ratio between the odds of independence on 1 occasion (t1, t2 or t4, respectively) and these odds at discharge (t3). The probability of independence at each occasion was then calculated as follows: probability =  $1/{1 + exp[-(regression equation)]}$ .<sup>157</sup> This probability gave insight into the likelihood a subject would be independent at this specific occasion. The regression coefficients and their confi-

dence intervals reflected changes in the probability of independence over time and the significance of these changes.

*Changes in physical independence over time in association with changes in physical capacity and complications* 

We determined which independent variables were associated with changes in physical independence. To determine the significance of their individual influence, the independent variables were first separately added to the model for FIMmotor, and also to the model for the probability of independence. The following independent variables were tested: peak oxygen uptake (L/min), peak power output (Watts), muscle strength (Newton), complications (0/1), bed rest (0/1), spasticity (0/1), musculoskeletal pain (0/1) and neurogenic pain (0/1). To investigate whether the associations changed over time, we modeled 3 interaction terms (between the dummy variables for t1, t2 and t4, and the independent variable, respectively) for each independent variable and determined their significance. Because age, gender, level and completeness of the lesion could have influenced physical independence, or could have confounded the associations on the other, the associations with age, gender, level and completeness of the lesion were tested.

Based on all individual associations, we made 1 multivariate model for FIMmotor and 1 for the probability of independence. Because the peak power output, peak oxygen uptake and muscle strength are related measures of physical capacity, possibly leading to collinearity,<sup>151,157</sup> the multivariate models included the measure of physical capacity with the strongest association with physical independence. Therefore, in this case, peak power output and all other significant independent variables and interaction terms ( $p \le 0.10$ ) were simultaneously added to the respective models. Age, gender, level and completeness of the lesion were treated in a similar way as the other independent variables. We narrowed down the multivariate models by alternately removing a non-significant (p > 0.05) independent variable, and rerunning the analyses.

This resulted in 1 final multivariate model with significantly associated variables for the FIMmotor, and another with significantly associated variables for the probability of independence. The regression coefficients in the FIMmotor model indicated an increase in FIMmotor associated with an increase in the independent variable of 1 unit. In the model for the probability of independence, the oddsratio (OR= exp[regression coefficient]) reflected the increase in the odds of independence associated with an increase in the independence associated with an increase in the independence of 1 unit.<sup>138</sup>

# *Physical capacity and complications as predictors of the probability of independence and the SIP68*

We made 1 multiple logistic regression model to establish whether discharge physical capacity and complications predicted the probability of independence 1 year after discharge (t4), and 2 multiple regression models to establish whether these factors predicted the social and physical dimensions of health-related functional status at t4, respectively (SPSS version 12.0.01). Because the multilevel random coefficient analyses provided knowledge on associations with the probability of independence, and because this probability was determined both at discharge (t3) and 1 year later (t4), whereas the SIP68 was only determined at t4, different modelling procedures were chosen for the probability of independence and for the SIP68.

A hierarchical entering procedure was used for the probability of independence model. First, we entered independency status at discharge (0/1), which was followed by other independent variables (t3) of significant value in the individual multilevel models. Therefore, the discharge peak power output (Watts), completeness of the lesion (0/1), complications (0/1) and musculoskeletal pain (0/1) at discharge were entered following the stepwise-forward procedure. From the resultant multivariate model, the odds of independence in the presence of a variable were calculated.

For the 2 models that predicted physical and social dimensions of the SIP68, respectively, a stepwise-forward procedure was chosen. The following independent variables were tested: discharge peak oxygen uptake, peak power output, muscle strength, complications, bed rest, musculoskeletal pain, neurogenic pain, spasticity, gender, age, level and completeness of the lesion. Variables were cross-tabulated, and if independent variables were individually associated (correlation coefficient > 0.80), the variable with the strongest association with health-related functional status was included for further analyses.<sup>157</sup> Of the measures of physical capacity, the peak power output was again most strongly associated with both domains of health-related functional status and was, therefore, included in the modelling procedure. In the resultant final multivariate models, the regression coefficients indicated an increase in score associated with an increase in the independent variable of 1 unit.

### Results

#### Changes in physical independence over time

Figures 1 and 2 illustrate changes in physical independence over time as estimated with multilevel random coefficient models. They indicate that the FIMmotor increased during inpatient rehabilitation, but did not change significantly after discharge. How-



**Figure 1** Functional Independence Measure (FIMmotor) during and after inpatient rehabilitation; p-values for significance of the difference between FIMmotor at this occasion and that at discharge as estimated with the multilevel random coefficient model.



**Figure 2** Probability of independence during and after inpatient rehabilitation; p-values for significance of the difference between the odds of independence at this occasion and those at discharge as estimated with the multilevel logistic random coefficient model

ever, the probability of independence increased significantly both during inpatient rehabilitation and during the year after discharge.

# *Changes in physical independence over time in association with physical capacity and complications*

Table 2 gives the individual associations between physical independence and separate independent variables. Both measures of physical independence were associated with peak oxygen uptake, peak power output and complications. The FIMmotor was additionally associated with muscle strength, whereas the probability of independence was also associated with musculoskeletal pain. The final multivariate multilevel model for FIMmotor showed a positive association with peak power output and a negative association with complications (Table 3). An increase in peak power output of 10 Watt was associated with an increase in FIMmotor of nearly 4 points (10 Watt multiplied by 0.36), and if complications had occurred the FIMmotor was nearly 3 points less. The

		-		
	Functional In Measure (FIM	dependence motor score)	Probability of	independence
Independent variable	Regression coefficient	Confidence Interval	Oddsratio	Confidence Interval
Peak oxygen uptake (L/min)	14.3	10.6 to 18.1	9.5	3.2 to 27.9
Peak power output (Watt)	0.40	0.32 to 0.48	1.1	1.0 to 1.1
Muscle strength (Newton)	0.02	0.01 to 0.02	1.0	1.0 to 1.0
Complications (0/1)	-2.15	-4.25 to -0.05	0.3	0.1 to 0.6
Bed rest (0/1)	-0.04	-2.31 to 2.23	0.8	0.3 to 1.8
Spasticity (0/1)	-1.44	-4.44 to 1.56	0.7	0.3 to 2.0
Musculoskeletal pain (0/1)	-1.08	-3.37 to 1.21	0.4	0.2 to 0.9
Gender (1 = men; 0 = women)	5.23	-10.0 to 20.5	1.51	0.44 to 5.18
Age (yrs)	-0.09	-0.25 to 0.07	0.99	0.95 to 1.03
Level $(1 = PP; 0 = TP)$	9.04	6.69 to 11.4	1.13	0.37 to 3.45
Completeness (1 = complete.; 0 = incomplete)	-0.97	-3.64 to 1.70	0.16	0.06 to 0.44

 Table 2 Individual associations with physical independence

Independent variables were separately added to the model, which included 3 dummy variables for changes over time. Significant associations are printed bold. The regression coefficient and oddsratio indicate the increase in FIMmotor score or in the odds of independence, associated with an increase in the independent variable of 1 unit. Complications, bedrest, pain or spasticity were either present (0) or not (1). Abbreviations: PP= paraplegia; TP= tetraplegia.

**Table 3** Functional Independence Measure (FIMmotor) in association with physical capacity and complications: multivariate multilevel random coefficient analyses

Association with	FIMmotor score		
Independent variable	Regression coefficient <sup>*</sup>	95% Confidence interval	p-value
Discharge; t3	58.82	53.02 to 64.62	< 0.01
Start minus discharge; t1 - t3 (0/1)	-22.18	-27.16 to -17.20	< 0.01
3 months minus discharge; t2 - t3 (0/1)	-12.20	-17.30 to -7.10	< 0.01
1 yr post discharge minus discharge; t4 - t3 (0/1)	2.40	-2.93 to 7.73	0.37
POpeak (Watt)	0.36	0.20 to 0.46	< 0.01
POpeak * t1 – t3 <sup>†</sup>	0.06	-0.06 to 0.18	0.32
POpeak * t2 – t3 <sup>†</sup>	0.20	0.08 to 0.32	< 0.01
POpeak * t4 – t3 <sup>†</sup>	-0.03	-0.13 to 0.07	0.55
Complications (1 = present; 0 = absent)	-2.70	-4.93 to -0.47	0.02

\*Regression coefficient indicates the increase in FIMmotor score associated with an increase in the independent variable of 1 unit. <sup>†</sup>Interaction terms are linked by asterisks (\*).

significant interaction term (between peak power output and dummy variable 't2 minus t3') indicated that the association with peak power output changed over time: between t2 and t3, the higher the peak power output, the smaller the change in FIMmotor. The final multivariate multilevel model for the probability of independence showed that the likelihood of independence increased with an increase in peak power output, that

#### Chapter 7

Association with the probabil	ity of indepe	endence	
Independent variable	Odds ratio <sup>*</sup>	95% Confidence Interval	p-value
Discharge; t3	0.23	0.05 to 1.15	0.07
Start compared to discharge; $t1 - t3 (0/1)$	0.05	0.01 to 0.21	< 0.01
3 months compared to discharge; $t2 - t3 (0/1)$	0.30	0.11 to 0.81	0.02
1 yr after discharge compared to discharge; $t4 - t3 (0/1)$	2.92	1.21 to 7.04	0.02
Peak power output (Watt)	1.06	1.04 to 1.08	< 0.01
Complications (0= present; 1= absent)	2.89	1.24 to 6.70	0.01
Completeness (0= complete; 1= incomplete)	8.85	3.13 to 25.00	< 0.01

**Table 4** Probability of independence in association with physical capacity and complications:

 multivariate multilevel random coefficient analyses

 $^{\circ}$ Oddsratio indicates the increase in the odds of independence associated with an increase in the variable of 1 unit.

the likelihood of independence was nearly 9 times higher in those with an incomplete lesion, and nearly 3 times higher in the absence of complications (Table 4).

# *Physical capacity and complications as predictors of the probability of independence and health-related functional status*

The predictive model for the probability of independence 1 year after discharge showed that, in addition to independence at discharge, peak power output and completeness of the lesion at discharge (t3) were significant predictors (Table 5 and Figure 3). The predictive models for health-related functional status 1 year after discharge showed that discharge peak power output significantly contributed to the prediction of both dimensions. Additionally, complications reported at discharge, completeness of the lesion at discharge and gender contributed to the prediction of the physical dimension (Table 6).

**Table 5** Predictive model for the probability of physical independence 1 yr after discharge: multiple logistic regression analyses (n= 106)

Probabi	lity of independ	ence 1 year after	discharge	
Independent variable at discharge	Oddsratio <sup>®</sup>	Confidence Interval	p-value <sup>†</sup>	Correct classification <sup>‡</sup>
Constant	0.08	n.a.	n.a.	52%
Independence (0=dependent; 1=independent)	11	3 to 34	<0.01	77%
Peak power output (Watts)	1.04	1.02 to 1.07	< 0.01	79%
Completeness of the lesion (0=complete; 1=incomplete)	14	4 to 52	0.02	81%

\*Oddsratio indicates the increase in the odds of independence associated with an increase in the independent variable of 1 unit. †p-value for the association between the odds of independence and this independent variable. ‡percentage of study sample correctly classified as independent or as dependent by this predictive model. Abbreviation: n.a. not applicable.



**Figure 3** Predicted probability of independence 1 year after discharge in association with discharge peak power output, independence at discharge and completeness of the lesion at discharge: estimated with multiple logistic regression analysis

Table 6 I	Predictive	models for	the physical	and social	dimension	of the	Sickness	Impact I	Profile
(SIP68) 1	year after	discharge:	multiple line	ear regressi	on analyses	(n = 1	102)		

	Physical dimens	ion		Social dimensio	п	
Independent variable at discharge	Regression coefficient (SE) <sup>*</sup>	Variance <sup>†</sup>	p-value	Regression coefficient (SE)	Variance	p-value
Constant	11.7 (1.3)	n.a.	n.a.	8.8 (0.8)	n.a.	n.a.
Peak power output (watts)	-0.13 (0.02)	18%	< 0.01	-0.08 (0.02)	16%	< 0.01
Complication <sup>‡</sup> (0/1)	3.0 (1.0)	8%	< 0.01	n.s.	n.a.	n.a.
Completeness (0/1)	2.3 (1.0)	3%	0.02	n.s.	n.a.	n.a.
Gender (0=women; 1 =men)	2.4 (1.2)	2%	0.04	n.s.	n.a.	n.a.
Accuracy¶	$RSD = 5.0 R^2 = .3$	31		$RSD = 3.9 R^2 =$	.16	

<sup>\*</sup>Regression coefficient indicates the difference in SIP68 score associated with an increase in the independent variable of 1 unit. <sup>†</sup>Percentage of variance explained by this independent variable. <sup>¶</sup>Adjusted R<sup>2</sup> and residual standard deviation (RSD) indicate the accuracy of the model. Abbreviation n.a. not applicable; n.s. non significant. Complications: 0=absent; 1=complications reported at discharge. Completeness: 0=incomplete; 1=complete.

### Discussion

#### Changes in physical independence over time

This prospective study of subjects with SCI showed that physical independence improved during active rehabilitation, but that the FIMmotor stabilized after discharge from inpatient rehabilitation. This unaltered FIMmotor is in agreement with other studies and may be explained by several factors.<sup>13,195,203</sup> First, in the Netherlands patients are usually not discharged from rehabilitation until certain skills have been acquired.<sup>3,13</sup> Therefore, it is understandable that after discharge no recovery could be measured in these specific activities. Second, outside the rehabilitation centre, patients may face an environment not adapted to their individual needs; they have acquired skills, but their surroundings do not allow them to put these skills into practice. At outpatient follow-up, clinicians need to address whether activities are limited by environmental factors, like home adaptations, transport facilities and recreational opportunities. Third, the FIMmotor only determines physical independence during basic activities of daily living.<sup>195,204</sup> For some patients, higher levels of activities and participation may need to be addressed at follow-up visits and in prospective studies.

The seemingly contrasting findings that after discharge the FIMmotor stabilized, whereas the probability of independence increased, called for an examination of our data. The distribution of FIMmotor at discharge was used to divide the study sample into deciles. Subsequently, comparisons between changes in FIMmotor after discharge showed that subjects in the middle deciles improved. Because their FIMmotor at discharge approximated 78 points, they often progressed from being categorized as dependent to being categorized as independent. Those in the higher deciles showed few changes, which suggests that FIMmotor may have plateaued in these subjects.<sup>195</sup> Finally, the lower deciles showed an actual decline in FIMmotor, which explains why, overall, the FIMmotor did not improve significantly. The decline in functioning after discharge in those subjects who were already categorized as dependent may partly be ascribed to the finding that a patient's ability to do something does not necessarily coincide with everyday performance or behaviour.<sup>15,52</sup> For example, if a patient can carry out one activity, but gets help for another activity, it is conceivable that he is also assisted with the former. Similarly, if a patient can ride a manual wheelchair indoors but requires an electric wheelchair for outdoor activities, he could choose to use the electric wheelchair indoors too. Furthermore, although a patient may be able to carry out his own self-care, he may choose to be assisted in order to be able to spend his energy on recreational activities or work. These choices may negatively affect the measured level of physical independence. However, being able to make these choices contributes to a sense of control and wellbeing.37,53 Therefore, we recommend rehabilitation professionals to strive for the optimal development of skills, because this allows patients to make a choice. Furthermore, clinicians need to regularly regard the patient's individual needs in the evaluation of the necessity of continuing or starting up post-discharge care.

# *Changes in physical independence over time in association with physical capacity and complications*

Conform the cross-sectional associations, changes in physical capacity were associated with changes in physical independence.<sup>33,133</sup> Because we included wheelchair users, the endurance capacity determined on a wheelchair treadmill was expected to be associated with the physical independence.<sup>33</sup> The peak power output was more strongly associated with independence than were peak oxygen uptake and muscle strength, probably because it is a more comprehensive measure of body function, more closely related to activities and influenced by the technique of wheelchair propulsion.<sup>25,133</sup> Furthermore, peak power output showed stronger associations with physical independence than did subject and lesion characteristics. We hypothesize that subject and lesion characteristics are embodied in the peak power output, and that the influence of skills and technique make the power output especially associated with activities in a wheelchair-bound population. Because of its consistent association with physical independence, we recommend the regular assessment of peak power output as an objective measure of rehabilitation outcome in wheelchair-bound patients with SCI.

The negative association between complications and physical independence found in our prospective study coincides with cross-sectional data reported by others.<sup>20,34</sup> Although bed rest may result in deconditioning, it was not associated with an expected functional decline in our study sample.<sup>163</sup> Therefore, bed rest may have caused subjects to temporarily carry out fewer activities, but did not affect physical independence. Future investigation of the effect of the duration of bed rest or of the severity of complications may provide meaningful information. Spasticity and pain have different consequences in different situations and this is probably why they were not associated with physical independence in our multivariate models.<sup>34,44</sup> Whilst spasticity may limit arm function, it may become useful in the legs when making a transfer.<sup>34</sup> Similarly, patients who are active may be exposed to overuse and, therefore, susceptible to pain. Conversely, pain may directly limit activities in others.<sup>178</sup> Although clinicians, patients and relatives are aware of the continuous threat of complications, they need to address concomitant changes in physical independence and, conversely, need to consider the presence of complications if sudden changes in physical independence occur.

# *Physical capacity and complications as predictors of the probability of independence and the SIP68*

Multilevel random coefficient models gave insight into a combination of longitudinal and cross-sectional associations,<sup>138</sup> but to establish causality we needed to extrapolate longitudinal associations. Although our results showed the significant predictive value of peak power output and complications, these variables made a modest contribution to the likelihood of independence at follow-up as compared to independence at discharge.<sup>157</sup> Therefore, intervention studies are needed to establish whether training physical capacity, and the prevention or treatment of complications is actually beneficial. Peak power output also contributed to the prediction of the physical and social dimensions of health-related functional status. However, a large proportion of variance remained unexplained, which may be due to several reasons.<sup>33</sup> First, discharge health-related functional status was not determined at discharge and, therefore, did not contribute to the predictive model. Second, we explained less variance than did Dallmeijer et al., probably because our study sample was larger and less heterogeneous than theirs.<sup>33</sup> Finally, after discharge other factors, such as education, relatives and social support, may also be associated with health-related functional status.<sup>20,33</sup>

#### Limitations

Our inclusion and exclusion criteria need to be considered when interpreting results. We included those subjects whose level of physical capacity could be established, which hampers the generalizability to all patients with a tetraplegia or a complete lesion. However, it could explain why physical independence at discharge was higher than that reported by others, 195,203 and why the association with the level of lesion was not as strong as expected.<sup>13,20,203</sup> The outcome measures may need some consideration. This study focussed on physical independence, and although SCI can also influence emotional and psychological function, we did not investigate psychological, communicative and emotional items of the FIM or SIP68, which are considered to be stable and less informative following SCI.195,198 The chosen threshold for the dichotomous rating of physical independence means that those categorized as independent were on average independent for the FIMmotor items. If, for example, one defines physical independence as independent on all separate FIMmotor items, only 22% of the sample would be considered independent 1 year after discharge.<sup>205</sup> Because subjects may require assistance for other, not specified, activities, the choice of threshold remains arbitrary. One has to consider that our finding that 55% of the study sample depends on others 1 year after discharge may be an underestimation of their level of dependency. This study did not determine changes in health-related functional status. However, health-related functional status is probably especially informative
after discharge, when a higher level of activities is required, and patients face different restrictions in participation. Therefore, we recommend that health-related functional status and its determinants are investigated in future follow-up studies of patients with SCI. Finally, one needs to consider the accuracy of the models. Some variance in outcome remained unexplained, the observed outcome may differ considerably from the predicted outcome and, although the associations with peak power output and complications were consistent, intervention studies are needed to establish complete causality.<sup>157</sup>

In conclusion, we recommend physical independence and health-related functional status to be determined both during rehabilitation and after discharge. Furthermore, clinicians may need to evaluate possible discrepancies between the measured level of activity and participation, the expected level of activity, and actual behaviour. In anticipation of their association with physical independence and health-related functional status, peak power output and complications also need to be determined regularly. Although the contribution of physical capacity and complications in the predictive models is modest, the results indeed suggest that not only functional training, but also training physical capacity, and the prevention or treatment of complications contribute to physical independence and health-related functional status, but future intervention studies are needed to confirm these proposed beneficial effects.



Discussion

### Main findings

This thesis describes physical capacity and medical complications in wheelchair-bound patients with SCI both during inpatient rehabilitation and during the year after discharge. The critical review in Chapter 2 discusses reasons behind the differences in the reported level of physical capacity. Prospective data of the Umbrella project allowed us to illustrate changes in physical capacity, in the occurrence of complications, and in physical independence in Chapters 3, 5 and 7, respectively. Furthermore, the data gave insight into interactions between aspects of functioning, and between functioning and contextual aspects. We showed associations among components of physical capacity in Chapter 3. Then we illustrated the influence of contextual factors and complications on physical capacity (Chapters 4 and 6), and determined risk factors of complications (Chapter 5). Finally, we showed the associations between physical independence and functional status on the one hand, and physical capacity and medical complications on the other (Chapter 7).

Following SCI, the level of physical capacity and physical independence is generally low. Most patients have complications, report pain and show spasticity at examination both during and after discharge from inpatient rehabilitation. This is especially worrying, because complications may necessitate readmission to hospital, and are negatively associated with physical independence and well-being.<sup>14,20,30</sup>

Fortunately, the overall level of functioning (in this case both the physical capacity and physical independence) improves during inpatient rehabilitation (Figure 1). Recovery of partially innervated muscles and the recovery from concurrent pathology contribute to this improvement,148,206 and rehabilitation programs have additional beneficial effects. 60,200 During rehabilitation patients are taught and trained to optimize their physical capacity and to learn new ways to perform daily activities. In agreement with others, we find muscle strength to be positively associated with peak endurance capacity, and endurance capacity to be positively associated with physical independence.<sup>54,69,144,200</sup> Therefore, we hypothesize that training muscle strength contributes to the performance of newly learnt activities and to physical independence, either directly, or through its association with endurance capacity.<sup>35,207</sup> Conversely, during the recurrent performance of activities, physical capacity may be enhanced.<sup>69</sup> Similarly, once patients acquire wheelchair skills, their physical capacity improves not only by training, but also by daily use of the wheelchair and by a more efficient wheelchair propulsion.35,79,140 Although these proposed mechanisms are beneficial, it is important to evaluate progress after discharge.

After discharge, some aspects of this generally low physical functioning improve, whereas others remain unchanged (Figure 1). The estimated improvements are



**Figure 1** Changes in physical capacity and functioning over time as determined with multilevel regression analyses. Outcomes were standardized as follows: peak power output as a proportion of 47W, peak oxygen uptake as a proportion of 1.40 L/min, and HHD (handheld dynamometry sumscore) as a proportion of 2000N. The FEV1 (forced expiratory flow per second) is a proportion of that expected in a matched able-bodied population. The MMT (manual muscle test sumscore) and FIMmotor (motor Functional Independence Measure) are expressed as a proportion of their maximum score. For complications, the proportion of the sample affected during the previous period is indicated.

comparatively small, and may be clinically less relevant as compared to changes found during inpatient rehabilitation.<sup>183,208</sup> This seems logical because improvements will be more pronounced at the start of active rehabilitation when there is much to be learnt and gained.<sup>49,209</sup> However, because the recovery after discharge is relatively small, the resultant physical capacity is still low and most patients rely on the assistance of others. Perhaps the environment outside the rehabilitation centre does not encourage an active lifestyle, which may eventually lead to a low level of physical capacity.36,38,39,46,57,210,211 An investigation of 40 patients with SCI shows an initial decline in their level of activity after discharge. This decline has largely recovered one year after discharge, but the activity level one year after discharge does not exceed the discharge level of activity and, therefore, remains distinctly low.<sup>212</sup> Obviously, patients are entitled to choose a physically less active life style or to get assistance from others for their daily activities, but they need to be aware that the cumulative effect of a low level of activity could eventually lead to a decline in physical capacity, and could expose them to the risk of complications. Although physical functioning has not actually declined during the year after discharge, patients and clinicians need to anticipate a potential decline during the following years.

The low level of physical capacity in patients with SCI is largely explained by the limited active muscle mass and cardiovascular changes.<sup>88,90,112</sup> In addition to these neurological consequences, factors such as education, hours of employment,



**Figure 2** Positive interactions between physical capacity on the one hand, and complications and contextual factors on the other as established in this thesis; non-significant determinants are printed grey. Adapted from International Classification of Disability, Functioning and Health (ICF).<sup>15</sup>

body mass index, stabilisation of the spine, a brief period of active rehabilitation and the absence of complications are all positively associated with physical capacity (Figure 2). Although the effects of some lifestyle characteristics remain unequivocal, the associations found identify patients at risk of a low physical capacity, which helps target training programs and evaluate progress.<sup>12,36,45,213</sup>

Lesion characteristics are predominant risk factors of complications following SCI. Furthermore, ageing, smoking and an increase in body mass index were associated with an increased risk (Figure 3). Because the occurrence of complications after discharge is additionally associated with physical capacity, we hypothesize that a low physical capacity predisposes to complications.<sup>214,215</sup> It is likely that the adverse effects of the neurological impairments cannot simply be prevented, which means that a low physical capacity, complications and physical dependency remain a constant threat.<sup>172,178</sup> We hypothesize that functional training, enhancement of physical capacity and the consistent prevention and treatment of complications will all contribute to an improvement in physical functioning following SCI.

#### Methodological considerations

#### Considerations of a prospective multi-centre research program

Methodological strengths and weaknesses of this multi-centre research program need to be considered. The contribution of eight Dutch rehabilitation centres provides information on a relatively large and representative sample of patients with SCI, but



**Figure 3** The interactions between complications and risk factors as established in the current thesis \*Gender: for some complications, men were at risk, for others women were at risk; tested but non-significant risk factors are printed grey (adapted from ICF model).<sup>15</sup>

one needs to consider whether the results can be generalised. Obviously, we could not include those who died soon after injury or who were discharged home at the end of their hospital stay. But because the program determined the recovery of mobility in wheelchair-bound patients, community walkers and those with progressive disease were also excluded, which makes our data less characteristic for all patients with SCI. Furthermore, the care for patients with SCI differs between countries which will influence, for example, the populations admitted for rehabilitation, the duration or contents of rehabilitation, and the follow-up care.<sup>13</sup> These differences have to be considered when one extrapolates our findings to rehabilitation settings elsewhere. A core dataset for patients with SCI has recently been developed and evaluated. The collection of these data in future investigations would facilitate the comparison of results between different studies.<sup>216</sup>

Another consideration of a multi-centre research program is that differences between observers or instruments may affect the reliability of the data, but we minimized these influences. The observers were rehabilitation professionals who were trained prior to and during the program and who followed a strict protocol. Furthermore, before use, all instruments were calibrated. The calculation of the peak power output was standardized.<sup>217</sup> The data were noted in identical measurement books, and finally collected in one database. In the multilevel random coefficient analyses, we additionally corrected for the dependency of more assessments by one observer at each rehabilitation centre on theoretical grounds.<sup>138</sup>

The prospective data allow us to evaluate interactions between changes in functioning on the one hand and changes in contextual factors on the other, but the timing of the assessments does need to be considered.<sup>138</sup> The assessments were marked by clinical interference (the start of active rehabilitation or discharge) or by actual moments in time (three months into active rehabilitation or one year after discharge). Therefore, the periods between assessments differed in duration, and during these periods different interventions may have been given. These differences need consideration when interpreting changes in physical functioning, and especially warrant caution when interpreting the reported occurrence of complications. However, the timing of the assessments was the same for all subjects, which facilitates the interpretation of the investigated interactions within subjects. Furthermore, it is important to evaluate a patient's progress, and because we assumed this progress to be determined by rehabilitation interventions rather than time alone, we believe it clinically more relevant to report on changes during phases of rehabilitation than to report, for example, on changes during six calendar months.

Another consideration of the prospective design is that those deceased, those discharged within three months of rehabilitation, those less motivated, and those who suffered from complications may not have completed one or more assessments.<sup>138</sup> Additionally, technical difficulties could have prevented data collection.<sup>217</sup> In unpublished data, no differences were found between subjects who performed the assessment one year after discharge and those who did not with regards to lesion characteristics, physical capacity or physical independence. Because both well-performing subjects and poorly-performing subjects missed assessments the net effect on outcome could have been positive, negative, or none at all.

#### Considerations of the outcome measures

We chose to focus on physical capacity and complications, because we hypothesized that these form the foundation of other aspects of functioning. Furthermore, because spinal cord rehabilitation puts effort into the improvement of physical capacity and the prevention or treatment of complications, we found it meaningful to investigate their progress. Overall, motivation and mood probably influenced the performance of tests and the completion of the questionnaires, but the accuracy and validity of some outcome measures need to be considered further.

Both the endurance capacity (as determined with a wheelchair on a treadmill) and respiratory function (as determined with spirometry) are valid and reliable measures of physical capacity in wheelchair-bound patients with SCI.<sup>17 124</sup> Similarly, the Functional Independence Measure (FIM) and the Sickness Impact Profile (SIP) have proven to be valid to determine aspects of activity and participation following SCI.<sup>52,198,218</sup> However, handheld dynamometry, although more informative than a manual muscle testing grade, provides information on muscle strength in only those muscles with moderate strength and its validity depends on the relative strength of the examiner.<sup>119,202</sup> Furthermore, although adding up the strength of 10 muscle groups provides a comprehensive measure for arm muscle strength, it does not con-

tribute to the validity of the results, and does not provide information on individual muscle groups.  $^{41}\,$ 

The registration of complications also needs to be considered, because in this study no objective guidelines were given for documenting the presence of different complications. Therefore, the clinical expertise of the physician may have influenced the data.

Although the FIM is valid and well established in rehabilitation practice, one could argue for more SCI-specific alternatives. However, the Quadriplegia Index of Functioning (QIF) is limited to those with a tetraplegia.<sup>219</sup> Furthermore, although the newly developed Spinal Cord Independence Measure (SCIM) accommodates SCI-related problems with items on wheelchair use, it had not been validated at the start of the research program.<sup>220,221</sup>

Information on many other aspects of physical functioning or complications (e.g. bladder and bowel control, sexual function, daily level of activity, medication use or glucose intolerance) may have been meaningful.<sup>3,222</sup> Furthermore, the Impact on Participation and Autonomy (IPA), which determines the perception of participation, and the Craig Handicap Assessment and Reporting Technique (CHART), which determines community reintegration, could have been informative in this population especially after discharge.<sup>223,224</sup> However, choices had to be made, and because the completion of all the assessments already took up nearly one complete day, we did not want to burden patients with additional tests.

### Statistical considerations

The prospective collection of data implies that outcome in a subject at one occasion is associated with his or her outcome at a later occasion.<sup>17,52</sup> Multilevel random coefficient analysis incorporates the association between outcome at successive occasions in the analyses of incomplete data sets. Because not all of our subjects completed four assessments, we chose to use this analysis to investigate changes over time.<sup>138</sup> Although this allowed us to make optimal use of the data, we need to emphasize that even subjects who completed only one assessment were included, and that this single assessment contributed to the estimated outcome in the sample.<sup>41</sup> An alternative would have been to include only those subjects for whom outcome and determinants were collected at all occasions. This strategy was chosen in Chapter 4 when we made prognostic models for physical capacity with multiple regression analyses. However, this has the limitation that these models are representative for an even smaller selection of patients with SCI.

The associations with determinants and risk factors need to be considered. The interactions found in the random coefficient models were based on combinations of longitudinal and cross-sectional comparisons, but conclusions on cause and effect

must be interpreted with caution.<sup>138,157</sup> The prognostic models give more insight into causality, because they investigate the influence of contextual factors at baseline on physical capacity at follow-up.<sup>154</sup> However, when we validated the models in our sample, their accuracy was limited and cross-validation with another population would probably have given even less promising results.<sup>154</sup> This inaccuracy means that the models may be complementary to clinical expertise, but that they cannot be used as a solitary decision tool. The arrows in Figures 2 and 3 indicate significant interactions and how we interpreted them, but we cannot rule out whether arrows need to be drawn in the opposite direction, whether associations are bidirectional, or whether associations are partly attributable to a third characteristic.

We investigated a selection of potential determinants and risk factors, and it is possible that other characteristics (e.g. living condition or social support) also influence physical functioning following SCI. However, in a sample of just over 200 subjects it would not have been statistically acceptable to investigate many characteristics simultaneously in one model, and choices had to be made. Therefore, we investigated the association with only a selection of those determinants suggested to be of influence in previous studies, those proven important in the able-bodied population, or those potentially modifiable.

### **Clinical implications**

This thesis shows that the generally low level of physical capacity is associated with several contextual factors. However, before we can use this knowledge to improve physical functioning, it is imperative to translate the estimated differences in physical capacity into clinical messages. What does it mean for a patient if his or her peak oxygen uptake is estimated to be 1.20 L/min at discharge from inpatient rehabilitation? Or, what does a difference in peak power output of 8 W between smokers and non-smokers suggest? An investigation in elderly able-bodied subjects, for example, showed that a peak oxygen uptake of >1.33 L/min was associated with a substantial increase in the likelihood of physical independence.<sup>183</sup> Another study showed that a power output of at least 30 W is required to ride a manual wheelchair at 4 km/h across high pile carpet.<sup>208</sup> If we assume that the mean level of peak power output is 30 W at the start of active rehabilitation, being a smoker or not can then actually make functional differences. For our sample, Chapter 7 shows the association between physical capacity and independence: the likelihood of becoming independent is nearly 20% higher if a patient with an incomplete lesion has a peak power output of 60 W, than if this same patient has a peak power output of 40 W. Because of the limited accuracy of the prognostic models and because interactions may work in both directions, they have to be interpreted with care. Furthermore, although many may acknowledge that taking certain precautions may improve physical functioning and reduce the risk of complications in future, it is difficult to change behaviour accordingly.

At admission to rehabilitation, patients with SCI and their relatives often ask about the prognoses.<sup>121</sup> Will they be dependent on others, will their house need to be changed, or will they be able to resume employment? This thesis does not answer these understandable individual questions. However, it does provide insight into the expected recovery of physical capacity and complications in this population. The expected recovery, together with insight into the influence of contextual factors and risk factors, enables clinicians to more adequately inform patients and relatives about functional prognoses and to evaluate progress. A deviation from the expected pattern or an unexpected decline in functioning could indicate the presence of complications and warrant further investigation. However, one does need to consider that a downward deviation from the expected curve of recovery may be within a normal range, and should not lead to unnecessary disappointment. Furthermore, a promising rate of recovery does not always guarantee that all is well, and does not, for example, justify an unhealthy lifestyle or the omission of follow-up visits.

The only way to ascertain whether a patient performs as expected is by regularly monitoring progress. Structural monitoring provides insight into individual progress, but could also inform about the effectiveness of rehabilitation programs.<sup>194</sup> Until now patient monitoring is not an integrated part of rehabilitation practice in The Netherlands. However, in answer to the first results of the rehabilitation research program, a patient-monitoring project has been developed. This project aimed to implement the repeated and structural assessment and evaluation of several aspects of functioning during spinal cord rehabilitation.<sup>225</sup> The first evaluation of this project among rehabilitation professionals showed that 94% of the respondents indicated that monitoring would improve the quality of rehabilitation practice, and 87% indicated patient monitoring needed to be a consistent component of rehabilitation practice.<sup>225</sup> Therefore, in order to accurately formulate rehabilitation goals and to plan follow-up care, we suggest that physical functioning be monitored at the start of active rehabilitation, during rehabilitation and at discharge. Both patient monitoring and the registration of the contents and intensity of training programs are needed to establish the effectiveness of rehabilitation and in order to practice evidence-based rehabilitation.

Although we found no significant decline in functioning one year after discharge, the overall level of physical capacity is still low, complications remain a threat, and most patients depend on the assistance of others. Outside the rehabilitation centre patients may find the environment not adapted to their needs, which could eventually influence functioning. Furthermore, the negative associations found with age, probably coincide with the hypothesis that the cumulative effects of ageing, an inactive lifestyle, and extensive use of the arms and shoulders become more apparent longer than one year after discharge.<sup>8</sup> This may eventually lead to a noticeable decline in functioning and well-being, and an increase in morbidity, hospital or rehabilitation admissions, and costs attributable to health care or loss of productivity.<sup>7</sup> Rehabilitation professionals need to be aware of these cumulative effects over time and, in anticipation of a potential decline, we recommend them to continue monitoring their patients' functioning and complications at yearly follow-up visits.

It may be unpractical to repeatedly monitor all components of physical capacity, and because of the interactions among these components as reported in Chapter 3, it may also be unnecessary to monitor all.<sup>52,129</sup> However, we suggest that the following components of physical capacity be determined during rehabilitation and at yearly intervals after discharge. Because of its association with cardiovascular disease, we suggest that the endurance capacity be determined.<sup>214,215</sup> Because of its association with mechanical efficiency, wheelchair skills and physical independence, the assessment of endurance capacity should include an assessment of the peak power output.<sup>25</sup> Because of its association with activities of daily living muscle strength needs to be determined,<sup>203</sup> and we suggest the use of a handheld dynamometer.<sup>202</sup> If muscle strength is insufficient for handheld dynamometry, manual muscle testing can be a practical alternative.<sup>21</sup> Because of its possible association with respiratory complications, we suggest that respiratory function be assessed.<sup>22,124</sup> Furthermore, because respiratory function can be determined in most patients, it forms an alternative for patients in whom the endurance capacity or muscle strength cannot be assessed.

Besides determining impairments in body function, it is important to evaluate limitations in activities and restrictions in participation.<sup>16,37</sup> In rehabilitation programs, the FIMmotor is the most widely recognized and uniform measure of physical independence, and we suggest that it be used when monitoring patients during rehabilitation. After discharge, a higher level of functioning needs to be addressed, and we suggest the use of the SIP68 at outpatient follow-up.<sup>198</sup> In addition to addressing performance at follow-up visits, it may be important to concomitantly address a patient's needs, and whether the provided care or assistive devices correspond with individual requirements.

Complications are a persistent problem, and may be of increasing importance in an ageing population.<sup>8</sup> Because of its association with physical capacity and independence, sudden changes in performance should make patients and relatives aware of the possible presence of complications. Patients need to learn how to recognize these changes, to learn how to take preventive measures, and to recognize the necessity to consult professionals in good time. In addition to this self-care, we recommend that screening for complications takes place at regular intervals after discharge from rehabilitation.

#### Suggestions for future research

This thesis illustrates changes in physical capacity and in the occurrence of complications following SCI, and how these changes interact with physical independence. Physical capacity and complications are also important aspects of health in patients with other conditions (e.g. stroke). Therefore, we propose that the work in this thesis, as based on the ICF model, may function as an example for future rehabilitation research among other patient populations.

Future research needs to explore other methods to determine physical capacity in all patients with SCI. For example, it is especially problematical to establish endurance capacity in patients with a cervical lesion or in those who only use an electrical wheelchair. Therefore, we may not be able to draw conclusions on changes in the endurance capacity for all persons with SCI population, and researchers need to develop and evaluate other methods to determine endurance capacity in these vulnerable patients. Perhaps the anaerobe capacity, the sprint power or the results of a sub-maximal exercise test will become suitable alternatives.<sup>144</sup>

The interactions with several determinants have been established, but some associations are worth investigating in future. For example, Chapter 6 gives insight into the association with duration of rehabilitation, but until now little is known about the effectiveness of rehabilitation following SCI. Does rehabilitation optimally improve functioning? Can rehabilitation programs be targeted more effectively? How do rehabilitation programs become more efficient? Answers to these questions could advance rehabilitation practice, but can only be given if individual progress is monitored and compared to the carefully registered rehabilitation input.<sup>226</sup> Furthermore, some modifiable determinants are proposed, but controlled trials need to determine whether stabilisation of the spine, the treatment of complications, a reduction in bed rest or the cessation of smoking, for example, actually improve physical capacity, reduce the risk of complications and consequently lead to an improved level of physical independence or functional status. Finally, before we can determine whether an activity program to promote a healthy lifestyle will enhance physical capacity, the level of activity needs to be objectively quantified in this population.<sup>227,228</sup> The future use of objective measures to quantify the level of activity are recommended.<sup>229</sup> The effect of leading a relatively inactive lifestyle may only become noticeable later postinjury. Therefore, future follow-up studies need to continue longer after discharge from inpatient rehabilitation.46

### Summary

Neurological deficit following a spinal cord injury (SCI) results in a low physical capacity and a high risk of medical complications. The low physical capacity makes activities of daily living (ADLs) more strenuous, and could make patients dependent on the assistance of others for seemingly straightforward activities. Most patients with SCI are wheelchair-bound, which interferes with accessibility. This, together with the constant threat of complications, the strain of ADLs and the physical dependence, might discourage participation in work or social activities. We hypothesized that the following sequence of events occurs: a low physical capacity and complications make activities strenuous, this reduces activity and social participation, the resultant inactive lifestyle further reduces physical capacity, and exposes the patient to more complications, which in turn reduce activity, and so on.

By analyzing changes in physical capacity and the occurrence of complications, their respective determinants and risk factors, and their interactions with physical independence and functional status, this thesis aimed to provide knowledge to help prevent this proposed sequence of events, and to help improve rehabilitation for SCI. The thesis was based on prospective data collected in the Umbrella project, which is part of the eight rehabilitation research programs of The Netherlands Organization of Research and Development (Zon-Mw): 'Physical strain, work capacity and mechanisms of restoration of mobility in the rehabilitation centres, with a spinal cord injury'. Research assistants in eight Dutch rehabilitation centres, with a specialized SCI unit collected data on 225 patients with SCI, who were wheelchairbound at initial rehabilitation.

The introductory *Chapter 1* describes SCI and its possible consequences. Physical capacity and medical complications are positioned in the International Classification of Functioning, Disability and Health (ICF) model of health. This model shows how physical capacity and complications could interact with other aspects of functioning. Chapter 1 provides a description of the Umbrella project and concludes with an outline of this thesis.

Physical capacity is defined as the joint ability of muscles and respiratory and cardiovascular systems to attain a peak level of activity. Both the separate functions of these systems, and their combined functions, inform us about the level of physical capacity. This means that in addition to components of physical capacity, such as muscle strength and respiratory function, the endurance capacity (peak oxygen uptake and peak power output) is investigated.

Summary

In order to address the complexity of defining or quantifying physical capacity, *Chapter 2* provides a literature review, which summarizes the reported level of physical capacity, and critically evaluates how physical capacity was measured following SCI. Based on topic-related criteria and methodological criteria, we selected 52 articles for the review. Overall, the reported level of physical capacity was low, and because those with a tetraplegia and women were underrepresented in the study samples, the actual level of physical capacity may be even less promising following SCI. Variation in results between the publications was caused by differences in the study population and in research protocols, e.g. the chosen mode of exercise or measuring device. Chapter 2 provides suggestions as to which components of physical capacity should be determined, and the appropriate methods to be used. As long as one takes into consideration the effect of differences in populations and protocols, the descriptive data reported in Chapter 2 can be used as reference material in rehabilitation and training.

Low physical capacity following SCI has been described in Chapter 2. However, we assume that during initial rehabilitation strength or endurance training will improve physical capacity. After discharge, when the environment may not be fully adapted to the patient's needs, a decline in activity may negatively influence physical capacity. To gain insight into these influences Chapter 3 used the prospective data to describe changes in physical capacity over time, how these changes interact with age, gender and level and completeness of the lesion, and how changes in components of physical capacity interact with one another. All the investigated components of physical capacity improved significantly during inpatient rehabilitation. The peak oxygen uptake and the peak power output, for example, improved by 24% and 41%, respectively. Not all components continued to improve during the year after discharge, and the overall level of physical capacity remained relatively low. In anticipation of a potential decline in physical capacity in association with an expected level of relative inactivity after discharge, we recommend the structural registration and evaluation of physical capacity at follow-up visits. Women, the elderly and those with a tetraplegia had a lower physical capacity. Over time, gender-related differences and the negative association with age became larger, whereas the negative association with a tetraplegia became smaller after discharge. In order to allow patients to develop their maximal potential of physical capacity, we recommended a further exploration of the reasons behind the differences in recovery between subpopulations. The separate components of physical capacity were in fact related. However, whether training of one component actually improves the other needs to be established in future intervention studies.

At the start of rehabilitation, patients, relatives and clinicians need to be informed about the expected prognosis, which will help to set realistic rehabilitation targets. In Chapter 3 we established changes over time as well as several associations, which allow us to estimate prognoses for different populations. However, these results were based on combinations of cross-sectional and longitudinal associations. Furthermore, they included differences between subpopulations that can be taken into account but, however, cannot be changed. Therefore, the aim of *Chapter 4* was to use the prospective data to make prognostic models, which also included some modifiable potential predictors. Muscle strength and peak power output were relatively well predicted, with explained variances ranging between 42% and 84%. Not surprisingly, most variation in outcome (51% to 82%) was explained by prior physical capacity, but (although their contribution was modest) former education and employment, stabilization of the spine, smoking, and body mass index were significant predictors of physical capacity at follow-up. When we compared the predicted outcome with the observed outcome in the same study sample, the accuracy of the models was limited. Therefore, we suggested that the models be used to estimate prognosis only in combination with clinical expertise and knowledge of the individual patient.

Impairment and disability may not only be caused by the neurological deficit following SCI, but can also arise from complications. A complication is defined as a condition that follows the SCI in time, and for which the risk is increased after SCI. *Chapter 5* describes changes in the occurrence of complications during and after inpatient rehabilitation, as well as risk factors of these complications. Most patients reported pain or showed spasticity at physical examination. There were no significant changes in the occurrence of complications and, therefore, complications remained a threat even during the year after injury, when urinary tract infections and pressure sores, for example, affected 49% and 36% of the study population, respectively. Those with a tetraplegia, a complete lesion or a traumatic lesion were at risk of several complications. Additionally, increases in age or in body mass index were significant risk factors. During rehabilitation and at regular follow-up visits clinicians need to systematically address and investigate the presence of complications.

Initial spinal cord rehabilitation programs include the treatment or prevention of complications, and the enhancement of physical capacity. Changes in physical capacity have been described in Chapter 3, but an exploration of the effect of rehabilitation programs on the recovery of physical capacity was considered additionally meaningful. Unfortunately, we could not determine the effect of the contents and intensity of training programs. However, we assumed that the duration of active rehabilitation, during which the patient participated in functional, strength and endurance training in the wheelchair, would correlate with the given hours of therapy. Therefore, in *Chapter 6* we investigated the complex interactions between the level of physical capacity on the one hand, and the duration of phases of rehabilitation and the occurrence of complications on the other. Results showed that the duration

#### Summary

of active rehabilitation, bed rest and the occurrence of complications after discharge were negatively associated with the level of physical capacity. Furthermore, duration of active rehabilitation and spasticity were negatively associated with the recovery of physical capacity over time. Future studies need to systematically investigate the effect of rehabilitation programs on the level of physical capacity.

Besides insight into physical capacity, the expected level of physical independence or functional status provides important and meaningful information for patients, relatives and clinicians. We hypothesized that patients require a basic level of physical capacity and no hindrance from complications before they can be active and involved. Therefore, Chapter 7 describes changes in physical independence over time. Furthermore, we determined the value of physical capacity and complications in predicting physical independence and functional status one year after discharge. The level of physical independence improved during inpatient rehabilitation and remained unchanged thereafter, whereas the likelihood of independence continued to improve. In addition to discharge dependency status and completeness of the lesion, physical capacity and the absence of complications at discharge were positively associated with physical independence and functional status at follow-up. These interactions suggest that not only functional training, but also the training of physical capacity and the treatment or prevention of complications could contribute to functional prognosis, but future intervention studies need to establish these proposed effects.

Finally, *Chapter 8* discusses the main findings of the studies, how these fit into the ICF model, and whether they correspond with the aforementioned sequence of events following SCI. We also addressed considerations related to the multi-centre prospective study and discussed issues applying specifically to the outcome measures and analyses in this thesis. We aimed to translate the findings of this thesis into meaningful clinical messages, and have provided suggestions for future research.

### Samenvatting

Na een dwarslaesie leiden het verlies van spierkracht en de verandering in cardiovasculaire controle tot een lage fysieke capaciteit en een hoog risico op complicaties. Door de verminderde fysieke capaciteit en het optreden van complicaties worden dagelijkse activiteiten meer belastend. Daarnaast is een groot deel van de patiënten rolstoelgebonden en aangewezen op de hulp van anderen. Door de relatief ontoegankelijke omgeving en de fysieke afhankelijkheid kunnen activiteit en deelname aan arbeid of aan sociale activiteiten verder worden bemoeilijkt. Dit kan op zijn beurt weer verdere achteruitgang van fysieke capaciteit veroorzaken en de kans op complicaties verder vergroten. In navolging op deze constateringen was onze hypothese dat de volgende kettingreactie ontstaat: een lage fysieke capaciteit maakt dagelijkse activiteiten belastend, wat leidt tot een vermindering in activiteit en participatie en een verhoogd risico op complicaties, die vervolgens de fysieke capaciteit verminderen, de activiteit verder belemmeren, en zo verder. Door veranderingen in fysieke capaciteit en in het optreden van complicaties te bestuderen, hun determinanten of risicofactoren te onderzoeken, en vervolgens fysieke capaciteit en complicaties te relateren aan fysieke zelfstandigheid en functionele status, beoogt dit proefschrift te leiden tot kennis die kan bijdragen aan het voorkomen of tegengaan van de omschreven negatieve kettingreactie en de verbetering van het revalidatieproces van mensen met een dwarslaesie.

Het proefschrift is gebaseerd op data die zijn verzameld in het landelijke Koepelproject. In het kader van dit project hebben acht gespecialiseerde revalidatiecentra gegevens verzameld van 225 rolstoelgebonden patiënten tijdens hun initiële dwarslaesie revalidatie. Het Koepelproject maakt onderdeel uit van een van de 8 onderzoeksprogramma's van het eerste Revalidatieonderzoeksprogramma van Zon-Mw: 'Mobiliteitsherstel in de revalidatie van patiënten met een dwarslaesie'.

In het inleidende *Hoofdstuk 1* worden de gevolgen van een dwarslaesie omschreven. Met behulp van het ICF (International Classification of Functioning, Disability and Health) model worden onze onderzoeksvragen betreffende fysieke capaciteit en complicaties in een kader geplaatst. Hoofdstuk 1 eindigt met een omschrijving van de opzet van het Koepelproject, en van de onderzoeksvragen die in het proefschrift aan bod komen.

Fysieke capaciteit is de gezamenlijke capaciteit van neuromusculaire, respiratoire en cardiovasculaire systemen om een bepaalde mate van activiteit of inspanning vol te houden. Het is een complex begrip dat zowel de afzonderlijke functies van de verschillende systemen als hun onderlinge samenwerking behelst. Om een goed Samenvatting

beeld van fysieke capaciteit te krijgen moet men dus zowel de spierkracht als de respiratoire functie als de aërobe inspanningstolerantie (piek zuurstofopname en piek vermogen) evalueren. Veel studies hebben zich gericht op het meten of verbeteren van fysieke capaciteit na een dwarslaesie, maar de gerapporteerde waarden van fysieke capaciteit zijn niet samengevat tot referentiewaarden voor een algemene populatie patiënten met een dwarslaesie. Daarom heeft Hoofdstuk 2 als doel deze studies te evalueren. Na de selectie van publicaties op basis van inhoudelijke en methodologische criteria, zijn de methoden en resultaten van 52 studies samengevat. De gerapporteerde mate van fysieke capaciteit bleek uiteenlopend, maar de fysieke capaciteit was over het algemeen laag in vergelijking met de fysieke capaciteit van mensen zonder een dwarslaesie. We concludeerden dat de gerapporteerde resultaten misschien zelfs een te rooskleuring beeld gaven wegens de ondervertegenwoordiging van vrouwen, patiënten met een tetraplegie en inactieve patiënten. We verklaarden de uiteenlopende resultaten doordat de bestudeerde studies verschilden in onderzoekspopulaties en in meetmethoden om fysieke capaciteit te kwantificeren. Wanneer het effect van deze verschillen in ogenschouw wordt genomen, geeft Hoofdstuk 2 bruikbare referentiewaarden voor de evaluatie van de individuele mate van fysieke capaciteit en het effect van revalidatie en training.

De revalidatie besteedt ruime aandacht aan het verbeteren van de inspanningstolerantie, en daarom namen we aan dat de fysieke capaciteit tijdens de klinische revalidatie verbetert. Omdat de omgeving buiten het revalidatiecentrum misschien onvoldoende is aangepast aan de mogelijkheden van een patiënt, kan een patiënt na ontslag zijn belemmerd in het uitvoeren van dagelijkse activiteiten. We veronderstelden dat deze mogelijke belemmering in activiteit kan leiden tot een afname van de fysieke capaciteit. Hoofdstuk 3 heeft als doel de veranderingen in fysieke capaciteit tijdens en na ontslag van klinische revalidatie te omschrijven, deze veranderingen te relateren aan leeftijd, geslacht, en niveau en compleetheid van de laesie en vervolgens de interacties tussen componenten van fysieke capaciteit onderling (spierkracht, longfunctie en inspanningstolerantie) te bestuderen. Uit de resultaten bleek dat de fysieke capaciteit verbetert tijdens klinische revalidatie. De piek zuurstofopname en het piek vermogen verbeteren respectievelijk 24% en 41%. Sommige componenten van de fysieke capaciteit verbeteren ook na ontslag, maar een jaar na ontslag blijft de totale fysieke capaciteit laag. Verder bleek dat mannen, jongeren en patiënten met een paraplegie een hogere fysieke capaciteit hebben dan vrouwen, ouderen en patiënten met een tetraplegie. Na verloop van tijd wordt de voorsprong van mannen en jongeren groter, maar het verschil tussen patiënten met een paraplegie en patiënten met een tetraplegie wordt dan kleiner. De componenten van fysieke capaciteit bleken onderling gerelateerd te zijn. Geconcludeerd werd dat het raadzaam is om de evaluatie van fysieke capaciteit te integreren in de poliklinische nazorg, omdat de

fysieke capaciteit een jaar na ontslag nog steeds laag is. De achterliggende redenen voor het verschil in herstel tussen de patiëntgroepen dienen te worden onderzocht.

Met informatie over de functionele prognose kunnen clinici en patiënten realistische revalidatiedoelen stellen en kan de noodzaak van nazorg vroegtijdig worden geëvalueerd. Hoofdstuk 4 heeft als doel voorspellende modellen te maken voor de fysieke capaciteit bij ontslag en een jaar na ontslag, waarbij de voorspellende waarde van specifiekere en mogelijk veranderbare factoren wordt geëvalueerd. Resultaten lieten zien dat spierkracht en piek vermogen relatief voorspelbaar zijn ( $R^2 = 0.42$  tot 0.84) in vergelijking met respiratoire functie of piek zuurstofopname. De grootste variantie (51% tot 82%) wordt verklaard door de fysieke capaciteit op een eerder meetmoment. De modellen toonden echter ook dat roken, body mass index, de stabilisatie van de wervelkolom, opleiding en arbeid voor de dwarslaesie een bescheiden maar significante bijdrage leveren aan het voorspellen van fysieke capaciteit. Wanneer de voorspelde uitkomst wordt vergeleken met de geobserveerde fysieke capaciteit bleek dat de nauwkeurigheid van de modellen te wensen overlaat. Dus adviseerden we dat de voorspellende modellen alleen een bijdrage kunnen leveren aan het evalueren van herstel of het plannen van zorg, wanneer ze worden gebruikt in aanvulling op professionele ervaring en kennis over de individuele patiënt.

Beperkingen ontstaan niet alleen ten gevolge van het neurologische functieverlies na de dwarslaesie, maar ook ten gevolge van complicaties. Complicaties zijn aandoeningen die optreden na het ontstaan van een andere aandoening; in dit geval na de dwarslaesie. We veronderstellen dat deze complicaties vaker voorkomen bij patiënten met een dwarslaesie dan bij mensen zonder dwarslaesie. *Hoofdstuk 5* heeft als doel het bestuderen van het optreden van complicaties, en het bepalen van hun risicofactoren. Uit de resultaten bleek dat de meerderheid van de patiënten pijn of spasticiteit heeft, zowel tijdens als na klinische revalidatie. Er weinig verandering op in het voorkomen van complicaties. Dit leidt ertoe dat ook gedurende het jaar na de revalidatie (wanneer bijvoorbeeld 49% en 36% van de patiënten respectievelijk een urineweginfectie of decubitus heeft) complicaties een veelvoorkomend probleem zijn. Naast een tetraplegie of een complete laesie, bleken een toename in leeftijd en in body mass index significante risicofactoren te zijn. We raadden dan ook aan dat de systematische preventie of behandeling van complicaties een integraal onderdeel dient te worden van zowel de klinische revalidatie als van de poliklinische nazorg.

Tijdens revalidatie wordt getracht de fysieke capaciteit te vergroten en complicaties te voorkomen of te behandelen. Het is belangrijk om inzicht te krijgen in de effectiviteit van deze revalidatie. Helaas zijn gedetailleerde gegevens over de inhoud, duur en intensiteit van revalidatieprogramma's voor dit onderzoeksproject niet beschikbaar. We veronderstelden echter dat de duur van klinische revalidatie correleert met de duur van therapie, en dat tijdens acute revalidatie (d.w.z. wanneer de patiënt nog niet een aaneengesloten periode kan zitten), mogelijk een andere behandelingsintensiteit wordt gehanteerd dan tijdens de actieve revalidatie, wanneer de patiënt langer dan drie uur in een rolstoel kan zitten. Daarom onderzochten we in *Hoofdstuk 6* de complexe relatie tussen fysieke capaciteit enerzijds, en het optreden van complicaties, de duur van acute revalidatie en de duur van actieve revalidatie anderzijds. Uit de resultaten bleek dat de duur van actieve revalidatie, bedrust en het optreden van complicaties na ontslag negatief geassocieerd zijn met het niveau van fysieke capaciteit. Bovendien zijn de duur van actieve revalidatie en spasticiteit negatief geassocieerd met het herstel van de fysieke capaciteit. We concludeerden dat systematische registratie van revalidatietherapie en het monitoren van het herstel van individuele patiënten noodzakelijk zijn om de effectiviteit van therapie in de toekomst in meer detail te bepalen.

Naast informatie over fysieke capaciteit of complicaties, is inzicht in de verwachte fysieke zelfstandigheid en de functionele status belangrijk. Een doel van de revalidatie is om het niveau van activiteit en participatie zo optimaal mogelijk te ontwikkelen. We veronderstelden dat een bepaald niveau van fysieke capaciteit en de afwezigheid van complicaties voorwaarden zijn voor het uitvoeren van bepaalde activiteiten of voor deelname aan bijvoorbeeld arbeid of andere sociale activiteiten. Daarom heeft Hoofdstuk 7 als doel veranderingen in de mate van fysieke zelfstandigheid gedurende en na klinische revalidatie te omschrijven en deze te relateren aan fysieke capaciteit en het optreden van complicaties, om vervolgens voorspellende modellen te maken voor fysieke zelfstandigheid en functionele status een jaar na ontslag. Resultaten toonden aan dat de mate van zelfstandigheid en de kans op fysieke zelfstandigheid verbeteren tijdens opname, terwijl gedurende het jaar na ontslag alleen de kans op zelfstandigheid toeneemt. Behalve de mate van zelfstandigheid bij ontslag en compleetheid van de laesie, zijn ook piek vermogen en het optreden van complicaties geassocieerd met fysieke zelfstandigheid en functionele status. Resultaten suggereerden dat niet alleen het oefenen van vaardigheden, maar ook het trainen van fysieke capaciteit en het voorkomen en behandelen van complicaties kunnen bijdragen aan het verbeteren van de fysieke zelfstandigheid en de functionele status. Interventiestudies dienen deze mechanismen en hun mogelijke effecten in de toekomst te onderzoeken.

Ten slotte bediscussieert *Hoofdstuk 8* de belangrijkste bevindingen van dit proefschrift. Methodologische kanttekeningen van een prospectief onderzoek binnen meerdere centra, en van de statistische procedures komen aan bod. Uiteindelijk vertaalden we de bevindingen van dit proefschrift in een aantal klinische boodschappen. Ten slotte deden we suggesties voor toekomstig onderzoek.

### References

- Maynard FM, Jr., Bracken MB, Creasey G, Ditunno JF, Jr., Donovan WH, Ducker TB, Garber SL, Marino RJ, Stover SL, Tator CH, Waters RL, Wilberger JE, Young W. International Standards for Neurological and Functional Classification of Spinal Cord Injury. American Spinal Injury Association. *Spinal Cord* 1997; 35: 266-74.
- 2. van Asbeck FW, Post MW, Pangalila RF. An epidemiological description of spinal cord injuries in The Netherlands in 1994. *Spinal Cord* 2000; 38: 420-4.
- 3. van Asbeck FWA. *Handboek dwarslaesierevalidatie.* Second ed. Houten: Bohn Stafleu van Loghum; 2007.
- 4. New PW, Rawicki HB, Bailey MJ. Nontraumatic spinal cord injury: demographic characteristics and complications. *Arch Phys Med Rehabil* 2002; 83: 996-1001.
- Schonherr MC, Groothoff JW, Mulder GA, Eisma WH. Rehabilitation of patients with spinal cord lesions in The Netherlands: an epidemiological study. *Spinal Cord* 1996; 34: 679-83.
- Post MW, van Asbeck FW, van Dijk AJ, Schrijvers AJ. Services for spinal cord injured: availability and satisfaction. *Spinal Cord* 1997; 35: 109-15.
- 7. www.spinalcord.uab.edu.
- 8. Capoor J, Stein AB. Aging with spinal cord injury. *Phys Med Rebabil Clin N Am* 2005; 16: 129-61.
- 9. Noreau L, Shephard RJ. Spinal cord injury, exercise and quality of life. *Sports Med* 1995; 20: 226-50.
- Tate DG, Kalpakjian CZ, Forchheimer MB. Quality of life issues in individuals with spinal cord injury. *Am J Phys Med Rehabil* 2002; 83: S18-25.
- 11. Noreau L, Proulx P, Gagnon L, Drolet M, Laramee MT. Secondary impairments after spinal cord injury: a population-based study. *Am J Phys Med Rebabil* 2000; 79: 526-35.
- 12. Janssen TW, Dallmeijer AJ, Veeger DJ, van der Woude LH. Normative values and determinants of physical capacity in individuals with spinal cord injury. *J Rehabil Res Dev* 2002; 39: 29-39.
- Post MW, Dallmeijer AJ, Angenot EL, van Asbeck FW, van der Woude LH. Duration and functional outcome of spinal cord injury rehabilitation in the Netherlands. *J Rehabil Res Dev* 2005; 42: 75-85.
- 14. Cardenas DD, Hoffman JM, Kirshblum S, McKinley W. Etiology and incidence of rehospitalization after traumatic spinal cord injury: a multicenter analysis. *Arch Phys Med Rehabil* 2004; 85: 1757-63.
- 15. World Health Organization. *International Classification of Functioning, Disability and Health: ICF.* Geneva: World Health Organization; 2001.
- Stucki G, Melvin J. The International Classification of Functioning, Disability and Health: a unifying model for the conceptual description of physical and rehabilitation medicine. *J Rehabil Med* 2007; 39: 286-92.
- 17. Stewart MW, Melton-Rogers SL, Morrison S, Figoni SF. The measurement properties of fitness measures and health status for persons with spinal cord injuries. *Arch Phys Med Rehabil* 2000; 81: 394-400.
- 18. Hooker SP, Greenwood JD, Hatae DT, Husson RP, Matthiesen TL, Waters AR. Oxygen uptake and heart rate relationship in persons with spinal cord injury. *Med Sci Sports Exerc* 1993; 25: 1115-9.
- 19. Meyers AR, Andresen EM, Hagglund KJ. A model of outcomes research: spinal cord injury. *Arch Phys Med Rebabil* 2000; 81: S81-90.
- Post MW, de Witte LP, van Asbeck FW, van Dijk AJ, Schrijvers AJ. Predictors of health status and life satisfaction in spinal cord injury. *Arch Phys Med Rehabil* 1998; 79: 395-401.

- 21. Noreau L, Vachon J. Comparison of three methods to assess muscular strength in individuals with spinal cord injury. *Spinal Cord* 1998; 36: 716-23.
- 22. Winslow C, Rozovsky J. Effect of spinal cord injury on the respiratory system. *Am J Phys Med Rehabil* 2003; 82: 803-14.
- 23. Dela F, Mohr T, Jensen CM, Haahr HL, Secher NH, Biering-Sorensen F, Kjaer M. Cardiovascular control during exercise: insights from spinal cord-injured humans. *Circulation* 2003; 107: 2127-33.
- 24. de Groot PC, van Kuppevelt DH, Pons C, Snoek G, van der Woude LH, Hopman MT. Time course of arterial vascular adaptations to inactivity and paralysis in humans. *Med Sci Sports Exerc* 2003; 35: 1977-85.
- 25. van der Woude LH, Bouten C, Veeger HE, Gwinn T. Aerobic work capacity in elite wheelchair athletes: a cross-sectional analysis. *Am J Phys Med Rebabil* 2002; 81: 261-71.
- 26. Astrand PO, Rodahl K. Textbook of work physiology. New York: McGraw-Hill; 1970.
- 27. van der Woude LH, de Groot G, Hollander AP, van Ingen Schenau GJ, Rozendal RH. Wheelchair ergonomics and physiological testing of prototypes. *Ergonomics* 1986; 29: 1561-1573.
- Rutchik A, Weissman AR, Almenoff PL, Spungen AM, Bauman WA, Grimm DR. Resistive inspiratory muscle training in subjects with chronic cervical spinal cord injury. *Arch Phys Med Rehabil* 1998; 79: 293-7.
- Hoffman MD. Cardiorespiratory fitness and training in quadriplegics and paraplegics. Sports Med 1986; 3: 312-30.
- 30. Putzke JD, Richards JS, Hicken BL, DeVivo MJ. Predictors of life satisfaction: a spinal cord injury cohort study. *Arch Phys Med Rehabil* 2002; 83: 555-61.
- 31. Hjeltnes N, Jansen T. Physical endurance capacity, functional status and medical complications in spinal cord injured subjects with long-standing lesions. *Paraplegia* 1990; 28: 428-32.
- Yekutiel M, Brooks ME, Ohry A, Yarom J, Carel R. The prevalence of hypertension, ischaemic heart disease and diabetes in traumatic spinal cord injured patients and amputees. *Paraplegia* 1989; 27: 58-62.
- 33. Dallmeijer AJ, van der Woude LH. Health related functional status in men with spinal cord injury: relationship with lesion level and endurance capacity. *Spinal Cord* 2001; 39: 577-83.
- 34. Dvorak MF, Fisher CG, Hoekema J, Boyd M, Noonan V, Wing PC, Kwon BK. Factors predicting motor recovery and functional outcome after traumatic central cord syndrome: a long-term follow-up. *Spine* 2005; 30: 2303-11.
- 35. Kilkens OJ, Dallmeijer AJ, Nene AV, Post MW, van der Woude LH. The longitudinal relation between physical capacity and wheelchair skill performance during inpatient rehabilitation of people with spinal cord injury. *Arch Phys Med Rehabil* 2005; 86: 1575-81.
- 36. Muraki S, Tsunawake N, Tahara Y, Hiramatsu S, Yamasaki M. Multivariate analysis of factors influencing physical work capacity in wheelchair-dependent paraplegics with spinal cord injury. *Eur J Appl Physiol* 2000; 81: 28-32.
- Carpenter C, Forwell SJ, Jongbloed LE, Backman CL. Community participation after spinal cord injury. *Arch Phys Med Rehabil* 2007; 88: 427-33.
- 38. Dallmeijer AJ, Hopman MT, van As HH, van der Woude LH. Physical capacity and physical strain in persons with tetraplegia; the role of sport activity. *Spinal Cord* 1996; 34: 729-35.
- 39. Manns PJ, Chad KE. Determining the relation between quality of life, handicap, fitness, and physical activity for persons with spinal cord injury. *Arch Phys Med Rehabil* 1999; 80: 1566-71.
- 40. Janssen TW, van Oers CA, van der Woude LH, Hollander AP. Physical strain in daily life of wheelchair users with spinal cord injuries. *Med Sci Sports Exerc* 1994; 26: 661-70.
- 41. Altman DG. Practical statistics for medical research. London: Chapman and Hall; 1991.
- 42. Aito S. Complications during the acute phase of traumatic spinal cord lesions. *Spinal Cord* 2003; 41: 629-35.
- 43. Middleton JW, Lim K, Taylor L, Soden R, Rutkowski S. Patterns of morbidity and rehospitalisation following spinal cord injury. *Spinal Cord* 2004; 42: 359-67.

- 44. Burchiel KJ, Hsu FP. Pain and spasticity after spinal cord injury: mechanisms and treatment. *Spine* 2001; 26: S146-60.
- 45. Davies DS, McColl MA. Lifestyle risks for three disease outcomes in spinal cord injury. *Clin Rehabil* 2002; 16: 96-108.
- 46. Manns PJ, McCubbin JA, Williams DP. Fitness, inflammation, and the metabolic syndrome in men with paraplegia. *Arch Phys Med Rebabil* 2005; 86: 1176-81.
- 47. Valtonen K, Karlsson AK, Alaranta H, Viikari-Juntura E. Work participation among persons with traumatic spinal cord injury and meningomyelocele1. *J Rehabil Med* 2006; 38: 192-200.
- 48. Curtis KA, McClanahan S, Hall KM, Dillon D, Brown KF. Health, vocational, and functional status in spinal cord injured athletes and nonathletes. *Arch Phys Med Rebabil* 1986; 67: 862-5.
- 49. Dallmeijer AJ, van der Woude LH, Hollander AP, van As HH. Physical performance during rehabilitation in persons with spinal cord injuries. *Med Sci Sports Exerc* 1999; 31: 1330-5.
- Knoops KT, de Groot LC, Kromhout D, Perrin AE, Moreiras-Varela O, Menotti A, van Staveren WA. Mediterranean diet, lifestyle factors, and 10-year mortality in elderly European men and women: the HALE project. *JAMA* 2004; 292: 1433-9.
- Sandvik L, Erikssen G, Thaulow E. Long term effects of smoking on physical fitness and lung function: a longitudinal study of 1393 middle aged Norwegian men for seven years. *BMJ* 1995; 311: 715-8.
- 52. Marino RJ. Domains of outcomes in spinal cord injury for clinical trials to improve neurological function. *J Rehabil Res Dev* 2007; 44: 113-22.
- 53. Whalley Hammell K. Experience of rehabilitation findings following spinal cord injury: a metasynthesis of qualitative findings. *Spinal Cord* 2007; 45: 260-274.
- 54. Beninato M, O'Kane KS, Sullivan PE. Relationship between motor FIM and muscle strength in lower cervical-level spinal cord injuries. *Spinal Cord* 2004; 42: 533-40.
- 55. www.fbw.vu.nl/onderzoek/A4zon/ZONenglish/index.htm.
- 56. www.zonmw.nl
- 57. Dallmeijer AJ, van der Woude LH, Hollander PA, Angenot EL. Physical performance in persons with spinal cord injuries after discharge from rehabilitation. *Med Sci Sports Exerc* 1999; 31: 1111-7.
- Noreau L, Shephard RJ, Simard C, Pare G, Pomerleau P. Relationship of impairment and functional ability to habitual activity and fitness following spinal cord injury. *Int J Rehabil Res* 1993; 16: 265-75.
- 59. Noreau L, Shephard RJ. Return to work after spinal cord injury: the potential contribution of physical fitness. *Paraplegia* 1992; 30: 563-72.
- 60. Dallmeijer AJ, Kilkens OJ, Post MW, de Groot S, Angenot EL, van Asbeck FW, Nene AV, van der Woude LH. Hand-rim wheelchair propulsion capacity during rehabilitation of persons with spinal cord injury. *J Rehabil Res Dev* 2005; 42: 55-64.
- 61. Verhagen AP, de Vet HC, de Bie RA, Kessels AG, Boers M, Bouter LM, Knipschild PG. The Delphi list: a criteria list for quality assessment of randomized clinical trials for conducting systematic reviews developed by Delphi consensus. *J Clin Epidemiol* 1998; 51: 1235-41.
- 62. Gardner MJ, Machin D, Campbell MJ. Use of check lists in assessing the statistical content of medical studies. *Br Med J (Clin Res Ed)* 1986; 292: 810-2.
- 63. May LA, Burnham RS, Steadward RD. Assessment of isokinetic and hand-held dynamometer measures of shoulder rotator strength among individuals with spinal cord injury. *Arch Phys Med Rehabil* 1997; 78: 251-5.
- 64. Burns SP, Breuninger A, Kaplan C, Marin H. Hand-held dynamometry in persons with tetraplegia: comparison of make- versus break-testing techniques. *Am J Phys Med Rehabil* 2005; 84: 22-9.
- 65. Herbison GJ, Isaac Z, Cohen ME, Ditunno JF, Jr. Strength post-spinal cord injury: myometer vs manual muscle test. *Spinal Cord* 1996; 34: 543-8.

- 66. Schwartz S, Cohen ME, Herbison GJ, Shah A. Relationship between two measures of upper extremity strength: manual muscle test compared to hand-held myometry. *Arch Phys Med Rehabil* 1992; 73: 1063-8.
- 67. Fujiwara T, Hara Y, Akaboshi K, Chino N. Relationship between shoulder muscle strength and functional independence measure (FIM) score among C6 tetraplegics. *Spinal Cord* 1999; 37: 58-61.
- 68. Bryden AM, Kilgore KL, Lind BB, Yu DT. Triceps denervation as a predictor of elbow flexion contractures in C5 and C6 tetraplegia. *Arch Phys Med Rehabil* 2004; 85: 1880-5.
- 69. Hjeltnes N, Wallberg-Henriksson H. Improved work capacity but unchanged peak oxygen uptake during primary rehabilitation in tetraplegic patients. *Spinal Cord* 1998; 36: 691-8.
- Marino RJ, Rider-Foster D, Maissel G, Ditunno JF. Superiority of motor level over single neurological level in categorizing tetraplegia. *Paraplegia* 1995; 33: 510-3.
- Bernard PL, Mercier J, Varray A, Prefaut C. Influence of lesion level on the cardioventilatory adaptations in paraplegic wheelchair athletes during muscular exercise. *Spinal Cord* 2000; 38: 16-25.
- 72. Bougenot MP, Tordi N, Betik AC, Martin X, Le Foll D, Parratte BJ, Rouillon, JD. Effects of a wheelchair ergometer training programme on spinal cord-injured persons. *Spinal Cord* 2003: 451-6.
- 73. Cooper RA, Horvath SM, Bedi JF, Drechsler-Parks DM, Williams RE. Maximal exercise response of paraplegic wheelchair road racers. *Paraplegia* 1992; 30: 573-81.
- 74. Coutts KD, McKenzie DC. Ventilatory thresholds during wheelchair exercise in individuals with spinal cord injuries. *Paraplegia* 1995; 33: 419-22.
- 75. Dallmeijer AJ, Zentgraaff ID, Zijp NI, van der Woude LH. Submaximal physical strain and peak performance in handcycling versus handrim wheelchair propulsion. *Spinal Cord* 2004; 42: 91-8.
- 76. Gass EM, Harvey LA, Gass GC. Maximal physiological responses during arm cranking and treadmill wheelchair propulsion in T4-T6 paraplegic men. *Paraplegia* 1995; 33: 267-70.
- Kerk JK, Clifford PS, Snyder AC, Prieto TE, O'Hagan KP, Schot PK, Myklebust JB, Myklebust BM. Effect of an abdominal binder during wheelchair exercise. *Med Sci Sports Exerc* 1995; 27: 913-9.
- 78. Knechtle B, Kopfli W. Treadmill exercise testing with increasing inclination as exercise protocol for wheelchair athletes. *Spinal Cord* 2001; 39: 633-6.
- 79. Martel G, Noreau L, Jobin J. Physiological responses to maximal exercise on arm cranking and wheelchair ergometer with paraplegics. *Paraplegia* 1991; 29: 447-56.
- 80. Rasche W, Janssen TW, Van Oers CA, Hollander AP, van der Woude LH. Responses of subjects with spinal cord injuries to maximal wheelchair exercise: comparison of discontinuous and continuous protocols. *Eur J Appl Physiol Occup Physiol* 1993; 66: 328-31.
- Schmid A, Huonker M, Barturen JM, Stahl F, Schmidt-Trucksass A, Konig D, Grathwohl D, Lehmann M, Keul J. Catecholamines, heart rate, and oxygen uptake during exercise in persons with spinal cord injury. *J Appl Physiol* 1998; 85: 635-41.
- Schmid A, Huonker M, Stober P, Barturen JM, Schmidt-Trucksass A, Durr H, Volpel HJ, Keul J. Physical performance and cardiovascular and metabolic adaptation of elite female wheelchair basketball players in wheelchair ergometry and in competition. *Am J Phys Med Rehabil* 1998; 77: 527-33.
- Veeger HE, Hadj Yahmed M, van der Woude LH, Charpentier P. Peak oxygen uptake and maximal power output of Olympic wheelchair-dependent athletes. *Med Sci Sports Exerc* 1991; 23: 1201-9.
- Vinet A, Bernard PL, Poulain M, Varray A, Le Gallais D, Micallef JP. Validation of an incremental field test for the direct assessment of peak oxygen uptake in wheelchair-dependent athletes. *Spinal Cord* 1996; 34: 288-93.
- Ellenberg M, MacRitchie M, Franklin B, Johnson S, Wrisley D. Aerobic capacity in early paraplegia: implications for rehabilitation. *Paraplegia* 1989; 27: 261-8.

- Hooker SP, Greenwood JD, Boyd LA, Hodges MR, McCune LD, McKenna GE. Influence of posture on arm exercise tolerance and physiologic responses in persons with spinal cord injured paraplegia. *Eur J Appl Physiol* 1993; 67: 563-6.
- Hopman MT, Houtman S, Groothuis JT, Folgering HT. The effect of varied fractional inspired oxygen on arm exercise performance in spinal cord injury and able-bodied persons. *Arch Phys Med Rehabil* 2004; 85: 319-23.
- 88. Hopman MT, Dueck C, Monroe M, Philips WT, Skinner JS. Limits to maximal performance in individuals with spinal cord injury. *Int J Sports Med* 1998; 19: 98-103.
- 89. Jacobs PL, Nash MS, Rusinowski JW. Circuit training provides cardiorespiratory and strength benefits in persons with paraplegia. *Med Sci Sports Exerc* 2001; 33: 711-7.
- 90. Knechtle B, Muller G, Willmann F, Eser P, Knecht H. Comparison of fat oxidation in arm cranking in spinal cord-injured people versus ergometry in cyclists. *Eur J Appl Physiol* 2003; 90: 614-9.
- 91. Mossberg K, Willman C, Topor MA, Crook H, Patak S. Comparison of asynchronous versus synchronous arm crank ergometry. *Spinal Cord* 1999; 37: 569-74.
- Steinberg LL, Lauro FA, Sposito MM, Tufik S, Mello MT, Naffah-Mazzacoratti MG, Cavalheiro EA, Silva AC. Catecholamine response to exercise in individuals with different levels of paraplegia. *Braz J Med Biol Res* 2000; 33: 913-8.
- Yamasaki M, Komura T, Tahara Y, Muraki S, Tsunawake N, Ehara Y, Fujiie K. Relationship between physical characteristics and physiological responses during maximal arm cranking in paraplegics. *Spinal Cord* 1998; 36: 579-83.
- 94. Dallmeijer AJ, Hopman MT, Angenot EL, van der Woude LH. Effect of training on physical capacity and physical strain in persons with tetraplegia. *Scand J Rehabil Med* 1997; 29: 181-6.
- 95. Gass GC, Watson J, Camp EM, Court HJ, McPherson LM, Redhead P. The effects of physical training on high level spinal lesion patients. *Scand J Rehabil Med* 1980; 12: 61-5.
- Hopman MT, Dallmeijer AJ, Snoek G, van der Woude LH. The effect of training on cardiovascular responses to arm exercise in individuals with tetraplegia. *Eur J Appl Physiol Occup Physiol* 1996; 74: 172-9.
- 97. Lasko-McCarthey P, Davis JA. Protocol dependency of VO2max during arm cycle ergometry in males with quadriplegia. *Med Sci Sports Exerc* 1991; 23: 1097-101.
- Almenoff PL, Spungen AM, Lesser M, Bauman WA. Pulmonary function survey in spinal cord injury: influences of smoking and level and completeness of injury. *Lung* 1995; 173: 297-306.
- Linn WS, Adkins RH, Gong H, Jr., Waters RL. Pulmonary function in chronic spinal cord injury: a cross-sectional survey of 222 southern California adult outpatients. *Arch Phys Med Rehabil* 2000; 81: 757-63.
- Schilero GJ, Grimm DR, Bauman WA, Lenner R, Lesser M. Assessment of airway caliber and bronchodilator responsiveness in subjects with spinal cord injury. *Chest* 2005; 127: 149-55.
- 101. Schilero GJ, Grimm D, Spungen AM, Lenner R, Lesser M. Bronchodilator responses to metaproterenol sulfate among subjects with spinal cord injury. *J Rebabil Res Dev* 2004; 41: 59-64.
- 102. Silva AC, Neder JA, Chiurciu MV, Pasqualin DC, da Silva RC, Fernandez AC, Lauro FA, de Mello MT, Tufik S. Effect of aerobic training on ventilatory muscle endurance of spinal cord injured men. *Spinal Cord* 1998; 36: 240-5.
- 103. Almenoff PL, Alexander LR, Spungen AM, Lesser MD, Bauman WA. Bronchodilatory effects of ipratropium bromide in patients with tetraplegia. *Paraplegia* 1995; 33: 274-7.
- 104. Grimm DR, Chandy D, Almenoff PL, Schilero G, Lesser M. Airway hyperreactivity in subjects with tetraplegia is associated with reduced baseline airway caliber. *Chest* 2000; 118: 1397-404.
- Liaw MY, Lin MC, Cheng PT, Wong MK, Tang FT. Resistive inspiratory muscle training: its effectiveness in patients with acute complete cervical cord injury. *Arch Phys Med Rehabil* 2000; 81: 752-6.
- Lougheed MD, Flannery J, Webb KA, O'Donnell DE. Respiratory sensation and ventilatory mechanics during induced bronchoconstriction in spontaneously breathing low cervical quadriplegia. *Am J Respir Crit Care Med* 2002; 166: 370-6.

- 107. Spungen AM, Grimm DR, Strakhan M, Pizzolato PM, Bauman WA. Treatment with an anabolic agent is associated with improvement in respiratory function in persons with tetraplegia: a pilot study. *Mt Sinai J Med* 1999; 66: 201-5.
- Spungen AM, Dicpinigaitis PV, Almenoff PL, Bauman WA. Pulmonary obstruction in individuals with cervical spinal cord lesions unmasked by bronchodilator administration. *Paraplegia* 1993; 31: 404-7.
- 109. Walker J, Cooney M, Norton S. Improved pulmonary function in chronic quadriplegics after pulmonary therapy and arm ergometry. *Paraplegia* 1989; 27: 278-83.
- Wang TG, Wang YH, Tang FT, Lin KH, Lien IN. Resistive inspiratory muscle training in sleepdisordered breathing of traumatic tetraplegia. *Arch Phys Med Rehabil* 2002; 83: 491-6.
- 111. Figoni SF. Perspectives on cardiovascular fitness and SCI. J Am Paraplegia Soc 1990; 13: 63-71.
- 112. Bergh U, Kanstrup IL, Ekblom B. Maximal oxygen uptake during exercise with various combinations of arm and leg work. *J Appl Physiol* 1976; 41: 191-6.
- 113. O'Toole ML, Hiller DB, Crosby LO, Douglas PS. The ultraendurance triathlete: a physiological profile. *Med Sci Sports Exerc* 1987; 19: 45-50.
- 114. Gass GC, Camp EM. The maximum physiological responses during incremental wheelchair and arm cranking exercise in male paraplegics. *Med Sci Sports Exerc* 1984; 16: 355-9.
- Wicks JR, Oldridge NB, Cameron BJ, Jones NL. Arm cranking and wheelchair ergometry in elite spinal cord-injured athletes. *Med Sci Sports Exerc* 1983; 15: 224-31.
- Simard C, Noreau L, Pare G, Pomerleau P. Maximal physiological response during exertion in quadriplegic subjects. *Can J Appl Physiol* 1993; 18: 163-74.
- Hjeltnes N. Cardiorespiratory capacity in tetra- and paraplegia shortly after injury. Scand J Rehabil Med 1986; 18: 65-70.
- Hartung GH, Lally DA, Blancq RJ. Comparison of treadmill exercise testing protocols for wheelchair users. *Eur J Appl Physiol Occup Physiol* 1993; 66: 362-5.
- 119. Phillips BA, Lo SK, Mastaglia FL. Muscle force measured using "break" testing with a hand-held myometer in normal subjects aged 20 to 69 years. Arch Phys Med Rehabil 2000; 81: 653-61.
- 120. Ditunno JF, Jr., Stover SL, Freed MM, Ahn JH. Motor recovery of the upper extremities in traumatic quadriplegia: a multicenter study. *Arch Phys Med Rehabil* 1992; 73: 431-6.
- Burns AS, Ditunno JF. Establishing prognosis and maximizing functional outcomes after spinal cord injury: a review of current and future directions in rehabilitation management. *Spine* 2001; 26: S137-45.
- van Tuijl JH, Janssen-Potten YJ, Seele HA. Evaluation of upper extremity motor function tests in tetraplegics. *Spinal Cord* 2002; 40: 51-64.
- 123. Marino RJ, Huang M, Knight P, Herbison GJ, Ditunno JF, Jr., Segal M. Assessing selfcare status in quadriplegia: comparison of the quadriplegia index of function (QIF) and the functional independence measure (FIM). *Paraplegia* 1993; 31: 225-33.
- 124. Linn WS, Spungen AM, Gong H, Jr., Adkins RH, Bauman WA, Waters RL. Forced vital capacity in two large outpatient populations with chronic spinal cord injury. *Spinal Cord* 2001; 39: 263-8.
- Kelley A, Garshick E, Gross ER, Lieberman SL, Tun CG, Brown R. Spirometry testing standards in spinal cord injury. *Chest* 2003; 123: 725-30.
- 126. DeVivo MJ, Krause JS, Lammertse DP. Recent trends in mortality and causes of death among persons with spinal cord injury. *Arch Phys Med Rehabil* 1999; 80: 1411-9.
- 127. Spungen AM, Grimm DR, Lesser M, Bauman WA, Almenoff PL. Self-reported prevalence of pulmonary symptoms in subjects with spinal cord injury. *Spinal Cord* 1997; 35: 652-7.
- Le Foll-de Moro D, Tordi N, Lonsdorfer E, Lonsdorfer J. Ventilation efficiency and pulmonary function after a wheelchair interval-training program in subjects with recent spinal cord injury. *Arch Phys Med Rehabil* 2005; 86: 1582-6.
- 129. Zoeller RF, Jr., Riechman SE, Dabayebeh IM, Goss FL, Robertson RJ, Jacobs PL. Relation between muscular strength and cardiorespiratory fitness in people with thoracic-level paraplegia. Arch Phys Med Rehabil 2005; 86: 1441-6.

- Coutts KD, Rhodes EC, McKenzie DC. Maximal exercise responses of tetraplegics and paraplegics. J Appl Physiol 1983; 55: 479-82.
- 131. Hjeltnes N. Changes in cardiovascular responses to graded arm ergometry in tetra- and paraplegic patients during primary rehabilitation. In: Meys PJ, van der Woude LH, Grinten BA, de Boer YA, editors, *Ergonomics of manual wheelchair propulsion: state of the art*. Amsterdam: IOS Press; 1993. p. 79-91.
- 132. American College of Sports Medicine. *ACSM's guidelines for exercise testing and prescription*. 6th ed. Philadelphia: Lippincott, William & Wilkins; 2000.
- 133. Kilkens OJ, Dallmeijer AJ, de Witte LP, van der Woude LH, Post MW. The Wheelchair Circuit: Construct validity and responsiveness of a test to assess manual wheelchair mobility in persons with spinal cord injury. *Arch Phys Med Rehabil* 2004; 85: 424-31.
- 134. Kendall FP, McCreary EK, Provance PG. *Muscles: Testing and function.* 4th ed. Baltimore: Williams and Wilkins; 1993.
- 135. Andrews AW, Thomas MW, Bohannon RW. Normative values for isometric muscle force measurements obtained with hand-held dynamometers. *Phys Ther* 1996; 76: 248-59.
- American Thoracic Society. Standardization of Spirometry, 1994 Update. Am J Respir Crit Care Med 1995; 152: 1107-36.
- 137. Quanjer PH, Tammeling GJ, Cotes JE, Pedersen OF, Peslin R, Yernault JC. Lung volumes and forced ventilatory flows. Report Working Party Standardization of Lung Function Tests, European Community for Steel and Coal. Official Statement of the European Respiratory Society. *Eur Respir J Suppl* 1993; 16: 5-40.
- 138. Twisk JW. *Applied longitudinal data analysis for epidemiology; a practical guide*. Cambridge: Cambridge University Press; 2003.
- 139. Rasbash J, Browne W, Goldstein H, Yang M, Plewis I, Healy M, Woodhouse G, Draper D, Langford I, Lewis T. *A user's guide to MLwiN*. London: University of London; 2002.
- 140. de Groot S, Dallmeijer AJ, Kilkens OJ, van Asbeck FW, Nene AV, Angenot EL, Post MW, van der Woude LH. Course of gross mechanical efficiency in handrim wheelchair propulsion during rehabilitation of people with spinal cord injury: a prospective cohort study. *Arch Phys Med Rehabil* 2005; 86: 1452-60.
- 141. Janssen TW, van Oers CA, Rozendaal EP, Willemsen EM, Hollander AP, van der Woude LH. Changes in physical strain and physical capacity in men with spinal cord injuries. *Med Sci Sports Exerc* 1996; 28: 551-9.
- 142. Anke A, Aksnes AK, Stanghelle JK, Hjeltnes N. Lung volumes in tetraplegic patients according to cervical spinal cord injury level. *Scand J Rehabil Med* 1993; 25: 73-7.
- 143. Sawka MN, Glaser RM, Laubach LL, Al-Samkari O, Suryaprasad AG. Wheelchair exercise performance of the young, middle-aged, and elderly. *J Appl Physiol* 1981; 50: 824-8.
- 144. Janssen TW, van Oers CA, Hollander AP, Veeger HE, van der Woude LH. Isometric strength, sprint power, and aerobic power in individuals with a spinal cord injury. *Med Sci Sports Exerc* 1993; 25: 863-70.
- 145. van Langeveld S, Aufdemkampe G, van Asbeck FW. Reliability of force measurement with handheld dynamometry in healthy subjects and force measurement in patients with poliomyelitis anterior acuta. *J Rehabil Sciences* 1996; 1: 2-10.
- 146. Rheault W, Beal JL, Kubik KR, Nowak TA, Shepley JA. Intertester reliability of the hand-held dynamometer for wrist flexion and extension. *Arch Phys Med Rehabil* 1989; 70: 907-10.
- 147. Lydell M, Baigi A, Marklund B, Mansson J. Predictive factors for work capacity in patients with musculoskeletal disorders. *J Rebabil Med* 2005; 37: 281-5.
- Ditunno JF, Jr., Cohen ME, Hauck WW, Jackson AB, Sipski ML. Recovery of upper-extremity strength in complete and incomplete tetraplegia: a multicenter study. *Arch Phys Med Rehabil* 2000; 81: 389-93.
- 149. Kim CM, Eng JJ, Whittaker MW. Level walking and ambulatory capacity in persons with incomplete spinal cord injury: relationship with muscle strength. *Spinal Cord* 2004; 42: 156-62.

- Iseli E, Cavigelli A, Dietz V, Curt A. Prognosis and recovery in ischaemic and traumatic spinal cord injury: clinical and electrophysiological evaluation. *J Neurol Neurosurg Psychiatry* 1999; 67: 567-71.
- 151. Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Dallmeijer AJ, de Groot S, van der Woude LH. Changes in Physical Capacity During and After Inpatient Rehabilitation in Subjects With a Spinal Cord Injury. Arch Phys Med Rehabil 2006; 87: 741-8; Chapter 3 this thesis.
- Kirshblum SC, O'Connor KC. Predicting neurologic recovery in traumatic cervical spinal cord injury. Arch Phys Med Rehabil 1998; 79: 1456-66.
- 153. Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Bussmann JB. Physical capacity in wheelchair-dependent persons with a spinal cord injury: a critical review of the literature. *Spinal Cord* 2006; 44: 642-52; *Chapter 2 this thesis.*
- 154. Altman DG, Royston P. What do we mean by validating a prognostic model? *Stat Med* 2000; 19: 453-73.
- Scivoletto G, Morganti B, Molinari M. Neurologic recovery of spinal cord injury patients in Italy. Arch Phys Med Rehabil 2004; 85: 485-9.
- Waters RL, Meyer PR, Jr., Adkins RH, Felton D. Emergency, acute, and surgical management of spine trauma. Arch Phys Med Rehabil 1999; 80: 1383-90.
- 157. Field A. Regression. In: Wright DB, editor. *Discovering Statistics: using SPSS for Windows*. London: Sage Publications Ltd; 2000. p. 102-62.
- 158. de Groot S, Dallmeijer AJ, Post MW, van Asbeck FW, Nene AV, Angenot EL, van der Woude LH. Demographics of the Dutch multicenter prospective cohort study 'Restoration of mobility in spinal cord injury rehabilitation'. *Spinal Cord* 2006; 44: 668-75.
- 159. Gondoni LA, Liuzzi A, Titon AM, Taronna O, Nibbio F, Ferrari P, Leonetti G. A simple tool to predict exercise capacity of obese patients with ischaemic heart disease. *Heart* 2006; 92: 899-904.
- Eastwood EA, Hagglund KJ, Ragnarsson KT, Gordon WA, Marino RJ. Medical rehabilitation length of stay and outcomes for persons with traumatic spinal cord injury--1990-1997. Arch Phys Med Rehabil 1999; 80: 1457-63.
- 161. Larsson Lund M, Nordlund A, Nygard L, Lexell J, Bernspang B. Perceptions of participation and predictors of perceived problems with participation in persons with spinal cord injury. J Rehabil Med 2005; 37: 3-8.
- 162. Bloemen-Vrencken JH, Post MW, Hendriks JM, De Reus EC, De Witte LP. Health problems of persons with spinal cord injury living in the Netherlands. *Disabil Rehabil* 2005; 27: 1381-9.
- 163. Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Post MW, Dallmeijer AJ, van der Woude LH. Physical fitness in persons with a spinal cord injury: the association with complications and duration of rehabilitation. *Clin Rehabil* 2007; 21: 932-40; *Chapter 6 this thesis.*
- 164. Tate DG, Forchheimer MB, Krause JS, Meade MA, Bombardier CH. Patterns of alcohol and substance use and abuse in persons with spinal cord injury: risk factors and correlates. Arch Phys Med Rehabil 2004; 85: 1837-47.
- 165. Buchholz AC, Bugaresti JM. A review of body mass index and waist circumference as markers of obesity and coronary heart disease risk in persons with chronic spinal cord injury. *Spinal Cord* 2005; 43: 513-8.
- Scelza WM, Kalpakjian CZ, Zemper ED, Tate DG. Perceived barriers to exercise in people with spinal cord injury. *Am J Phys Med Rebabil* 2005; 84: 576-83.
- Schonherr MC, Groothoff JW, Mulder GA, Schoppen T, Eisma WH. Vocational reintegration following spinal cord injury: expectations, participation and interventions. *Spinal Cord* 2004; 42: 177-84.
- Krause JS, Broderick L. Outcomes after spinal cord injury: comparisons as a function of gender and race and ethnicity. *Arch Phys Med Rehabil* 2004; 85: 355-62.
- 169. Westgren N, Levi R. Quality of life and traumatic spinal cord injury. *Arch Phys Med Rehabil* 1998; 79: 1433-9.

- 170. Kirschblum S, Campagnolo DI, DeLisa JA. *Spinal Cord Medicine*. Philadelphia: Lippincott Williams & Wilkins; 2002.
- 171. Werhagen L, Budh CN, Hultling C, Molander C. Neuropathic pain after traumatic spinal cord injury--relations to gender, spinal level, completeness, and age at the time of injury. *Spinal Cord* 2004; 42: 665-73.
- McKinley WO, Jackson AB, Cardenas DD, DeVivo MJ. Long-term medical complications after traumatic spinal cord injury: a regional model systems analysis. *Arch Phys Med Rehabil* 1999; 80: 1402-10.
- 173. Chen Y, Devivo MJ, Jackson AB. Pressure ulcer prevalence in people with spinal cord injury: age-period-duration effects. *Arch Phys Med Rehabil* 2005; 86: 1208-13.
- 174. Vanden Bossche L, Vanderstraeten G. Heterotopic ossification: a review. *J Rebabil Med* 2005; 37: 129-36.
- 175. New P, Rawicki HB, Bailey MJ. Nontraumatic spinal cord injury rehabilitation: pressure ulcer pattern, prediction and impact. *Arch Phys Med Rehabil* 2004; 85: 87-93.
- Chen D, Apple DF, Jr., Hudson LM, Bode R. Medical complications during acute rehabilitation following spinal cord injury--current experience of the Model Systems. *Arch Phys Med Rehabil* 1999; 80: 1397-401.
- 177. Siddall PJ, Loeser JD. Pain following spinal cord injury. Spinal Cord 2001; 39: 63-73.
- 178. van Drongelen S, de Groot S, Veeger HE, Angenot EL, Dallmeijer AJ, Post MW, van der Woude LH. Upper extremity musculoskeletal pain during and after rehabilitation in wheelchair-using persons with a spinal cord injury. *Spinal Cord* 2006; 44: 152-9.
- 179. Bloemen-Vrencken JH, de Witte LP. Post-discharge nursing problems of spinal cord injured patients: on which fields can nurses contribute to rehabilitation? *Clin Rebabil* 2003; 17: 890-8.
- DeVivo MJ, Kartus PL, Rutt RD, Stover SL, Fine PR. The influence of age at time of spinal cord injury on rehabilitation outcome. *Arch Neurol* 1990; 47: 687-91.
- 181. Janssen TW, van Oers CA, van Kamp GJ, TenVoorde BJ, van der Woude LH, Hollander AP. Coronary heart disease risk indicators, aerobic power, and physical activity in men with spinal cord injuries. *Arch Phys Med Rehabil* 1997; 78: 697-705.
- 182. Teasell RW, Arnold JM, Krassioukov A, Delaney GA. Cardiovascular consequences of loss of supraspinal control of the sympathetic nervous system after spinal cord injury. Arch Phys Med Rehabil 2000; 81: 506-16.
- Cress ME, Meyer M. Maximal voluntary and functional performance levels needed for independence in adults aged 65 to 97 years. *Phys Ther* 2003; 83: 37-48.
- 184. Lance JW. Spasticity: Disordered Motor Control. In: Feldman RG, Young RR, Koella WP, editors. Symposium Synopsis. Chicago: Year Book Medical Publishers; 1980. p. 485-495.
- Convertino VA. Cardiovascular consequences of bed rest: effect on maximal oxygen uptake. Med Sci Sports Exerc 1997; 29: 191-6.
- 186. Capelli C, Antonutto G, Kenfack MA, Cautero M, Lador F, Moia C, Tam E, Ferretti G. Factors determining the time course of VO2(max) decay during bedrest: implications for VO2(max) limitation. *Eur J Appl Physiol* 2006; 98: 152-60.
- 187. Rittweger J, Belavy D, Hunek P, Gast U, Boerst H, Feilcke B, Armbrecht G, Mulder E, Schubert H, Richardson C, de Haan A, Stegeman DF, Schiessl H, Felsenberg D. Highly demanding resistive vibration exercise program is tolerated during 56 days of strict bed-rest. *Int J Sports Med* 2006; 27: 553-9.
- 188. Bleeker MW, De Groot PC, Rongen GA, Rittweger J, Felsenberg D, Smits P, Hopman MT. Vascular adaptation to deconditioning and the effect of an exercise countermeasure: results of the Berlin Bed Rest study. *J Appl Physiol* 2005; 99: 1293-300.
- 189. Maynard FM, Karunas RS, Waring WP, 3rd. Epidemiology of spasticity following traumatic spinal cord injury. *Arch Phys Med Rehabil* 1990; 71: 566-9.
- 190. Lechner HE, Frotzler A, Eser P. Relationship between self- and clinically rated spasticity in spinal cord injury. *Arch Phys Med Rebabil* 2006; 87: 15-9.

- Biering-Sorensen F, Nielsen JB, Klinge K. Spasticity-assessment: a review. Spinal Cord 2006; 44: 708-22.
- 192. Patrick E, Ada L. The Tardieu Scale differentiates contracture from spasticity whereas the Ashworth Scale is confounded by it. *Clin Rebabil* 2006; 20: 173-82.
- 193. Braddom R, Buschbacher R, Dumitru D, Johnson E, Matthews D, Sinaki M. *Physical Medicine and Rebabilitation.* Second ed. Philadelphia: Saunders; 2000.
- 194. Tooth L, McKenna K, Geraghty T. Rehabilitation outcomes in traumatic spinal cord injury in Australia: functional status, length of stay and discharge setting. *Spinal Cord* 2003; 41: 220-30.
- 195. Hall KM, Cohen ME, Wright J, Call M, Werner P. Characteristics of the Functional Independence Measure in traumatic spinal cord injury. *Arch Phys Med Rehabil* 1999; 80: 1471-6.
- 196. Nanda U, McLendon PM, Andresen EM, Armbrecht E. The SIP68: an abbreviated sickness impact profile for disability outcomes research. *Qual Life Res* 2003; 12: 583-95.
- 197. Hobart JC, Lamping DL, Freeman JA, Langdon DW, McLellan DL, Greenwood RJ, Thompson AJ. Evidence-based measurement: which disability scale for neurologic rehabilitation? *Neurology* 2001; 57: 639-44.
- Post MW, de Bruin A, de Witte L, Schrijvers A. The SIP68: a measure of health-related functional status in rehabilitation medicine. *Arch Phys Med Rehabil* 1996; 77: 440-5.
- Price GL, Kendall M, Amsters DI, Pershouse KJ. Perceived causes of change in function and quality of life for people with long duration spinal cord injury. *Clin Rebabil* 2004; 18: 164-71.
- Nash MS, van de Ven I, van Elk N, Johnson BM. Effects of circuit resistance training on fitness attributes and upper-extremity pain in middle-aged men with paraplegia. *Arch Phys Med Rehabil* 2007; 88: 70-5.
- 201. Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Post MW, Bussmann JB. Complications following spinal cord injury: occurrence and risk factors in a longitudinal study during and after inpatient rehabilitation. *J Rehabil Med* 2007; 39: 393-8; *Chapter 5 this thesis.*
- Sisto SA, Dyson-Hudson T. Dynamometry testing in spinal cord injury. J Rehabil Res Dev 2007; 44: 123-36.
- 203. Fisher CG, Noonan VK, Smith DE, Wing PC, Dvorak MF, Kwon BK. Motor recovery, functional status, and health-related quality of life in patients with complete spinal cord injuries. *Spine* 2005; 30: 2200-7.
- 204. Middleton JW, Harvey LA, Batty J, Cameron I, Quirk R, Winstanley J. Five additional mobility and locomotor items to improve responsiveness of the FIM in wheelchair-dependent individuals with spinal cord injury. *Spinal Cord* 2006; 44: 495-504.
- 205. Verhoef M, Barf HA, Post MW, van Asbeck FW, Gooskens RH, Prevo AJ. Functional independence among young adults with spina bifida, in relation to hydrocephalus and level of lesion. *Dev Med Child Neurol* 2006; 48: 114-9.
- 206. Mange KC, Ditunno JF, Jr., Herbison GJ, Jaweed MM. Recovery of strength at the zone of injury in motor complete and motor incomplete cervical spinal cord injured patients. *Arch Phys Med Rehabil* 1990; 71: 562-5.
- 207. Janssen TW, van Oers CA, Veeger HE, Hollander AP, van der Woude LH, Rozendal RH. Relationship between physical strain during standardised ADL tasks and physical capacity in men with spinal cord injuries. *Paraplegia* 1994; 32: 844-59.
- 208. van der Woude LH, Geurts A, Winkelman H, Veeger HE. Measurement of wheelchair rolling resistance with a push technique. In: van der Woude LH, Hopman MT, Kemenade CH, editors. *Biomedical aspects of manual wheelchair propulsion: the state of the art II.* Amsterdam: IOS press; 1999. p. 194-196.
- Drolet M, Noreau L, Vachon J, Moffet H. Muscle strength changes as measured by dynamometry following functional rehabilitation in individuals with spinal cord injury. *Arch Phys Med Rehabil* 1999; 80: 791-800.
- 210. Tasiemski T, Bergstrom E, Savic G, Gardner BP. Sports, recreation and employment following spinal cord injury--a pilot study. *Spinal Cord* 2000; 38: 173-84.

- 211. Kerstin W, Gabriele B, Richard L. What promotes physical activity after spinal cord injury? An interview study from a patient perspective. *Disabil Rehabil* 2006; 28: 481-8.
- 212. van den Berg-Emons HJ, Bussmann JB, Haisma JA, van der Woude LH, Stam HJ, Bergen MP. Poor physical activity after spinal cord injury; a prospective study on physical activity levels during inpatient rehabilitation till one year after discharge. Submitted to *Arch Phys Med Rebabil*.
- Linn WS, Spungen AM, Gong H, Jr., Bauman WA, Adkins RH, Waters RL. Smoking and obstructive lung dysfunction in persons with chronic spinal cord injury. *J Spinal Cord Med* 2003; 26: 28-35.
- 214. Mora S, Redberg RF, Cui Y, Whiteman MK, Flaws JA, Sharrett AR, Blumenthal RS. Ability of exercise testing to predict cardiovascular and all-cause death in asymptomatic women: a 20-year follow-up of the lipid research clinics prevalence study. *Jama* 2003; 290: 1600-7.
- 215. Myers J, Lee M, Kiratli J. Cardiovascular disease in spinal cord injury: an overview of prevalence, risk, evaluation, and management. *Am J Phys Med Rehabil* 2007; 86: 142-52.
- 216. DeVivo M, Biering-Sorensen F, Charlifue S, Noonan V, Post M, Stripling T, Wing P. International Spinal Cord Injury Core Data Set. *Spinal Cord* 2006; 44: 535-40.
- de Groot S, Zuidgeest M, van der Woude LH. Standardization of measuring power output during wheelchair propulsion on a treadmill Pitfalls in a multi-center study. *Med Eng Phys* 2006; 28: 604-12.
- 218. Suzuki M, Omori M, Hatakeyama M, Yamada S, Matsushita K, Iijima S. Predicting recovery of upper-body dressing ability after stroke. *Arch Phys Med Rehabil* 2006; 87: 1496-502.
- 219. Gresham GE, Labi ML, Dittmar SS, Hicks JT, Joyce SZ, Stehlik MA. The Quadriplegia Index of Function (QIF): sensitivity and reliability demonstrated in a study of thirty quadriplegic patients. *Paraplegia* 1986; 24: 38-44.
- 220. Catz A, Itzkovich M, Agranov E, Ring H, Tamir A. SCIM--spinal cord independence measure: a new disability scale for patients with spinal cord lesions. *Spinal Cord* 1997; 35: 850-6.
- 221. Catz A, Itzkovich M. Spinal Cord Independence Measure: Comprehensive ability rating scale for the spinal cord lesion patient. *J Rebabil Res Dev* 2007; 44: 65-8.
- 222. Hicks AL, Martin KA, Ditor DS, Latimer AE, Craven C, Bugaresti J, McCartney N. Long-term exercise training in persons with spinal cord injury: effects on strength, arm ergometry performance and psychological well-being. *Spinal Cord* 2003; 41: 34-43.
- 223. Cardol M, de Jong BA, van den Bos GA, Beelem A, de Groot IJ, de Haan RJ. Beyond disability: perceived participation in people with a chronic disabling condition. *Clin Rebabil* 2002; 16: 27-35.
- 224. Whiteneck GG, Charlifue SW, Gerhart KA, Overholser JD, Richardson GN. Quantifying handicap: a new measure of long-term rehabilitation outcomes. *Arch Phys Med Rehabil* 1992; 73: 519-26.
- 225. de Groot S, Post MW, Niezen H, Rijken F, Wolding K, Postma K, Bloemen-Vrenken J, Valent LJ, Luthart P, Schuitemaker M, van der Woude LH. Patiëntmonitoring tijdens de dwarslaesierevalidatie: het interpreteren van testgegevens. *Revalidata* 2007; 135: 17-9.
- 226. van Langeveld SA, Post MW, van Asbeck FW, Postma K, Leenders J, Liefhebber H. Bruikbaarheid van een classificatie van paramedische dwarslaesie behandelingen. *Revalidata* 2007; 135: 20-4.
- 227. Heath GW, Fentem PH. Physical activity among persons with disabilities--a public health perspective. *Exerc Sport Sci Rev* 1997; 25: 195-234.
- 228. Washburn RA, Zhu W, McAuley E, Frogley M, Figoni SF. The physical activity scale for individuals with physical disabilities: development and evaluation. *Arch Phys Med Rehabil* 2002; 83: 193-200.
- 229. Bussmann JB, Hartgerink I, van der Woude LH, Stam HJ. Measuring physical strain during ambulation with accelerometry. *Med Sci Sports Exerc* 2000; 32: 1462-71.

### Dankwoord

Dr JBJ Bussmann en Dr LHV van der Woude, bedankt voor jullie gedegen begeleiding de afgelopen 4 jaren. Hans, je hebt mij gestimuleerd om kritisch na te denken en te vragen: "waarom zo, en niet anders?". Je hebt structuur geboden en vertrouwen gegeven. Luc, je kennis is verhelderend en inspirerend. Je altijd merkbare enthousiasme heeft me steeds gemotiveerd.

Prof Dr HJ Stam, bedankt voor een inspirerende onderzoekstijd op de afdeling revalidatiegeneeskunde. De vriendelijke werksfeer maakte dat ik dagelijks met plezier naar mijn werk ging.

Dr MP Bergen en Drs TAR Sluis, bedankt voor jullie deelname aan de projectgroep vanuit Rijndam revalidatiecentrum. Michael en Tebbe, jullie brachten de onderzoekswereld en complexe analyses dichterbij de realiteit van de dagelijkse patiëntenzorg.

De onderzoeksassistenten uit Revalidatiecentrum de Hoogstraat, Revalidatiecentrum Amsterdam, Het Roessingh, Revalidatiecentrum Hoensbroeck, Revalidatiecentrum Beatrixoord, Revalidatiecentrum Heliomare en Rijndam Revalidatiecentrum ben ik zeer dankbaar voor de enorme inspanning en zorgvuldigheid bij het verzamelen van de data.

Dr Sonja de Groot, zonder jouw kennis van data en analyses, zonder jouw geduld, bereik- en benaderbaarheid was dit echt niet mogelijk geweest. Bovendien was het erg gezellig om met jou, Linda Valent, Karin Postma en Rogier Broeksteeg op ISCOS-pad te zijn.

Dr Annet Dallmeijer, Dr Marcel Post, Dr Rita van den Berg-Emons: bedankt, voor jullie leerzame en kritische bijdrage als medeauteur.

Channah Nieuwenhuizen, Wilma van der Slot, Mark de Niet, Marian Michielsen, Agnes Willemse-van Son, Diana Wiegerink, Berbke van Ginneken, Bart Bartels, Wim Janssen, Fabiënne Schasfoort, Robert Pangalila, Laurien Aben, Mireille Donkervoort en Laurien Buffart: zo gemotiveerd, behulpzaam, getalenteerd en tegelijkertijd kritisch en niet bang. Ik wil jullie bedanken voor een goede tijd op de 16e etage. Ook de collega's van beneden wil ik bedanken. Betty Koehorst, Marij Roebroek, Rita van den Berg, Herwin Horemans, Ruud Selles, Monique Ardon, Ton Schreuders, Jetty van Meeteren, Henri Hurkmans, Jiska Kempen, Winifred Paulis, jullie hebben vaak interesse getoond en geholpen bij het maken van keuzes.

Agnes Willemse-van Son en Laurien Buffart: ik voel me sterk omdat jullie straks aan weerszijde van mij zullen staan. Agnes, je doortastendheid, menselijk inzicht en doorzettingsvermogen zijn zeer waardevol. Laurien, je bent een voorbeeld van opgewektheid en energie.

Uiteindelijk zegt men dan iets om de allerbelangrijkste personen te bedanken. Lieve Henk en Rommy, bedankt voor jullie onvoorwaardelijke liefde, vertrouwen en steun. Ik houd zo veel van jullie. Liefste Martin, jij maakt me gelukkiger. Ole, allerliefste, jij bent het meest bijzondere wat ons ooit is overkomen.

## **Curriculum vitae**

The author, Janneke Haisma, was born in Groningen on the 9th of September 1974. In 1991 she completed the O-level examinations in Harare, Zimbabwe. She finished her VWO, voortgezet wetenschappelijk onderwijs, in Groningen in 1992.

In 1997 the author completed her training as a physical therapist at the Hanzehogeschool Groningen. Hereafter, she studied medicine at the Rijksuniversiteit Groningen. In 2001 she did a master's research project at the Department of Neurology, Western General Hospital in Edinburgh, Scotland. In 2003 she completed the residency training at Deventer Hospital.

After the author received her medical degree in 2003, she worked at the Department of Neurology, Leyenburg Hospital in The Hague. In March 2004 she began the PhD training at the Department of Rehabilitation Medicine, Erasmus University MC in Rotterdam. In January 2008 the author started her training in rehabilitation medicine at Rijndam Rehabilitation Centre in Rotterdam.

# Articles in international journals

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Bussmann JB. Physical capacity in wheelchair-dependent persons with a spinal cord injury: a critical review of the literature. *Spinal Cord* 2006; 44: 642-52.

Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Dallmeijer AJ, de Groot S, van der Woude LH. Changes in Physical Capacity During and After Inpatient Rehabilitation in Subjects With a Spinal Cord Injury. *Arch Phys Med Rehabil* 2006; 87: 741-8.

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Groot de S, Dallmeijer AJ, Bussmann JB. Prognostic models for physical capacity at discharge and 1 year after discharge from rehabilitation in persons with a spinal cord injury. *Arch Phys Med Rehabil* 2007; 88: 1694-1703.

Haisma JA, van der Woude LH, Stam HJ, Bergen MP, Sluis TA, Post MW, Bussmann JB. Complications following spinal cord injury: occurrence and risk factors in a longitudinal study during and after inpatient rehabilitation. *J Rehabil Med 2007*, 39: 393-8.

Haisma JA, Bussmann JB, Stam HJ, Sluis TA, Bergen MP, Post MW, Dallmeijer AJ, van der Woude LH. Physical fitness in persons with a spinal cord injury: the association with complications and duration of rehabilitation. *Clin Rehabil* 2007; 41: 932-40.

Hand PJ, Wardlaw JM, Rowat AM, Haisma JA, Lindley RI, Dennis MS. Magnetic resonance brain imaging in patients with acute stroke: feasibility and patient related difficulties. *J Neurol Neurosurg Psychiatry* 2005; 76: 1525-7.

Hand PJ, Haisma JA, Kwan J, Lindley RI, Lamont B, Dennis MS, Wardlaw JM. Interobserver agreement for the bedside clinical assessment of suspected stroke. *Stroke* 2006; 37: 776-80.

Haisma JA, Bussmann JB, Stam HJ, Bergen MP, Sluis TA, van den Berg-Emons HJ, Post MW, van der Woude LH. Physical independence and health-related functional status following a spinal cord injury: a prospective study on the association with physical capacity and complications. Submitted to: *J Rehabil Med.* 

van den Berg-Emons HJ, Bussmann JB, Haisma JA, van der Woude LH, Stam HJ, Bergen MP. Poor physical activity after spinal cord injury; a prospective study on physical activity levels during inpatient rehabilitation till one year after discharge. Submitted to: *Arch Phys Med Rehabil.*